

City Council

James A. Kaminski Ward 1

Annmarie P. Roff Ward 2

Thomas M. Clark Ward 3

Gordon C. Short Ward 4

James E. Carbone At-Large

Kelly A. Kosek At-Large

Brian M. Spring At-Large

Aimee Pientka, MMC Clerk of Council

City of Strongsville

16099 Foltz Parkway Strongsville, Ohio 44149-5598 Phone: 440-580-3110 www.strongsville.org

September 12, 2024

MEETING NOTICE

City Council has scheduled the following meetings for Monday, September 16, 2024, to be held in the Caucus Room and the Council Chamber at the Mike Kalinich Sr. City Council Chamber, 18688 Royalton Road:

<u>Caucus will begin at 6:45 p.m.</u> All committees listed will meet immediately following the previous committee:

6:45 P.M.

<u>Public Service & Conservation Committee</u> will meet to discuss Ordinance Nos. 2024-121 and 2024-135.

<u>Planning, Zoning & Engineering Committee</u> will meet to discuss Ordinance Nos. 2024-136, 2024-137 and 2024-138.

<u>Public Safety & Health Committee</u> will meet to discuss Ordinance No. 2024-139.

<u>Finance Committee</u> will meet to discuss Ordinance No. 2024-140 and Resolution No. 2024-141.

<u>Committee of the Whole</u> will meet to discuss Ordinance No. 2024-142.

7:00 P.M. Regular Council Meeting

Any other matters that may properly come before this Council may also be discussed.

BY ORDER OF THE COUNCIL:

Aimee Pientka, MMC Clerk of Council

STRONGSVILLE CITY COUNCIL REGULAR MEETING MONDAY, SEPTEMBER 16, 2024 AT 7:00 P.M.

Mike Kalinich Sr. City Council Chamber 18688 Royalton Road, Strongsville, Ohio

AGENDA

- 1. CALL TO ORDER:
- PLEDGE OF ALLEGIANCE:
- CERTIFICATION OF POSTING:
- 4. ROLL CALL:
- 5. COMMENTS ON MINUTES:
 - Regular Council Meeting September 3, 2024
- 6. APPOINTMENTS, CONFIRMATIONS AWARDS AND RECOGNITION:
- 7. REPORTS OF COUNCIL COMMITTEE:
 - ECONOMIC DEVELOPMENT Clark
 - BUILDING & UTILITIES Roff
 - PUBLIC SAFETY AND HEALTH Roff
 - RECREATION AND COMMUNITY SERVICES Spring
 - SCHOOL BOARD Spring
 - FINANCE Short
 - SOUTHWEST GENERAL HEALTH SYSTEM Short
 - COMMUNICATIONS AND TECHNOLOGY Kaminski
 - PLANNING, ZONING AND ENGINEERING Kosek
 - PUBLIC SERVICE AND CONSERVATION Kosek
 - COMMITTEE-OF-THE-WHOLE Carbone
- 8. REPORTS AND COMMUNICATIONS FROM THE MAYOR, DIRECTORS OF DEPARTMENTS AND OTHER OFFICERS:
 - MAYOR PERCIAK:
 - FINANCE DEPARTMENT:
 - LAW DEPARTMENT:
- 9. AUDIENCE PARTICIPATION:

10. ORDINANCES AND RESOLUTIONS:

- Ordinance No. 2024-121 by Mayor Perciak and All Members of Council. AN ORDINANCE AUTHORIZING THE MAYOR TO EMPLOY SPECIAL LEGAL COUNSEL AND ENTER INTO A RETAINER AGREEMENT FOR PROFESSIONAL SERVICES TO THE CITY OF STRONGSVILLE IN CONNECTION WITH MATTERS RELATING TO PROPERTY DAMAGE INCURRED AT THE WESTWOOD LIFT STATION ON WESTWOOD DRIVE IN STRONGSVILLE, AND DECLARING AN EMERGENCY. First reading 09-03-24.
- Ordinance No. 2024-135 by Mayor Perciak and All Members of Council. AN ORDINANCE AUTHORIZING THE MAYOR TO ENTER INTO A CONTRACT FOR THE OPERATIONS, MAINTENANCE AND MANAGEMENT SERVICES OF THE CITY'S WASTEWATER TREATMENT FACILITIES AND APPURTENANCES, AND DECLARING AN EMERGENCY.
- Ordinance No. 2024-136 by Mayor Perciak and All Members of Council. AN ORDINANCE AMENDING SECTIONS 1242.07(b)(10) AND 1258.06(a)(6), OF TITLE SIX OF PART TWELVE-PLANNING AND ZONING CODE OF THE CODIFIED ORDINANCES OF THE CITY OF STRONGSVILLE IN ORDER TO UPDATE THE REGULATION OF CERTAIN ESTABLISHMENTS AS CONDITIONAL USE PERMITS IN CERTAIN DISTRICTS, AND DECLARING AN EMERGENCY.
- Ordinance No. 2024-137 by Mayor Perciak and All Members of Council. AN ORDINANCE AUTHORIZING AND CONSENTING TO COLUMBIA TOWNSHIP PREPARING AND SUBMITTING AN APPLICATION FOR STATE OF OHIO ISSUE 1 FUNDING FOR THE ASPHALT RESURFACING OF NORTH MARKS ROAD FROM SNELL (ALBION) ROAD TO SPRAGUE ROAD, IN THE CITY OF STRONGSVILLE AND COLUMBIA TOWNSHIP; AUTHORIZING AND DIRECTING THE MAYOR TO EXECUTE A COOPERATION AGREEMENT REGARDING THE PROJECT; AND DECLARING AN EMERGENCY.
- Ordinance No. 2024-138 by Mayor Perciak and All Members of Council. AN ORDINANCE AUTHORIZING AND DIRECTING THE MAYOR TO ISSUE AND APPROVE CHANGE ORDER NOS. 12 THROUGH 14 FOR AN INCREASE IN THE CONTRACT PRICE CONSISTENT WITH THE PROVISIONS OF THE CONTRACT BETWEEN THE CITY OF STRONGSVILLE AND DIGIOIA-SUBURBAN EXCAVATING, LLC, IN CONNECTION WITH THE PROSPECT ROAD STORM SEWER IMPROVEMENTS PROJECT, AND DECLARING AN EMERGENCY.
- Ordinance No. 2024-139 by Mayor Perciak and All Members of Council. AN ORDINANCE AUTHORIZING THE MAYOR TO ENTER INTO A STATE OF OHIO ADMINISTRATIVE SERVICES CONTRACT FOR THE PURCHASE OF ONE (1) NEW SUTPHEN STOCK #568 CUSTOM PUMPER ON SUTPHEN MONARCH CHASSIS WITH RELATED VEHICLE MODIFICATIONS AND APPURTENANCES, FOR USE BY THE FIRE DEPARTMENT, AND DECLARING AN EMERGENCY.
- Ordinance No. 2024-140 by Mayor Perciak. AN ORDINANCE MAKING APPROPRIATIONS FOR THE ANNUAL EXPENSES AND OTHER EXPENDITURES OF THE CITY OF STRONGSVILLE, OHIO, FOR THE YEAR 2024 AND REPEALING ORDINANCE NUMBER 2024-093.

- Resolution No. 2024-141 by Mayor Perciak and All Members of Council. A RESOLUTION ACCEPTING THE AMOUNTS AND RATES AS DETERMINED BY THE BUDGET COMMISSION AND AUTHORIZING THE NECESSARY TAX LEVIES AND CERTIFYING THEM TO THE COUNTY FISCAL OFFICER, AND DECLARING AN EMERGENCY.
- Ordinance No. 2024-142 by Mayor Perciak and All Members of Council. AN ORDINANCE AUTHORIZING THE MAYOR TO ENTER INTO A PURCHASE AGREEMENT FOR CERTAIN PROPERTY LOCATED AT 21500 LUNN ROAD, AND DECLARING AN EMERGENCY.
- 11. COMMUNICATIONS, PETITIONS AND CLAIMS:

Application for Permit: **NEW - D1 - D2 - D3:** To: Board of Park Commissioners Cleveland Metropolitan Park District, **DBA: The Chalet**, 16200 Valley Parkway, Strongsville, Ohio 44136 (Responses must be postmarked no later than 09/25/2024).

- 12. MISCELLANEOUS BUSINESS:
- 13. ADJOURNMENT:

CITY OF STRONGSVILLE, OHIO

ORDINANCE NO. 2024 - 121

By: Mayor Perciak and All Members of Council

AN ORDINANCE AUTHORIZING THE MAYOR TO EMPLOY SPECIAL LEGAL COUNSEL AND ENTER INTO A RETAINER AGREEMENT FOR PROFESSIONAL SERVICES TO THE CITY OF STRONGSVILLE IN CONNECTION WITH MATTERS RELATING TO PROPERTY DAMAGE INCURRED AT THE WESTWOOD LIFT STATION ON WESTWOOD DRIVE IN STRONGSVILLE, AND DECLARING AN EMERGENCY.

BE IT ORDAINED BY THE COUNCIL OF THE CITY OF STRONGSVILLE, COUNTY OF CUYAHOGA AND STATE OF OHIO:

- Section 1. That the Mayor be and is hereby authorized and directed to employ special legal counsel and enter into a retainer agreement with ATTORNEY DAVID M. CUPPAGE, and the law firm of McCarthy, Lebit, Crystal & Liffman Co., Lpa, to provide professional legal services to the City with regard to a matter involving property damage to the Westwood Lift Station located at 14600 Westwood Drive in connection with the Westwood Lift Station Rehabilitation Project, as a result of prohibited discharges emanating from EMSCO Distributors located at 22350 Royalton Road, Strongsville, Ohio, in accordance with their proposal attached hereto as Exhibit "A", and as approved by the Law Director.
- **Section 2.** That the funds for the purpose of such services and said agreement have been appropriated and shall be paid from the Sanitary Sewer Fund; and the Director of Finance be and is hereby authorized and directed to issue payment in accordance with the terms and conditions of such proposal and retainer agreement.
- **Section 3.** That it is found and determined that all formal actions of this Council concerning and relating to the adoption of this Ordinance were adopted in an open meeting of this Council; and that all deliberations of this Council, and any of its committees, that resulted in such formal action were in meetings open to the public in compliance with all legal requirements.
- **Section 4.** That this Ordinance is hereby declared to be an emergency measure immediately necessary for the preservation of the public peace, property, health, safety and welfare of the City, and for the further reason that the retention of such legal services is immediately necessary in order to protect the legal interests of the City and provide representation to the City in connection with property damage incurred at the Westwood Lift Station, and conserve public funds. Therefore, provided this Ordinance receives the affirmative vote of two-thirds of all members elected to Council, it shall take effect and be in force immediately upon its passage and approval by the Mayor; otherwise from and after the earliest period allowed by law.

President of Council	Approved:Mayor	
Date Passed:	Date Approved:	

CITY OF STRONGSVILLE, OHIO ORDINANCE NO. 2024 – <u>121</u> Page 2

	<u>Yea</u>	<u>Nay</u>	Attest:Clerk of Council
Carbone Clark Kaminski Kosek Roff Short Spring			Ord. No 2024-121 Amended:
			Public HrgRef:Ref:



David M. Cuppage
Attorney at Law
Phone: 216,696,1422, Ext. 232
dmc@nccarthylebit.com

AUG 1 2 2024

LAW DEPARTMENT CITY OF STRONGSVILLE

August 12, 2024

VIA EMAIL ONLY

Mayor Thomas P. Perciak c/o Neal M. Jamison, Law Director City of Strongsville 16009 Foltz Parkway Strongsville, Ohio 44149

Strongsville.law@strongsville.org

Re: Retainer Agreement

Dear Mayor Perciak:

You have asked this Firm to provide legal counsel and representation to the City of Strongsville (the "City") in the matter involving property damage to the Westwood Lift Station located at 14600 Westwood Drive in connection with the Westwood Lift Station Rehabilitation Project. The property damage appears to be the result of prohibited discharges emanating from EMSCO Distributors, located at 22350 Royalton Road, Strongsville, Ohio 44136.

This letter sets forth our Agreement concerning our representation of the City.

- 1. Unless otherwise agreed to in advance, hourly charges will be billed at the Firm's guideline rates for partners, associates, paralegals and legal assistants. My hourly rate for this representation will be \$350 per hour. In some instances, I may utilize other attorneys to do specific tasks under my direction because they will be able to perform them approximately in the same amount of time or less but at a lower cost to the City. We ask that you promptly review all billing statements and immediately bring to our attention any discrepancies or questions that you might have.
- 2. We shall have the authority to make advances for expenses on the City's behalf on such amounts we shall determine best in representing the City in this matter, but not to exceed \$250 without advance approval from you. Those advances may include (but are not necessarily limited to) expenses such as long distance telephone calls, photocopies (Xerox), travel, parking, deposition expenses, court costs and other disbursements which we deem are necessary to assist the City in the proper handling of the matter for which we are being retained.
- 3. This Agreement may be executed in counterparts. The Agreement may be modified only in writing signed by the parties to be bound.

Mayor Thomas P. Perciak c/o Neal M. Jamison, Law Director August 12, 2024 Page 2

- 4. This Agreement represents the entire agreement between us as further governed by Ohio's Code of Professional Responsibility.
- 5. It is our practice to communicate by email where it is practical and appropriate. If, at any time you provide us with an email address or contact us by email, we will assume that you are agreeable to this form of communication.

We appreciate the opportunity to represent the City.

Please signify your acceptance of this Agreement by signing in the space provided below, completing the date and returning the originally-signed document to me.

Very truly yours,

McCARTHY, LEBIT, CRYSTAL & LIFFMAN CO., L.P.A.

DAVID M. CUPPAGE

READ, APPROVED AND AGREED TO THIS ___ DAY OF XXXXXXX, 2024.
September

The City of Strongsville

By: Thomas P. Perciak

Its: Mayor

CITY OF STRONGSVILLE, OHIO

ORDINANCE NO. 2024 - 135

By: Mayor Perciak and All Members of Council

AN ORDINANCE AUTHORIZING THE MAYOR TO ENTER INTO A CONTRACT FOR THE OPERATIONS, MAINTENANCE AND MANAGEMENT SERVICES OF THE CITY'S WASTEWATER TREATMENT FACILITIES AND APPURTENANCES, AND DECLARING AN EMERGENCY.

WHEREAS, by and through Resolution No. 2024-091, Council authorized the Mayor to advertise a Request for Proposals ("RFP") for consulting services for the operations, maintenance and management services of the City's Wastewater Treatment facilities and appurtenances; and

WHEREAS, the City has received one (1) proposal from a company having substantial experience with the City, and which is recommended by the City's independent engineering consultant that finds such proposal to be advantageous, competitive, in compliance with the specifications required by the City through its RFP, and in the best interests of the City; and

WHEREAS, Council is, therefore, desirous of proceeding to award and enter into a contract for such services for a five (5) year period, and in accordance with the RFP requirements and the contract document incorporated therein.

NOW, THEREFORE, BE IT ORDAINED BY THE COUNCIL OF THE CITY OF STRONGSVILLE, COUNTY OF CUYAHOGA AND STATE OF OHIO:

Section 1. That this Council finds and determines that the proposal submitted by VEOLIA WATER NORTH AMERICA-CENTRAL, LLC for the operations, maintenance and management services of the City's Wastewater Treatment facilities and appurtenances in the City for a five (5) year period commencing October 1, 2024, meets the requirements set forth in the Request for Proposals on file in the office of the Director of Public Service, is in compliance with the applicable requirements for proposals and contracts established by the laws of the City and the State, and is the lowest and best proposal for the contract. All other proposals for this contract, if any, are hereby rejected.

Section 2. That the Mayor be and hereby is authorized and directed to enter into an Agreement with VEOLIA WATER NORTH AMERICA-CENTRAL, LLC for the operations, maintenance and management services of the City's Wastewater Treatment facilities and appurtenances, substantially in the form attached hereto as Exhibit "A," and incorporated herein, together with various Exhibits identified in said Agreement, and in an amount for the first contract year of an annual fixed fee not to exceed the sum of \$2,573,133.00.

Section 3. That the funds necessary for the purposes of said contract have been appropriated for the year 2024, and shall be paid now and in future contract years pursuant to lawful appropriation ordinances, all from the Sanitary Sewer Fund.

Section 4. That it is found and determined that all formal actions of this Council concerning and relating to the adoption of this Ordinance were adopted in an open meeting of this Council; and that all deliberations of this Council, and any of its committees, that resulted in

CITY OF STRONGSVILLE, OHIO ORDINANCE NO. 2024 – <u>135</u> Page 2

such formal action were in meetings open to the public in compliance with all legal requirements.

Section 5. That this Ordinance is hereby declared to be an emergency measure necessary for the immediate preservation of the public peace, property, health, safety and welfare of the City, and for the further reason that it is immediately necessary to enter into said contract in order to avoid legal issues arising from the currently expiring contract, to continue the operations, maintenance and management of the aforesaid facilities and appurtenances, provide for safe disposal of sanitary waste within the City, and to conserve public funds. Therefore, provided this Ordinance receives the affirmative vote of two-thirds of all members elected to Council, it shall take effect and be in force immediately upon its passage and approval by the Mayor; otherwise from and after the earliest period allowed by law.

-	President of	Council	Mayor
Date Passed:			Date Approved:
	<u>Yea</u>	<u>Nay</u>	Attest:Clerk of Council
Carbone Clark Kaminski Kosek Roff Short Spring			Ord. No. 2624 - 135 Amended:
			Public HrgRef: Adopted: Defeated:

Annround.

CITY OF STRONGSVILLE, OHIO WASTEWATER FACILITIES OPERATIONS, MAINTENANCE AND MANAGEMENT SERVICES AGREEMENT

2024



CITY OF STRONGSVILLE 16099 FOLTZ PARKWAY STRONGSVILLE, OHIO 44149

CITY OF STRONGSVILLE, OHIO WASTEWATER FACILITIES OPERATIONS, MAINTENANCE AND MANAGEMENT SERVICES

AGREEMENT INDEX

Article 1 - I	Occuments	D-5
1.01	Agreement Documents	D-5
Article 2 - T	erm of Agreement and Authorized Representatives	D-5
2.01	Term of Agreement	D-5
2.02	Authorized Representatives	D-5
Article 3 - I	nterpretation and Definitions	D-5
3.01	Interpretation	D-5
3.02	Definitions	D-6
Article 4 - F	Lepresentations and Warranties	D-12
4.01	Representations and Warranties of the City	D-12
4.02	Representations and Warranties of the Company	D-13
4.03	Materiality of Representations	D-14
	ystem Description	D-14
5.01	General	D-14
5.02		D-14
	Operation, Maintenance and Management	D-15
6.01	General	D-15
6.02	Exclusions	D-15
6.03	Operation of Facilities	D-15
6.04	Maintenance of Facilities	D-16
6.05	Odor Control	D-17
6.06	Compliance With Applicable Law	D-17
6.07	Industrial Pretreatment	D-17
6.08	Mercury Pollutant Minimization Program	D-18
6.09		D-18
	Residuals Management	D-19
6.10 6.11		D-19
6.12	Equipment, Fixed assets, Consumables and Spare Parts	D-19
	Emergency Response	D-20
6.13	Safety of Persons and Property	D-20
6.14	Disposal of Surplus Equipment and Scrap Metal	D-20
6.15	Loss, Damage or Destruction of Assets	
6.16	Uncontrollable Circumstance	D-21
6.17	Reporting and Record Keeping	D-21
	Personnel	D-22
7.01	Adequate Staffing	D-22
7.02	Labor Organizations and Collective Bargaining	D-22
7.03	Training of Employees	D-22
7.04		D-22
7.05	No Restriction on Employment	D-23
7.06		D-23
	Project Manager	D-23
7.08		D-24
7.09		D-24
	Compensation	D-24
8.01	Service Fee	D-24
8.02	Additional Services: Change in Scope of Services	D-26
8.03	Billing and Payment	D-27
8.04	Estimates and Adjustments	D-27
8.05	Interest on Overdue Payments	D-27
8.06	Payment Disputes	D-27
8.07	Offsets	D-28
8.08	Tax Exemption of Facilities	D-28
8.09	Sales Tax	D-28

the state of the s	D-28
7,01 Octobra 1.11,111,111,111,111,111,111,111,111,11	D-28
9.02 Procedures for Implementing of Capital Improvements	D-28
9.03 Capital Improvements at Company's Request	D-29
	D-29
	D-30
	D-30
	D-30
	D-31
	D-31
1010 Mediana Chambre 1111	D-31
10100 0 000 0 0000000000000000000000000	D-31
10100 observe mira i ion observe a miranti mir	D-32
Atticle 11 "1 citothique Bond :	D-32
**************************************	D-32
	D-32 D-32
	D-32 D-32
, _,	D-32 D-33
	D-33
toto in a management of the first of the fir	D-33 D-34
13102 marining in any transfer and the second secon	
10100 11000 million at the artist of the art	D-35
TOTAL PROPERTY OF THE PROPERTY	D-35
15.05 Remotics Cumulative	D-36
13.00 Entitled of Endoling	D-36
15.01	D-36
	D-36
THE COUNTY OF TH	D-36
The standard and the standard	D-37
2	D-37
The second of th	D-37
	D-38
10101 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	D-38
10102 101111111111111111111111111111111	D-38
15.03 Termination for Labor Unrest I	D-38
1010 1 2 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	D-38
15.05 Termination for Insufficient Funding	D-38
	D-38
15.07 Remedies of the City	D-38
	D-38
	D-38
15.10 Operations Cooperation and Transfer of Personnel	D-39
· · · · · · · · · · · · · · · · · · ·	D-40
	D-40
THEORY IT INDOORMITOODS THE THEORY IN THE PROPERTY OF THE PROP	D-40
2, 101 Audi Gilliani Color Col	D-41
	D-41
17.05 Relationship of Factors 1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	D-41
17.04 Cost baostalitation ::::::::::::::::::::::::::::::::::::	D-42
17.05 1 dono records Ban 11.11.11.11.11.11.11.11.11.11.11.11.11.	D-42
17,000 tituliteipai ineotite 1411 1711 1711 1711 1711 1711 1711 171	D-42
11101 11011000 11111 11111011-1111111111	D-43
17,00 (7,00)	D-43
17.09 Entire 18 Continue, tribunitations and 1 months in 1.10 to 1.10	D-43

17.12 17.13 17.14 17.15	11 Consent to JurisdictionD-12 Non-Waiver of ImmunityD-13 CounterpartsD-14 SeverabilityD-15 Liability of Officers and EmployeesD-16 Third Party BeneficiaryD-		
Exhibi Exhibi Exhibi Exhibi Exhibi Exhibi	it 2 it 3 it 4 it 5 it 6	Facilities WWTP B NPDES Permit WWTP C NPDES Permit Performance Bond Insurance Requirements Industrial Pretreatment Ordinance Mercury Pollutant Minimization Program	

CITY OF STRONGSVILLE, OHIO WASTEWATER FACILITIES OPERATIONS, MAINTENANCE AND MANAGEMENT SERVICES

AGREEMENT

THIS AGREEMENT is made and entered into on this1st day of October, 2024, by and between the CITY OF

STRONGSVILLE, OHIO, an Ohio municipal corporation (hereinafter "City"), whose address is 16099 Foltz Parkway,

Strongsville, Ohio 44149 and VEOLIA WATER NORTH AMERICA - CENTRAL, LLC, a Delaware Limited Liability

Company (hereinafter "Company"), whose address is 53 State Street, 14th Floor, Boston, MA 02109.

WITNESSETH

WHEREAS, the City owns and is responsible for the operation and maintenance of the City's wastewater Facilities, which include but are not limited to two (2) wastewater treatment plants and nineteen (19) lift stations, and appurtenances (collectively, the "Facilities"); and

WHEREAS, the City desires to have the Facilities operated. maintained and managed in the most effective manner possible, while complying with all federal, state and local laws, rules and regulations; and

WHEREAS, the operation, maintenance and management of the Facilities require unique and specialized professional skills; and

WHEREAS, the City desires to maintain ownership of the Facilities and to contract with an organization that has the specialized professional skills and experience to operate, maintain and manage the Facilities; and

WHEREAS, the Company responded to the Request for Proposals issued by the City for the operation, maintenance and management of the Facilities; and

WHEREAS, the City has selected the Company, pursuant to the Request for Proposals and the Company's proposal to operate, maintain and manage the Facilities in accordance with the terms, conditions and provisions of this Agreement, in reliance on the Company's representations in its response to the Request for Proposals of its skill, expertise and past successful experience of supplying operation, maintenance and management services for facilities such as the Facilities; and

WHEREAS, the City desires to engage the services of the Company for the operation, maintenance and management of the Facilities and the Company desires to perform such services for the compensation provided herein; and

WHEREAS, the Company shall have executed prior to the commencement date the Performance Bond presented in Exhibit 4 hereto; and

WHEREAS, the Company shall have provided a certificate of insurance providing the insurance coverage required by Exhibit 5 hereto; and

WHEREAS, the City expects and desires that the relationship between the City and the Company will be a cooperative one, devoted to achieving the cost-savings goals of the City while also providing safe, economical and efficient services, meeting the needs of the current and future customers of the Facilities, maintaining the long-term integrity of the Facilities implementing capital improvements to the Facilities, and assuring safe and environmentally sound service while utilizing Prudent Industry Practices;

NOW, THEREFORE, in consideration of the mutual promises and covenants of the parties hereto contained in this Agreement, the receipt and sufficiency of which are hereby acknowledged, the City and the Company hereby agree as follows:

ARTICLE 1 - DOCUMENTS

1.01 Agreement Documents. The following Exhibits are attached hereto and made a part of this Agreement. In the event of a conflict or inconsistency between or among the Exhibits and sections of this Agreement, the parties hereto agree that the more stringent and restrictive requirements applicable to the Company shall govern and control; provided, however, that the Company shall at all times be able to rely on the limitations and conditions set forth in the sections of this Agreement as applicable, even if such limitations and conditions are not repeated in the Exhibits to this Agreement.

Exhibit 1	Facilities
Exhibit 2	WWTP B NPDES Permit
Exhibit 3	WWTP C NPDES Permit
Exhibit 4	Performance Bond
Exhibit 5	Insurance Requirements
Exhibit 6	Industrial Pretreatment Ordinance
Exhibit 7	Mercury Pollutant Minimization Program

This Agreement, together with the foregoing Exhibits, constitutes the entire agreement between the City and the Company with respect to the operation, maintenance and management of the Facilities which shall govern exclusively the obligations of the parties hereto, and supersedes all prior negotiations, representations, documents and agreements, written or oral, occurring prior to the date of this Agreement.

ARTICLE 2 - TERM OF AGREEMENT AND AUTHORIZED REPRESENTATIVES

- 2.01 Term of Agreement. The effective date of this Agreement shall be the Agreement Date. The term of this Agreement shall begin on the Commencement Date and, subject to earlier termination as provided herein, shall expire five years from the Commencement Date (the "Initial Term"). Not less than six (6) months prior to the expiration of the Initial Term, the City in its sole and absolute discretion may deliver to the Company written notice requesting that the Parties extend the Initial Term by an additional five (5) year term (the "Extended Term"). If the City desires to extend the Initial Term, the Parties shall cooperate in attempting to agree upon such modifications to this Agreement as may be required or desirable and to implement such modifications in accordance with the terms of this Agreement and Applicable Law. The City shall negotiate exclusively with the Company with respect to the Extended Term, as applicable, for a minimum period of sixty (60) days following delivery of the City's written notice; provided, however, that (1) neither Party shall be bound to an agreement for an Extended Term that such Party determines is not in its best interests and (2) each Party shall maintain the right to discontinue negotiations with respect to an Extended Term at the conclusion of the sixty (60) day period. The Initial Term, together with the Extended Term, if any, is referred to herein as the "Term".
- 2.02 <u>Authorized Representatives</u>. Prior to the commencement of work under this Agreement, each Party shall designate in writing an employee of the designating Party (the "Authorized Representative") who shall have the authority and responsibility to administer this Agreement. Each Party may change its Authorized Representative by written notice as provided in this Agreement. Each Authorized Representative shall be sent a copy of all notices required under this Agreement.

ARTICLE 3 - INTERPRETATION AND DEFINITIONS

3.01 <u>Interpretation</u>.

- A. As used in this Agreement, (i) the word "or" is not exclusive, (ii) the words "consent" and "approval" are synonymous and are deemed to be followed by the phrase "which shall not be unreasonably withheld or delayed," (iii) the words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation," (iv) any pronoun shall include the corresponding masculine, feminine and neuter forms, (v) words in the singular number shall include words in the plural and vice versa, unless the context of the usage of such term clearly indicates otherwise, and (vi) accounting terms that are used, but not otherwise defined herein, are to be construed and interpreted in accordance with "generally accepted accounting principles" and procedures in effect on the Agreement Date.
- B. Captions, headings and the Table of Contents in this Agreement, exclusive of Exhibits, are for convenience of

reference only and do not constitute a part of this Agreement or affect its meaning, construction or effect.

- C. Except as expressly stated to the contrary elsewhere herein, in computing the number of days for purposes of this Agreement, all days should be counted, including Saturdays, Sundays and legal holidays for the City; provided, however, that if the final day of any time period falls on a Saturday, Sunday or legal holiday for the City, then the final day shall be deemed to be the next day which is not a Saturday, Sunday or legal holiday for the City.
- 3.02 <u>Definitions</u>. For the purposes of this Agreement and Exhibits hereto, the following words and terms shall have the respective meanings set forth below applicable to both the singular and plural thereof. Unless otherwise specified in this Agreement and Exhibits hereto, all capitalized terms herein refer to described terms in this Agreement and Exhibits hereto.

Acceptable Disposal Site means for purposes of Residuals disposal, a lawful land application site, a sewage sludge incinerator, a sanitary landfill or other lawfully authorized disposal facility for non-hazardous waste which; (1) is located in the United States; (2) does not appear on any federal or state list of sites, including the National Priority List, which list is maintained for the purpose of designating landfills or other sites which are reasonably expected to require remediation on account of the release or threat of release of Hazardous Substances; (3) to the knowledge of the Company, is being operated at the time of disposal in accordance with Applicable Law, as evidenced by the absence of any regulatory sanctions or any significant enforcement actions with respect to material environmental matters; and (4) has committed by written agreement of the owner or operator to receive Residuals.

Addenda means the written or graphic instruments issued prior to the receipt of Proposals which clarify, correct, or change the Proposal requirements or the proposed Agreement and Exhibits thereto.

<u>Additional Services</u> means the additional services provided by the Company which are outside of the general scope of Services set forth in this Agreement but are consistent with the purposes hereof and the effective operation of City government.

<u>Administrative Consent Orders</u> means the administrative orders issued by governmental regulatory agencies exercising jurisdiction over the City's wastewater treatment activities.

<u>Affiliates</u> means the Company, and any corporation, partnership, limited liability company, joint venture or other entity controlled by, controlling or under common control with, directly or indirectly, the Company or any one of such entities.

<u>Agreement</u> means this Agreement for the Operation, Maintenance and Management of the Wastewater Facilities between the City and the Company, including the Exhibits hereto, as the same may be amended or modified from time to time in accordance herewith.

Agreement Date means the date when this Agreement has been executed by the City and the Company.

Amendment means any written amendment to this Agreement approved by both Parties.

Applicable Law means as of the Agreement Date (1) any federal, state or local law, code, or regulation, any Administrative Consent Order, any Consent Order or any agreement with any Regulating Entity having appropriate jurisdiction; (2) any formally adopted and generally applicable rule, requirement, determination, standard, policy, implementation schedule or order of any Regulating Entity having appropriate jurisdiction; (3) any established interpretation of law or regulation utilized by an appropriate Regulating Entity if such interpretation is documented by such regulatory body and generally applicable; and (4) any Governmental Approval, in each case having the force of law and applicable from time to time to (a) the siting, permitting, design, acquisition, construction, equipping, financing, ownership, possession, start-up, testing, operation, maintenance, repair, replacement or management of wastewater treatment systems; (b) the conveyance, treatment or discharge of Influent and Effluent to and from the Facilities; (c) the transfer, handling, processing, transportation or disposal of Residuals; (d) the air emissions from the Facilities; (e) the health and welfare of individuals at or visiting the Facilities; or (f) any other transaction or matter contemplated hereby (including, without limitation, any of the foregoing which pertain to sewer treatment, waste disposal, health, safety, fire, environmental protection, labor relations, building codes, the payment of prevailing or minimum wages and non-discrimination).

<u>Authorized Representative</u> means the representative of the City or the Company and any successor designated pursuant to Section 2.02 of this Agreement with the authority and responsibility to administer this Agreement.

<u>Billing Month</u> means each calendar month. Any computation made on the basis of a Billing Month shall be adjusted on a pro rata basis to take into account any Billing Month of less than the actual number of days in the month to which such Billing Month relates.

Biologically Toxic influent means Influent containing a biologically toxic substance or combination of substances (that individually may or may not be biologically toxic substances but which collectively quality as a biologically toxic substance) in sufficiently high concentrations so as to materially interfere with the biological processes necessary for the removal of the organic and chemical constituents of the Influent required to meet the performance guarantees.

<u>Buildings</u> means all buildings owned by the City related to operations, maintenance and management of the Facilities under this Agreement.

Capital Improvement means (a) replacement of an Equipment item or replacement of directly connected components of an Equipment item that has or have met or exceeded their useful life, unless such replacement was a result of the Company's failure to comply with the proper operation and maintenance of the Facilities, provided such replacement of components is anticipated to extend the useful life of the Equipment five (5) years or more as determined by the manufacturer or according to industry standards or, in the case of the replacement of the whole of the Equipment, such replaced whole is anticipated to have a useful life of five (5) years or more as determined by the manufacturer or according to industry standards, or (b) items for construction and placement of new facilities (e.g., piping, equipment, including material costs) and capital purchases that significantly improve operations and/or maintenance, aesthetics, long-term capital conditions or other aspects not generally associated with ongoing operation and maintenance; in each case with a construction, installation (including materials, site preparation costs, demolition and removal costs and labor to construct and install the Capital Improvements, but not for design assessment or oversight or other labor costs that are included in the Services to be provided under this Agreement) or purchase value in excess of the Capital Improvement Threshold Amount.

A "Capital Improvement" shall not (i) include operations activities; (ii) include preventive, predictive, routine and/or corrective maintenance activities; or (iii) allow the aggregation of unrelated items or activities to meet the Capital Improvement Threshold Amount, unless deemed eligible by the City prior to the commencement of the work activity.

Capital Improvement Threshold Amount means an amount equal to Ten Thousand Dollars (\$10,000).

<u>Change in Control</u> means change associated with the control of the Company as described in Section 15.06 of this Agreement.

<u>Change in Law</u> means any of the following acts, events or circumstances to the extent that compliance therewith materially increases or decreases the cost of performing or materially increases or decreases the scope of a Party's obligations hereunder;

- A. the adoption, amendment, promulgation, issuance, modification, or repeal of any Applicable Law, or the written changes in administrative or judicial interpretation of any Applicable Law, on or after the Agreement Date, unless such Applicable Law was, on or prior to the Agreement Date, duly adopted, promulgated, issued or otherwise officially modified or changed in interpretation, in each case in final form, to become effective without any further action by any Regulating Entity;
- B. the order or judgment of any Regulating Entity issued on or after the Agreement Date (unless such order or judgment is issued to enforce compliance with Applicable Law which was effective as of the Agreement Date) to the extent such order or judgment is not the result of willful or negligent action, error or omission or lack of reasonable diligence of the Company or of the City, whichever is asserting the occurrence of a Change in Law; provided, however, that the contesting in good faith or the failure in good faith to contest any such order or judgment shall not constitute or be construed as such a willful or negligent action, error or omission or lack of reasonable diligence; or
- C. the denial of an application for, a delay in the review, issuance or renewal of, or the suspension, termination,

or interruption of any Governmental Approval, or the imposition of a term, condition or requirement which is more stringent or burdensome than the Contract Standards in connection with the issuance, renewal or failure of issuance or renewal of any Governmental Approval, to the extent that such occurrence is not the result of willful or negligent action, error or omission or a lack of reasonable diligence of the Company or of the City, whichever is asserting the occurrence of a Change in Law; provided, however, that the contesting in good faith or the failure in good faith to contest any such occurrence shall not be construed as such a willful or negligent action, error or omission or lack of reasonable diligence.

It is specifically understood, however, that change in the nature or severity of the actions typically taken by a Regulating Entity to enforce compliance with Applicable Law which was effective as of the Agreement Date shall not constitute a "Change in Law".

Chemicals means all process chemicals used for wastewater treatment at the Wastewater Treatment Facilities.

<u>City</u> means the City of Strongsville, Ohio acting through the Department of Public Service.

City Default means an Event of Default by the City as described in Section 14.02 of this Agreement.

<u>City Indemnitee</u> means the individual or collective entity of the City, the Department and their respective officials, officers, employees, agents, representatives, consultants, contractor, and subcontractors.

<u>Commencement Date</u> means the date on which the Company takes responsibility for the day-to-day operation, maintenance and management of the Facilities under this Agreement, which date shall be October 1, 2024, and upon which the Initial Term begins, unless mutually agreed otherwise by the City and the Company.

Company means the entity that has entered into this Agreement with the City.

Company Default means an Event of Default by the Company as described in Section 14.01 of this Agreement. Consent

Order means an order of a court associated with the Facilities.

Contract Standards means the terms, conditions, methods, techniques, practices and standards imposed or required by: (1) Applicable Law; (2) the performance guarantees; (3) Prudent Industry Practices; (4) the Operation and Maintenance Manuals; (5) applicable manufacturers' instructions for maintenance and care of Equipment; (6) Required Insurance; and (7) any other standard, term, condition or requirement specifically provided in this Agreement to be observed by the Company.

Contract Year means the period commencing with the Commencement Date and ending at 12:00 a.m. midnight on the anniversary of the Commencement Date and, for each successive Contract Year thereafter, the period commencing on 12:00 a.m. midnight on the most recent anniversary of the Commencement Date and ending on 12:00 a.m. midnight of the next succeeding anniversary of the Commencement Date.

<u>Cost Substantiation</u> means documentation to substantiate costs of the Company for Additional Services, Capital Improvements or work performed under this Agreement on an emergency basis, all as set forth in Section 17.04 hereof.

County means Cuyahoga County, Ohio

<u>CPI Index</u> means the Consumer Price Index series for All Urban Consumers, Midwest Urban Area, All Items, Monthly, Not Seasonally Adjusted as published by the U.S. Department of Labor, Bureau of Labor Statistics, or any successor series (or a replacement index as mutually agreed upon by the Parties and which provides for the measurement of the Adjustment Factor pursuant to a specified, objective, external standard that is not linked to the output or efficiency of the Facilities or any portion thereof).

Daily Operation and Maintenance Log means the daily log as required by OEPA to be maintained on site.

Default Notice means written notice as described in Section 14.03 of this Agreement.

Department means the Department of Public Service of the City of Strongsville, Ohio.

<u>Effluent</u> means the Influent which has been received and treated by and discharged from the Wastewater Treatment Facilities as permitted by Applicable Law.

Encumbrance means any lien, lease, mortgage, security interest, charge, judgment, judicial award, attachment or encumbrance of any kind with respect to the Facilities.

EPA means the United States Environmental Protection Agency and any successor agency.

Equipment means all machinery, components, process-related equipment, property or assets (including pumps, bar screens, grit handling equipment, sludge handling equipment, chemical feed storage equipment and tank covers), parts and materials contained within the Facilities which are owned or leased by the City and in use as of the Commencement Date, or procured or provided on or after the Commencement Date by the Company or the City pursuant to this Agreement.

Event of Default means those items in Section 14.01 of this Agreement with respect to the Company and those items in Section 14.02 of this Agreement with respect to the City.

Extended Term means the five (5) year renewal term subsequent to the Initial Term as described in Section 2.01 hereof.

<u>Facilities</u> means all wastewater and ancillary facilities, Buildings and Grounds, Equipment, consumables, spare parts, and Fixed Assets, including subsurface structures such as pipes, in the Facilities existing at the Commencement Date or placed into service during the Term of this Agreement, including but not limited to, Capital Improvements, both when under construction and after being placed in service as part of the Facilities.

<u>Fixed Asset</u> means items that are not permanently installed, including computer and video equipment, office furniture, storage units, hand tools, hand-help communication devices, and laboratory equipment, and with a value of Five Thousand Dollars (\$5,000) or more.

Fixed Fee means the portion of the Service Fee described in Section 8.01 of this Agreement.

<u>Governmental Approvals</u> means all orders of approval, permits, licenses, authorizations, consents, certifications, exemptions, rulings, entitlements and approvals issued by a Regulating Entity of whatever kind and however described which are required under Applicable Law to be obtained or maintained by any Person with respect to the Services.

<u>Grounds</u> means the lands or areas indicated in this Agreement and Exhibits hereto as being furnished by the City upon which the Services are to be performed, including rights-of-way and easements for access thereto and such other lands furnished by the City which are designated for the use of the Company.

<u>Hazardous Residuals</u> means any portion of the Residual that constitutes or contains Hazardous Substances in such concentrations or volumes as to render the Residuals unable to be normally disposed of as contemplated in this Agreement.

Hazardous Substance means any flammable, explosive, corrosive or ignitable material, characteristic waste, listed waste, radon, radioactive material, asbestos, ureaformaldehyde foam insulation, polyclorinated biphenyls, petroleum and petroleum-based wastes, methane gas, hazardous materials, hazardous wastes, hazardous or toxic substances or related materials, mixtures or derivatives having the same or similar characteristics and effects, as described in, listed under, or regulated by various federal, state or local environmental statutes, including, without being limited to, the Comprehensive Environmental Response, Compensation and Liability Act of 1980 (42 S.S.C. § 9601, et seq.) ("CERCLA"), the Hazardous Materials Transportation Act (49 U.S.C.§ 1801, et seq.), the Emergency Planning and Community Right to Know Act (42 U.S.C.§ 11001, et seq.), the Resource, Conservation and Recovery Act (42 U.S.C.§ 6901, et seq.), the Toxic Substances Control Act (15 U.S.C.§ 2601, et seq.), the Clean Water Act (33 U.S.C.§ 1251, et seq.), the Clean Air Act (42 U.S.C.§ 7401, et seq.), the Safe Drinking Water Act (42 U.S.C.§ 300, et seq.), and including all amendments, modifications or successor legislation; or as such substances are described under any similar federal or state laws or regulations.

<u>Industrial Pretreatment Ordinance</u> means Chapter 1050 entitled Pretreatment Requirements of the Codified ordinances of the City of Strongsville and which is included in <u>Exhibit 6</u>.

Influent means all flows reaching the Wastewater Treatment Facilities from all connected sources, including residential, commercial, municipal and industrial sources, infiltration and inflows.

Initial Term means the five (5) year period starting from the Commencement Date of this Agreement.

<u>Legal Proceeding</u> means every action, suit, litigation, arbitration, administrative proceeding and other legal or equitable proceeding having a material effect on this Agreement or the performances of the Parties hereunder, and all appeals therefrom.

Lift Station means those lift stations and pump stations used to convey wastewater as referenced in Exhibit 1 hereto.

<u>Losses</u> is defined in Section 13.01 (A) of this Agreement.

Memorandum of Understanding means a written memorandum between the Parties providing for the clarification or interpretation of a provision of this Agreement or memorializing the outcome of a dispute resolved pursuant to Article 16 hereof.

Non-Specified Influent means influent entering a wastewater treatment facility that is characterized by (i) a solids or organic content that exceeds one or more of the design Influent parameters set forth in Exhibit 1 or (ii) Biologically Toxic Influent.

NPDES Permit means any National Pollutant Discharge Elimination System Permit issued from time to time by OEPA to the City with respect to the Facilities during the Term of this Agreement, including any revision or modification to such a permit. The NPDES Permits in effect as of the Agreement Date are: with respect to Plant B, NPDES Permit No. OHOO27570 issued by OEPA to the City and included in Exhibit 2; with respect to Plant C, NPDES Permit No. OH0033707, issued by OEPA to the City and included in Exhibit 3. For purposes of this Agreement, all of the provisions and requirements of these NPDES Permits shall be deemed to apply to the City and the Company except as specifically excluded by this Agreement.

OEPA means the Ohio Environmental Protection Agency or any successor agency.

Off-site means elsewhere other than on or at the Facilities.

On-site means on, at or within the Facilities.

Operational Change means an adjustment in routine operating procedures that will be made as a matter of routine practice by the Company to operate the Wastewater Treatment Facilities and the Lift Stations.

Operation and Maintenance Manuals means (a) the existing Wastewater Facilities Operations and Maintenance Manuals, (b) Equipment, supplier or vendor operation and maintenance manuals and (c) any new operation and maintenance manuals.

Operator of Record means the licensed operator designated to oversee the technical operation of a wastewater treatment facility and/or sewerage collection system in accordance with OEPA requirements.

Party or Parties means the parties to this Agreement.

<u>Permits</u> means all regulatory permits and licenses to which the Facilities and their operation are subject, including, without limitation, the NPDES Permits.

<u>Person(s)</u> means any natural or artificial entity including an individual, corporation, limited liability company, partnership, joint venture, association, joint-stock company, trust, unincorporated organization, or government, agency or authority (including federal, state, county, municipal or other local agency) or political subdivision thereof.

Project Manager has the meaning specified in Section 7.07 hereof.

Prudent Industry Practices means the methods, techniques, standards and practices which, at the time they are to

be employed and in light of the circumstances known or reasonably believed to exist at such time, are generally recognized and accepted as good operation, maintenance, repair, replacement and management practices in the municipal wastewater treatment industry as observed in the United States.

<u>Regulating Entity</u> means any federal, state, regional, county or local entity, court, agency or body, or subdivision thereof, having governmental or quasi-governmental authority and jurisdiction over the Facilities.

<u>Required Insurance</u> means the insurance required to be maintained by the Company as provided in Article 12 hereof and <u>Exhibit 5</u> hereto.

<u>Residuals</u> means any solid, semi-solid or liquid residue or sludge that is collected, produced or otherwise handled from the Facilities and disposed of Off-site. Residuals includes sludge produced at the wastewater treatment Facilities and grit, screenings and other materials removed from System Influent and Influent. System Influent, Influent, and Effluent shall not constitute Residuals.

Request for Proposals means the Request for Proposals for Wastewater Facilities Operations and Maintenance Services issued by the City dated August 16, 2024 and any Addenda or amendments thereto.

Service Fee means the fee as described in Section 8.01 of this Agreement.

<u>Services</u> means the operation, maintenance, repair and management services provided by the Company pursuant to this Agreement, including, without limitation, wastewater treatment, customer services, as well as the provision of project management for the Facilities and all required ancillary activities in accordance with the requirements of the Request for Proposal, this Agreement and the Exhibits hereto.

SOP means a standard operating procedure.

State means the State of Ohio and all of its relevant administrative, contracting and regulatory agencies and offices.

<u>Subcontractor</u> means an individual or entity having a direct contract with the Company or with any other subcontractor for the performance of a part of the Services.

<u>Substantial Completion</u> means that the City has issued a certificate of substantial completion for a Capital Improvement.

System Influent means all wastewater, inflow/infiltration entering the Wastewater Treatment Facilities.

Term has the meaning specified in Section 2.01 of this Agreement.

<u>Termination Date</u> means the date on which this Agreement terminates and is no longer in force or effect, which date shall be the end of the Initial Term or the Extended Term, as applicable, unless earlier terminated as provided herein.

<u>Transition Period</u> means the period between the Agreement Date and the Commencement Date.

<u>Uncontrollable Circumstances</u> means any act, event or condition that is beyond the reasonable control of the Party relying thereon as justification for not performing an obligation or complying with any condition required of such Party under this Agreement, and that materially interferes with or materially increases the cost of performing its obligations hereunder (other than payment obligations, including any payment obligations the City may have to reimburse or compensate the Company for an Uncontrollable Circumstance), to the extent that such act, event or condition is not the result of the willful or negligent act, error or omission, failure to exercise reasonable diligence, or breach of this Agreement on the part of such Party.

<u>Upset</u> means an exceptional incident in which there is unintentional and temporary noncompliance with technology based permit effluent limitations because of factors beyond the reasonable control of the permittee. An "Upset" does not include noncompliance to the extent caused by operational error, improperly designed treatment facilities, inadequate treatment facilities, lack of preventative maintenance, or careless or improper operation.

<u>Utilities</u> means any and all utility services and installations whatsoever (including natural gas, electricity, telephone, and telecommunications), and all piping, wiring, conduits, and other fixtures of every kind whatsoever related thereto or used in connection therewith.

<u>Vehicles</u> means all cars, trucks, vans or other modes of transportation that require a license plate used in connection with operation of the Facilities for transporting people or things or used for other necessary functions in the operation or maintenance of the Facilities.

Wastewater Treatment Facilities means those wastewater treatment facilities referenced in Exhibit 1 hereto.

ARTICLE 4 - REPRESENTATIONS AND WARRANTIES

- 4.01 Representations and Warranties of the City. The City represents and warrants to the Company that:
 - A. <u>Organization and Powers</u>. The City is a municipal corporation duly organized and existing under the laws of the State and is duly authorized and empowered to enter into and perform its obligations pursuant to the terms and provisions of the Agreement and to execute all documents related hereto to which it is a party.
 - B. <u>Due Authorization and Binding Obligations</u>. The execution and delivery of this Agreement has been duly authorized by all necessary actions of the City, none of which actions have been rescinded or otherwise modified. This Agreement is a legal, valid and binding obligation of the City, enforceable against the City in accordance with its terms, except as the same may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally or by general principles of equity. All payments to be made by the City pursuant to this Agreement are subject to appropriation by the City's legislative body, and the City agrees to use all reasonable efforts to include such appropriation in each of the City's budgets during the Term.
 - C. No Conflict. The execution and delivery of this Agreement and any other documents related hereto to which the City is a party, the consummation by the City of the transactions contemplated hereby and thereby, and the fulfillment by the City of the terms and conditions hereof and thereof do not and will not (i) conflict with, violate or result in a material breach of any constitution, law, corporate charter or by-laws applicable to the City or (ii) conflict with, violate or result in the material breach of any terms or conditions of any order, judgment or decree, or any restriction, contract, agreement or instrument to which the City is not a party or by which it is bound, or constitute a default under any such restriction, contract, agreement or instrument.
 - D. No Litigation. There is no action, suit, claim, investigation or proceeding, at law or in equity, before or by any court or Regulating Entity, pending or, to the best of the City's knowledge, threatened against the City, in which an unfavorable decision, ruling or finding could reasonably be expected to have a material adverse effect on the execution and delivery of this Agreement by the City or the validity, legality or enforceability of this Agreement against the City, or any other agreement or instrument entered into by the City in connection with the transactions contemplated hereby, or on the ability of the City to perform its obligations hereunder or under any such other agreement or instrument.
 - E. No Default. The City is not in default under, and, to the best of its knowledge, no condition exists that with notice or lapse of time or both would constitute a default under, (i) any mortgage, loan agreement, lease, lease purchase, indenture or evidence of indebtedness for borrowed money to which the City is a party or by which any material amount of the assets of the City is bound that would materially affect the City's entering into, or performance of, this Agreement, or (ii) any judgment, order, injunction, rule, regulation or other judicial or administrative mandate of any court, arbitrator or governmental agency or instrumentality, which default or potential default could reasonably be expected to have a material adverse effect on the City's entering into, or performance of, this Agreement.
 - F. Compliance With Law. The City is presently in compliance with all Applicable Laws in connection with the operation and maintenance of the Facilities and to the best of the City's knowledge, no event has occurred which would constitute reasonable grounds for a claim that noncompliance has occurred or is occurring. There are no outstanding complaints, orders, citations, notices or orders of violation or noncompliance issued to the City relating to the operations, maintenance or condition of the Facilities, which could reasonably be expected to have a material adverse effect on the City's entering into, or performance of, this Agreement.

- 4.02 Representations and Warranties of the Company. The Company represents and warrants to the City that:
 - A. <u>Due Authorization and Binding Obligation</u>. The execution and delivery of this Agreement has been duly authorized by all necessary corporate actions of the Company, none of which actions have been rescinded or otherwise modified. The Company has full power and authority to execute and deliver this Agreement and to perform its obligations hereunder. No consents, approvals or permits are required for the execution and delivery of this Agreement or the performance of the terms and provisions herein, or, if any such consents, approvals or permits are required, they have been, or will be prior to the Commencement Date, obtained. This Agreement is a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as the same may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally o by general principles of equity.
 - B. No Conflict. The execution and delivery of this Agreement and any others documents related hereto to which the Company is a party, the consummation by the Company of the transactions contemplated hereby and thereby, and the fulfillment by the Company of the terms and conditions hereof and thereof do not and will not (i) conflict with or violate any of the terms of the Company's formation documents or, to the Company's knowledge, and Applicable Laws, (ii) conflict with, violate or result in a breach of any of the terms and conditions of, or result in the acceleration of any indebtedness or obligations under, any order, judgment or decree, or any restriction, contract, agreement, obligation or instrument to which the Company is now a party or by which it is bound or to which any property of the Company is subject, or constitute a default under any such order, judgment, decree, restriction, contract, agreement, obligation or instrument, or (iii) conflict with, violate or result in or constitute a default under, a breach of or grounds for termination of, any Permits or other Governmental Approvals to which the Company is a party or by which the Company may be bound, or result in the violation by the Company of any Applicable Laws to which the Company or any assets of the Company may be subject, which would materially adversely affect the transactions contemplated herein.
 - C. No Litigation. There is no action, suit, claim, investigation or proceeding, at law or in equity, before or by any court or Regulating Entity, pending or, to the best of the Company's knowledge, threatened against the Company, in which an unfavorable decision, ruling or finding could reasonably be expected to have a material adverse effect on the execution and delivery of this Agreement by the Company or the validity, legality or enforceability of this Agreement against the Company, or any other agreement or instrument entered into by the Company in connection with the transactions contemplated hereby, or on the ability of the Company to perform its obligations hereunder or under any such other agreement or instrument.
 - D. No Default. The Company is not in default under, and, to the best of its knowledge, no condition exists that with notice or lapse of time or both would constitute a default under, (i) any mortgage, loan agreement, lease, lease purchase, indenture or evidence of indebtedness for borrowed money to which the Company is a party or by which any material amount of the assets of the Company is bound that would materially affect the Company's entering into, or performance of, this Agreement, or (ii) any judgment, order, injunction, rule, regulation or other judicial or administrative mandate of any court, arbitrator or governmental agency or instrumentality, which default or potential default could reasonably be expected to have a material adverse effect on the Company's entering into, or performance of, this Agreement.
 - E. <u>Ability to Perform</u>. The Company is financially solvent. The Company and each of its employees, agents, subcontractors and contractors are competent to perform the services required under this Agreement.
 - F. Agreement Feasible. The Company has examined and analyzed the provisions and requirements of this Agreement. The Company understands the nature of the services required under this Agreement. From its own analysis, the Company has satisfied itself as to the nature of all things needed for its performance of this Agreement, the general and special conditions, and all other matters which in any way affect its performance of this Agreement. The time available to the Company for such examination, analysis and preparation was adequate, This Agreement is feasible of performance in accordance with all of its provisions and requirements.
 - G. <u>Technical Knowledge</u>. The Company has, or will have as of the Commencement Date, adequate capacity, technical knowledge and qualified employees to fulfill all of its obligations, covenants and agreements pursuant to the terms of this Agreement.

- H. <u>Compliance With Law</u>. The Company is presently in compliance with all Applicable Laws, and to the best of the Company's knowledge, no event has occurred which would constitute reasonable grounds for a claim that noncompliance has occurred or is occurring.
- I. Opportunity to Review and Inspect. Prior to the Agreement Date, the Company had the opportunity to review and inspect Governmental Approvals and permit applications regarding the Facilities and to meet with and ask questions of an receive answers from representatives of the City connected with the Facilities. The Company's Proposal was based primarily upon those reviews and inspections and supplemented by the City's information so provided.
- J. <u>Pledge of Credit; No Impairment</u>. The Company shall not pledge the City's credit or the revenues generated by the System or make the City a guarantor of payment or surety for any contract, debt, obligation, judgement, lien or any form of indebtedness. The Company further warrants and represents that it has no obligation or indebtedness that would impair its ability to fulfill the terms of this Agreement.
- K. Accuracy of Company Representations. The representations made by the Company in the Company's Proposal and in all other information and documentation submitted to the City by the Company were true and accurate as of the date they were made and are true and accurate as of the Agreement Date. The representations made by the Company in the Company's Proposal did not contain any material misrepresentations or omissions of any material facts as of the date that they were made, and the representations made therein, except to the extent such representations are modified in this Agreement, do not contain any material misrepresentations or omissions of any material facts as of the Agreement Date.
- L. <u>Financial Condition</u>. The Company has not filed, nor have creditors of the Company filed, any type of bankruptcy proceeding.
- M. <u>Approvals</u>. The Company has obtained and is in compliance with all the Governmental Approvals which are required for the Company to perform its obligations pursuant to this Agreement.
- N. <u>Prior Experience</u>. The Company has trained personnel who will be assigned to perform the Services required under this Agreement and such Company personnel have experience operating and maintaining facilities similar to the Facilities in accordance with Prudent Industry Practices.
- O. <u>Employment Matters</u>. During the Term of this Agreement, the Company agrees to observe all municipal ordinances, state laws, and federal laws related to labor and employment matters that are applicable.
- P. Representations and Warranties Generally. No statement, representation or warranty by the Company in this Agreement, including the Exhibits hereto, contains any untrue statement of material fact, or, to the best of the Company's knowledge, omits to state any material fact, necessary to make such statements, representations and warranties not misleading.
- 4.03 <u>Materiality of Representations</u>. The representations and warranties enumerated in Sections 4.01 and 4.02 hereof are material for purposes of this Agreement, and misrepresentations are grounds for an Event of Default hereunder.

ARTICLE 5 - SYSTEM DESCRIPTION

- 5.01 General. A description of the Facilities is attached hereto as Exhibit 1.
- 5.02 Facilities Condition Confirmation.
 - A. <u>Inspection of the System</u>. The Company acknowledges that: (i) the Company's agents and representatives have visited, inspected and observed the above-ground portion of the Facilities, and its design and physical condition relevant to the obligations of the Company pursuant to this Agreement, including operating conditions, roads, utilities, topographical conditions and historical Influent conditions; (ii) the Company's agents and representatives have reviewed information available relating to the below-ground portion of the facilities relating to its design and physical condition relevant to the obligations of the Company pursuant

to this Agreement; (iii) the Company's agents and representatives have reviewed the scope of services for the Facilities; (iv) the Company is familiar with all local conditions which may be material to the Company's performance of its obligations under this Agreement (including, without limitation, transportation; season, climate and ambient air; access, availability, handling, storage and disposal of materials, supplies and Equipment; and availability and quality of labor and utilities); (v) the Company has received and reviewed records and information pertaining to the Facilities that is has deemed necessary to receive and review for the purposes of entering into and performing this Agreement; and (vi) based on the foregoing, the Company is of the opinion that the Facilities can be managed, operated, maintained, repaired and replaced, in accordance with the terms and conditions and limitations hereof.

B. "As-Is" Condition of the Facilities. Based on its review and inspection of the Facilities and the scope of services for the Facilities as described in paragraph (A) above and other inquiries and investigations made by the Company prior to the Agreement Date, which the Company acknowledges to be sufficient for this purpose, the Company assumes the risk of the adequacy and sufficiency of the Facilities and the existing, "as- is" condition of the Facilities to the extent such condition may affect the ability of the Company to comply with Applicable Law, or otherwise comply with all of the other terms and provisions of this Agreement. The Company agrees that the physical condition of the Facilities which exists as of the Agreement Date or which may be revealed during the performance hereof shall not be an Uncontrollable Circumstance.

ARTICLE 6 - OPERATIONS, MAINTENANCE AND MANAGEMENT

6.01 General.

- A. Ownership of System. The Facilities and all Capital Improvements shall be owned by the City throughout and following the Term of this Agreement, and the Company shall have no ownership interest therein or in any revenues generated by the Facilities. During the Term hereof, the City shall provide the Company with all required access and the Company may enter upon, occupy, and use the Facilities to operate, maintain, repair, replace and manage the Facilities.
- B. Operation and Management Responsibility. Commencing on the Commencement Date, the Company shall operate and manage the Facilities on a twenty-four (24) hour per day, seven (7) day per week basis, and shall collect, receive and treat Influent, subject to the terms and conditions of this Agreement, produce and discharge Effluent, handle, transport and dispose of Residuals and operating wastes, provide all information necessary to each Regulating Entity, and otherwise operate and manage the Facilities so as to comply with Applicable Law and other terms and conditions of this Agreement.
- C. <u>Subcontractors</u>. The City shall have the right to require prior notice of all Subcontractors to be hired by the Company for performing services on behalf of the Company in carrying out its obligations hereunder (other than administrative, financial, accounting or legal services). The City reserves the right to object to the hiring of any Subcontractor subject to Section 16.01 of this Agreement. The Company shall maintain direction and control of all Subcontractors, and shall promptly pay all amounts owing and due to all Subcontractors retained, engaged, employed or directed by the Company to provide materials or services to the Facilities. If a Subcontractor has not satisfactorily performed under other contracts with the City, the City may request that the Company not use such Subcontractors to perform services under this Agreement.
- 6.02 <u>Exclusions</u>. The NPDES Permit Storm Water Pollution Prevention requirements contained in each permit and addressed in Part IV Storm Water Pollution Prevention Plans, Part V Numeric Effluent Limitations and Monitoring Requirements and Part VI Other Storm Water Requirements, Definitions and Authorizations are specifically excluded from Company's operations, maintenance and management responsibilities and will be performed by the City.
- 6.03 Operation of Facilities. On and after the Commencement Date and for the duration of the Term, the Company shall operate the Facilities in compliance with the terms of this Agreement. The Company's obligations pursuant to this Agreement also shall include but not be limited to:
 - A. Maintaining and revising as necessary the Operation and Maintenance Manuals and the written SOPs to the extent necessary to supplement or refine procedures provided in the Operation and Maintenance Manuals, or to describe operations practices not specified in such manuals in order to facilitate and clarify operation procedures and techniques.

- B. Responding in a timely manner to all consumer complaints and comments relative to wastewater treatment and discharge, including odor complaints. Responses shall be made and documented in writing, all complaints and responses shall be kept on file and available for public viewing at the Facilities and a copy of such complaints and responses shall be filed with the City.
- C. Consulting with the City prior to responding to inquiries from the media regarding the Facilities, the City or the Company with respect to the Facilities.

6.04 Maintenance of Facilities.

- A. General. The obligations of the Company under this Section 6.03 are intended to assure that the machinery, Equipment, Fixed Assets, Buildings and Grounds and improvements constituting the Facilities are properly and regularly maintained, repaired and replaced in order to preserve their long term reliability, durability and efficiency, and in a way such that when the Facilities are returned to the City at the end of the Term, they are in a condition similar to that which existed on the Commencement Date absent reasonable wear and tear given the age and environment of the Facilities and which with reasonable and routine efforts, does not require the City to undertake a significant overhaul or immediate replacements in order to continue to provide reasonably priced and efficient wastewater treatment.
- B. Ordinary Maintenance. The Company shall perform and record all normal and ordinary maintenance of the machinery, Equipment, Buildings and Grounds, improvements and all other property constituting the Facilities, shall keep the Facilities in good working order, condition and repair, in a neat and orderly condition, and shall maintain the aesthetic quality of the Facilities as originally constructed. The Company shall provide or make provisions for all labor, materials, supplies, equipment, spare parts, consumables and services which are necessary for the normal and ordinary maintenance of the Facilities and shall conduct predictive, preventative and corrective maintenance of the Facilities.
- C. Repair and Maintenance of Buildings and Grounds. The Company shall keep the Building and Grounds in a neat, clean, functional and orderly condition. In addition, the Company shall provide all landscaping services to maintain the Buildings and Grounds in an aesthetically attractive manner and shall maintain and routinely repair (but not replace) the Buildings and Grounds' parking lots, roadways and walkways subject to other provisions of this Agreement. The City shall be responsible for snow clearing and deicing all parking lots and roadways.
- D. Major Maintenance, Repair and Replacements. The Company shall perform all major maintenance, repairs and replacement of the machinery, equipment, structures, improvements and all other property constituting the Facilities during the Term of this Agreement without limitation, unless it constitutes a Capital Improvement, in which event the City shall have the approval rights and provide the funding as set forth in Article 9 hereof. The Company shall prepare and submit to the City a five (5) year forecast of the major repair and replacement activities that the Company believes needs to be performed at the Facilities during such five (5) year period to keep the Equipment in good working condition and repair so as to be able to properly perform the Services. To the extent any repair or replacement is required as a result of Uncontrollable Circumstances, the City shall pay the costs thereof.
- E. Residential Grinder Pump Repairs. Company shall perform minor repairs of residential grinder pumps which are currently being maintained by the City, provided that the Company is granted access by the residential property owner to make said repairs and provided that the residential property owner agrees to hold the Company harmless for claims relating to such pump repair. The scope of the work shall be limited to the grinder pump sump up to but not including the discharge piping slip connection and will include repair or replacement of the pump, discharge piping, check valve and float switch but shall not include the electric supply to the pump. City shall be responsible for furnishing replacement pumps for any such pump that cannot be repaired.
- F. Replacements Constituting Capital Improvements. The Company shall bear the cost and expense of all maintenance, repairs and replacements required under this Article 6, except for any replacements that constitute Capital Improvements as defined herein or needed as a result of any Uncontrollable Circumstance. In those instances where the failure of the Company to properly operate and maintain results in the replacement of Equipment (the replacement of which would have constituted a Capital Improvement if not for such failure), the Company shall share in the expense of the replacement in a proportion to be determined by the Department based upon the Company's lack of performance.

- G. <u>Chemicals</u>. The Company shall, as part of the Service Fee, arrange for the supply of and pay the bills for all required process chemicals and negotiate any and all contracts or agreements for the supply of chemicals.
- H. <u>Utilities.</u> The Company shall, as part of the Service Fee, have the responsibility to arrange for the supply of and pay the bills for all utilities required by the Facilities including but not limited to, electricity, natural gas, water and telephone.
- 6.05 Odor Control. The Company shall operate and maintain the Facilities in a manner that minimizes odors at the Facilities boundaries.
- 6.06 Compliance With Applicable Law.
 - A. <u>Compliance Obligation</u>. The Company shall perform the Services in accordance with Applicable Law, and shall require all Subcontractors to comply with Applicable Law.
 - B. Sampling, Testing and Laboratory Work. The Company shall perform and provide all sampling, laboratory testing and analyses, and quality assurance and quality control procedures and programs required by OEPA. All testing laboratories shall follow acceptable laboratory procedures as approved by OEPA and EPA, as applicable, and be certified for the applicable test. All sampling and test data shall be available for review by, and reported to, the City in accordance with Section 6.18 hereof. The Company assumes the risk of incorrect sampling, testing and laboratory work and any consequences thereof or actions taken or corrections needed based thereon for laboratory services it or its Subcontractors provide, both as to failures to detect and as to false detections.
 - C. Investigations of Noncompliance. In connection with any actual or alleged event of noncompliance with Applicable Law, the Company shall, in addition to any other duties which Applicable Law may impose: (1) fully and promptly respond to all inquiries, investigations, inspections, and examinations undertaken by any Regulating Entity; (2) attend all meetings and hearings required by any Regulating Entity; (3) provide all corrective action plans, reports, submittals and documentation required by any Regulating Entity; (4) in conjunction with the City, communicate in a timely and effective manner with the general public as to the nature of the event, the impact on the public, and the nature and timetable for the planned remediation measures; (5) immediately upon receipt thereof, provide the City with a true, correct and complete copy of any written notice of violation or noncompliance with Applicable Law, and true and accurate transcripts of any verbal notice of noncompliance with Applicable Law, issued or given by any Regulating Entity; and (6) assist the City with the defense of any Legal Proceeding arising out of a noncompliance event. The Company shall furnish the City with an immediate written notice describing the occurrence of any event or the existence of any circumstance which does or may result in any such notice of violation or noncompliance to the extent the Company has knowledge of any such event or circumstance, and of any Legal Proceeding alleging such noncompliance.
 - D. Fines, Penalties and Remediation. Except to the extent excused by Uncontrollable Circumstances, in the event that the Company or any Subcontractor fails at any time to comply with Applicable Law, the Company shall, without limiting any other remedy available to the City upon such an occurrence and notwithstanding any other provision of this Agreement; (1) immediately correct such failure and resume compliance with Applicable Law; (2) indemnify and hold harmless the City from any Loss resulting therefrom; (3) pay or reimburse the City for any resulting damages, fines, assessments, levies, impositions, penalties or other charges; (4) make all changes in operating and management practices which are necessary to assure that the failure of compliance with Applicable Law will not recur; and (5) comply with any corrective action plan filed with or mandated by any Regulating Entity in order to remedy a failure of the Company to comply with Applicable Law.
 - E. No Nuisance Covenant. The Company shall keep the Facilities neat, clean and litter-free and ensure that the operation of the Facilities does not create any odor, litter, noise, fugitive dust, vector, excessive light or other adverse environmental effects constituting, with respect to each of the foregoing, a nuisance condition under Applicable Law. Should any such nuisance condition occur which is not caused by Uncontrollable Circumstances, the Company shall immediately remedy the condition, pay any fines or penalties imposed on the City or the Company as a result thereof, make all improvements and changes in operating and management practices necessary to prevent a recurrence of the nuisance condition, and

indemnify and hold harmless the City from any third party Loss relating thereto in the manner provided in Section 13.01 hereof.

- 6.07 <u>Industrial Pretreatment</u>. The Company shall continue the implementation of the existing Industrial Pretreatment Ordinance requirements as presented in <u>Exhibit 6</u> and in compliance with OEPA requirements.
- Mercury Pollutant Minimization Program. The Company shall continue the implementation of the existing Mercury Pollutant Minimization Program in compliance with OEPA requirements, NPDES Permit requirements and the Mercury Pollutant Minimization Program requirements as presented in Exhibit 7. Any additions or modifications to the Mercury Pollutant Minimization Program required by OEPA or EPA and/or recommended by the Company shall be made by the Company and incorporated into Exhibit 7 hereto after being approved by OEPA and the City.

6.09 Residuals Management.

- A. <u>Disposal of Residuals</u>. Except as provided in Sections 6.09(E) and 6.09(F) of this Agreement, the Company shall have full responsibility for and shall bear all costs and expense associated with disposal of Residuals. The Company shall ensure that the Residuals disposal is performed in accordance with Applicable Law. All Residuals shall be disposed of at an Acceptable Disposal Site. The Company shall provide evidence satisfactory to the City, from time to time as requested by the City, that the disposal location conforms with the requirements of this Section 6.09.
- B. Residuals Disposal Information. The Company shall keep and maintain such logs, records, manifests, bills of lading or other documents pertaining to Residuals as are necessary or appropriate to comply with Applicable Law and to monitor and confirm compliance by the Company with the requirements of this Article 6, and shall collect and promptly provide the City with a copy of all weights and measures data and information relating to quantities of Residuals generated, transported and disposed of hereunder. The Company shall prepare and file, on behalf of the City and subject to the City's review, approval and, as applicable, signature, all reports and records required to be prepared and filed with any Regulating Entity pursuant to the 503 Regulations and other Applicable Law with respect to all Residuals. The City, only to the extent required by Applicable Law, shall sign all permits, manifests or similar documents required for the transportation or disposal of Facilities Residuals.
- C. <u>Protocol for Hazardous Residuals</u>. The Parties acknowledge that, notwithstanding the administration and enforcement of the Industrial Pretreatment Program, Influent may from time to time contain materials that cause Residuals to constitute Hazardous Substances. The Company shall identify and hold harmless the City, and provide for the proper disposal of said Hazardous Substances.
- D. Notification and Reporting. In the event Hazardous Residuals are identified, whether On-Site or Off-site, the Company, after first notifying the City, shall be responsible for fulfilling all notification and reporting requirements established by Applicable Law and shall prepare a memorandum evidencing such notification and reporting and provide copies thereof to the City, along with any documents provided to the relevant Regulating Entity regarding such Hazardous Residuals. The City shall have the right to witness and to document any action taken by the Company with respect to Hazardous Residuals.
- E. Off-site Disposal and On-Site Remediation. The Company, in the most expeditious manner possible, shall cause any Hazardous Residuals to be removed from the Facilities and transported to and disposed of at an Off-site disposal facility authorized to receive and dispose of such Hazardous Residuals under Applicable Law, and shall take all necessary On-site remediation steps. The Company shall only be responsible for the costs to perform any identification, testing, removal, temporary storage, On-site remediation, or Off-site transportation and disposal of Hazardous Residuals that are incurred due to the failure of the Company to comply with terms of the Agreement.
- F. Off-site Remediation. If Hazardous Residuals are transported and disposed of Off-site, and the Company can show that it had no knowledge that it was transporting and disposing of Hazardous Residuals, the Company shall only be responsible for the costs associated with any necessary Off-site remediation measures that are the result of the Company's failure to comply with the terms of the Agreement. Each Party shall bear any Losses it may incur resulting from any Legal Proceeding originated by a third party arising from the Off- site transportation or disposal of Hazardous Residuals; provided, however, that the Company shall indemnify, defend and hold harmless the City from any such Loss to the extent the Loss would have been avoided and the Company complied with the terms of the Agreement or to the extent the

Company had, or should have had, knowledge that it was transporting or disposing of Hazardous Residuals.

- G. Risks Assumed by the Company. The Company shall bear the transportation and disposal risks associated with the disposal of Residuals, including all risks relating to (1) the prices and business terms and conditions which may prevail from time to time over the Term in the market for the transportation and disposal of Residuals; (2) the availability and cost to the Company from time to time of Acceptable Disposal Sites for Residuals; (3) the availability and cost to the Company from time to time of transportation and disposal services for Residuals; and (4) the performance or non-performance by any Subcontractor of the Company engaged in the disposal of Residuals.
- H. Reimbursement of Costs. Any costs paid by the Company to unrelated third parties and chargeable to the City under this Section 6.09 shall be reimbursed to the Company as a Reimbursable Costs Charge in accordance with Section 8.01 hereof.

6.10 Operation and Maintenance Manuals.

- A. Company Responsibilities. The Services shall be performed substantially in compliance with the Operation and Maintenance Manuals and the SOPs. The Company shall keep these documents current, shall make all appropriate updates, supplements or revisions thereto and shall make such documents available to the City upon request. Any review of the operation and maintenance documents shall not: (1) relieve the Company of any of its responsibilities under this Agreement; (2) be deemed to constitute a representation by the City that operating the Facilities pursuant to the operation and maintenance documents will cause the Facilities to be in compliance with this Agreement or Applicable Law; or (3) impose any liability upon the City.
- B. <u>Supplements for Capital Improvements</u>. The Company shall prepare supplements and revisions to the existing Operation and Maintenance Manuals, SOPs, and other documents which are required due to the design, construction and installation of all Capital Improvements based upon manuals provided by the engineers and/or contractors for the Capital Improvements. The cost and expense of all such supplements and revisions shall be borne by the Company, except with respect to supplements and revisions necessitated by Capital Improvements directed by the City or required by a a Change in Law or other Uncontrollable Circumstance.

6.11 Equipment, Fixed Assets, Consumables and Spare Parts.

- A. <u>Equipment</u>. All Equipment provided by the City or the Company on or after the Commencement Date, including any Equipment on order by the Company or the City, shall be deemed to be owned by the City and shall remain a part of the Facilities upon the termination or expiration of this Agreement.
- B. <u>Fixed Assets</u>. All Fixed Assets provided by the City or the Company on or after the Commencement Date, including any Fixed Assets on order by the Company or the City, shall be owned by the City and shall remain a part of the Facilities upon the termination or expiration of this Agreement.
- C. Consumables. On or about the Commencement Date, the City and Company shall prepare an itemized inventory and valuation of all consumables in stock within the Facilities which are to be transferred to the Company that are jointly determined to be necessary for the operation and maintenance of the Facilities. In a like manner, on or about thirty (30) days prior to the end of the Term, or the Termination Date, an itemized inventory and valuation of all consumables in stock within the Facilities necessary and dedicated for the maintenance of the Facilities which are to be transferred to the City shall be performed by the City and Company. The final valuation amount shall be compared to the initial valuation amount as adjusted by the CPI Index. Consumables that are no longer usable for systems and Equipment shall be excluded from the initial and final valuation amount. The amount of any valuation shortfall or excess shall be paid by the Company to the City, or by the City to the Company, as appropriate, on or before the end of the Term or the Termination Date.
- D. <u>Spare Parts</u>. On or about the Commencement Date, the City and the Company shall prepare an itemized inventory and valuation of all spare parts in stock within the Facilities which are to be transferred to the Company that are jointly determined to be necessary for the operation and maintenance of the Facilities. The value of each inventory item turned over to the Company shall be recorded as of the Commencement

Date. In a like manner, on or about thirty (30) days prior to the end of the Term or the Termination Date, an itemized inventory and valuation of all spare parts in stock within the Facilities necessary and dedicated for the operation and maintenance of the Facilities which are to be transferred to the City shall be performed by the City and the Company. The final valuation amount shall be compared to the initial valuation amount as adjusted by the CPI Index. Spare parts that are no longer usable for Facilities shall be excluded from the initial and final valuation amount. The amount of any valuation shortfall or excess shall be paid by the Company to the City or by the City to the Company, as appropriate, on or before the end of the Term or the Termination Date.

6.12 Emergency Response.

- A. Emergency Plan. The Company shall provide the City with a draft plan of action to be implemented in the event of an emergency, including but not limited to, fire, weather, environmental, health (e.g., a pandemic), safety, breach of security and other potential emergency conditions. The plan shall; (i) provide for appropriate notifications to the City and all other Regulating Entities having jurisdiction and for measures which facilitate coordinated emergency response actions by the City and all such other appropriate Regulating Entities; (ii) specifically include spill prevention and response measures; (iii) address the discharge of Effluent which does not comply with Applicable Law; (iv) assure the timely availability of all personnel required to respond to any emergency; and (v) be prepared and updated in accordance with Applicable Law throughout the Term. The emergency plan shall be reviewed by the Parties annually as part of the review of the annual operations report and updated when necessary.
- B. Emergency Action. Notwithstanding any requirement of this Agreement requiring City approval or consent to reports or submittals, if at any time the Company determines in good faith that an emergency situation exists such that action must be taken to protect the safety of the public or its employees, to protect the safety or integrity of the Facilities, or to mitigate the immediate consequences of an emergency event, then the Company shall take all such action it deems in good faith to be reasonable and appropriate under the circumstances. As promptly thereafter as is reasonable, the Company shall notify the City of the event, and the Company's response thereto, at an emergency phone number from a list supplied by the City. The cost of the Company's response measure shall be borne by the Company except to the extent the emergency event was caused by an Uncontrollable Circumstance, in which case the City shall bear the cost.

6.13 Safety of Persons and Property.

- A. Safety. The Company shall prepare a safety plan in accordance with and shall maintain the safety of the Facilities at a level consistent with generally accepted standards. Without limiting the foregoing, the Company shall; (1) take all reasonable precautions for the safety, maintenance, and repair of, and provide all reasonable protection to prevent, repair and replace damage, injury or loss by reason of our related to the operation or maintenance of the Facilities to, (i) all employees working at the Facilities and all other persons who may be involved with the operation, construction, maintenance, repair and replacement of the Facilities, (ii) all visitors to the Facilities, (iii) all materials and Equipment under the care, custody or control of the Company on the Buildings and Grounds, (iv) other property constituting part of the Facilities and (v) property owned by the City; (2) establish and enforce all reasonable safeguards for safety and protection, including (i) posting danger signs and other warnings against hazards and (ii) promulgating safety regulations; (3) give all notices and comply with all Applicable Laws relating to the safety of persons or property or their protection from damage, injury or loss; (4) designate a qualified and responsible employee at the Facilities whose duty shall be the supervision of Facilities safety, the prevention of fire and accidents and the coordination of such activities as shall be necessary with Federal, State and City officials; (5) operate all Equipment, Vehicles and Rolling Stock in a manner consistent with the manufacturer's safety recommendations; (6) provide for safe and orderly vehicular movements; and (7) develop and carry out a specific safety program, including employee training and periodic inspections.
- B. <u>Security</u>. The Company shall be responsible for the security of the Facilities, and shall maintain suitable fences, gates and locks at the Facilities. The Company shall guard against and be responsible for all damage or injury to such properties caused by trespass, negligence, vandalism or malicious mischief of third parties. The Company shall comply with all Applicable Law guidelines regarding security measures against terrorist threats and activities.
- 6.14 Disposal of Surplus Equipment and Scrap Metal. The Company may remove, dispose of and sell, in

accordance with Applicable Law, equipment constituting part of the Facilities that is obsolete or no longer needed, subject to approval by the City, and if necessary, in the opinion of the City's Law Director, by the City's Council in accordance with law. The proceeds from any cash sale shall be the property of the City, and in the event of a trade-in or like kind exchange of such surplus property, the City shall be entitled to the benefit thereof.

6.15 Loss, Damage or Destruction of Assets.

- A. Prevention and Repair. The Company shall use care and diligence, and shall take all appropriate precautions, to protect the Facilities from loss, damage or destruction. The Company shall promptly report to the City and the insurers, upon obtaining knowledge thereof, any damage or destruction to the Facilities and as soon as practicable thereafter shall submit a full report to the City. The Company shall also submit to the City within twenty-four (24) hours of receipt, copies of all accident and other reports filed with, or given to the Company by, any insurance company, adjuster or Regulating Entity. If the Company damages the Facilities or caused loss or destruction of the Facilities in connection with the performance of, or the failure to perform, the Services, except to the extent any such damage, loss or destruction was caused by Uncontrollable Circumstances, the Company shall promptly commence and proceed with due diligence to complete the repair, replacement and restoration of the Facilities at its own cost to at least the character or condition thereof existing immediately prior to the loss, damage or destruction, in accordance with and subject to the procedures set forth in Articles 9 and 12 hereof, as applicable. The repairs and replacements shall restore the damaged property to its character and condition existing immediately prior to the damage. The City shall have the right to monitor, review and inspect the performance of any repair, replacement and restoration work by the Company as if such work constituted part of the Facilities as originally constructed hereunder. If the damage, loss or destruction was not the result of the Company's actions, or if damage, loss or destruction was caused by Uncontrollable Circumstances, the Company will not be responsible for the costs of repair, replacement or restoration.
- B. <u>Insurance and Other Third-Party Payments</u>. To the extent that any repair, replacement or restoration costs incurred pursuant to this Section 6.15 can be recovered from any insurer or from another third party, each Party shall assist each other in exercising such rights as it may have to effect such recovery. Each Party shall provide each other with copies of all relevant documentation at no cost to the other Party, and shall cooperate with and assist the other Party upon request by participating in conferences, negotiations and litigation regarding insurance claims.
- C. Repair of Private Property. The Company shall promptly repair or replace all private property damaged by the Company or any officer, director, employee, representative or agent of the Company in connection with the performance of, or the failure to perform, the Services, except to the extent any such damage was caused by Uncontrollable Circumstances. The repair and replacements shall restore the damaged property to its character and condition existing immediately prior to the damage.
- 6.16 Uncontrollable Circumstance. If either Party shall rely on an Uncontrollable Circumstance as the basis for not performing its obligations under the terms and provisions of this Agreement, then the Party relying on such Uncontrollable Circumstance shall (i) provide prompt notice to the other Party of the occurrence of the act, event or condition giving an estimation of its expected duration and the probable impact on the performance of it's obligations hereunder, (ii) exercise its commercially reasonable efforts to continue to perform its obligations hereunder, (iii) in accordance with this Agreement, expeditiously take action to correct or cure the act, event or condition preventing performance, (iv) exercise its commercially reasonable efforts to mitigate or limit damages to the other Party to the extent such action will not materially adversely affect its own interests, and (v) provide prompt notice to the other Party of the cessation of the act, event or condition giving rise to its inability to perform.

6.17 Reporting and Record Keeping

- A. <u>Daily Operational Log.</u> The Company shall maintain a Daily Operation and Maintenance Log for each wastewater treatment plant as required by OEPA. These logs shall be maintained in formats acceptable to OEPA, shall be maintained on each treatment plant site and shall be made available for inspection by OEPA, City or emergency response personnel on a 24 hour per day basis. The log shall be current and shall include a minimum of the previous three months of data at all times. These logs will be required, as a minimum, to contain the following items:
 - 1. Identification of the public wastewater treatment facility

- 2. Signature and certificate number of the operator and the signature of persons making entries
- 3. Date and time of arrival and departure for each certified operator
- 4. Specific operation and maintenance activities and any repairs made
- 5. Results of tests performed and samples taken
- 6. Performance of preventative maintenance and repairs or requests for repairs of equipment

The Company shall submit an operational summary of the above information and data to the City in electronic format on a monthly basis.

B. <u>OEPA Reports</u>. The Company shall prepare and submit the monthly and annual reports as required by the NPDES permits for each treatment plant to OEPA and the City by the 25th of the following month or as otherwise agreed to by the City and the Company.

ARTICLE 7 - PERSONNEL

- 7.01 Adequate Staffing. The Company shall maintain a staffing level and qualifications at the Facilities sufficient to enable the Company to perform all of its obligations pursuant to the terms and provisions of this Agreement in a timely and efficient manner and in compliance with all Applicable Laws and OEPA requirements. The Company shall maintain staffing at or above the minimum staffing level of eight and one half (8 1/2).
- 7.02 <u>Labor Organizations and Collective Bargaining</u>. To the extent required or permitted by law, the Company shall recognize and bargain in good faith with any labor organization that represents the employees of the Company who perform work that is subject to this Agreement.
 - A. Any collective bargaining agreement entered into by the Company with the employees of the bargaining unit for the Facilities shall contain a "no strike/slowdown" provision in a form acceptable to the City.
 - B. Should any employees performing work under this Agreement not be represented by a labor organization but, during the Term of the Agreement become represented by a labor organization, and following certification in accordance with Applicable Law that a majority of such employees in a bargaining unit have authorized a labor organization to represent them for collective bargaining with the Company, the Company shall recognize the labor organization and bargain with it in good faith for a collective bargaining agreement.
- 7.03 Training of Employees. The Company shall ensure that all management and personnel education and training relative to the operation, maintenance, repair and management of the Facilities is continually updated on a scheduled basis, and that such management and personnel shall be recertified or relicensed, as applicable, as required or as recommended pursuant to Applicable Law. The Company shall provide training for all personnel involved in providing the Services, including training to integrate any new Equipment, Capital Improvements and processes that may be added to the Facilities during the Term of this Agreement. It is the sole responsibility of the Company to ensure that all personnel are fully knowledgeable of his or her duties and responsibilities. The Company shall be expected to develop, implement and maintain a formal training and retraining program for all personnel.
- 7.04 <u>Equal Employment Opportunity</u>. The Company confirms that it is an equal opportunity employer and during the performance of this Agreement, the Company agrees as follows:
 - A. The Company will not discriminate against any employee or applicant for employment because of race, color, religion, sex, or national origin. The Company will take affirmative action to ensure that applicants are employed and the employees are treated during employment, without regard to their race, color, religion, sex, or national origin. Such action shall include, but not be limited to the following: employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The Company agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the contracting officer setting forth the provisions of this nondiscrimination clause.
 - B. The Company will, in all solicitations or advertisements for employees placed by or on behalf of the Company, state that all qualified applicants will receive consideration for employment without regard to race, religion, sex or national origin.

- C. The Company will send to each labor union or representative of workers with which he has a collective bargaining agreement or other Agreement or understanding, a notice, to be provided by the agency contracting officer, advising the labor union or workers' representative of the Company's commitments under Section 202 of Executive Order No. 11246 of September 24, 1965, and shall post copies of the Notice in conspicuous places available to employees and applicants for employment.
- D. The Company will comply with all provisions of Executive Order No. 11246 of September 24, 1965, and of the Executive Order No. 11375 of October 13, 1967, and of the rules, regulations, and relevant order of the Secretary of Labor.
- E. The Company will furnish all information and reports required by Executive Order No. 11246 of September 24, 1965, and by the rules, regulations, and orders of the Secretary of Labor, or pursuant thereto and will permit access to his books, records, and accounts by the City and the Secretary of Labor for the purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- F. In the event of the Company's noncompliance with the nondiscrimination clauses of this Agreement or with any of such rules, regulations, or orders, this Agreement may be cancelled, terminated or suspended in whole or in part and the Company may be declared ineligible for further government contracts in accordance with procedures authorized in Executive Order No 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order No. 11246 of September 24, 1965 or by rules, regulation, or order of the Secretary of Labor, or as otherwise provided by law.
- G. The Company will include the provisions of Paragraphs A through G in every Subcontract or Purchase Order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to Section 204 of Executive Order No. 11246 of September 24, 1965, so that such provisions will be binding upon each Subcontractor or Vendor. The Company will take such action with respect to any Subcontract or Purchase Order as the City may direct of a means of enforcing such provisions including sanctions for noncompliance. Provided, however, that in the event the Company becomes involved in, or is threatened with, litigation with a Subcontract or Vendor as a result of such direction by the City, the Company may request the United States to enter into such litigation to protect the interest of the United States.
- H. The standard Federal Equal Employment Opportunity Construction Contract Specifications (Executive Order 11246), pages EEO-6 through EEO-13 of the specifications are incorporated into this Agreement by reference as if fully rewritten herein.
- Section 3 of the Housing and Urban Development Act of 1968 Summary of Requirements, pages EEO-18 through EEO-27 of the Specifications, are incorporated into this Agreement by reference as if fully rewritten herein.
- 7.05 No Restriction on Employment. At or prior to the Termination Date, the Company shall not place any restriction upon the ability of the employees at the Facilities to become employees of the City or employees of any contractor which may in the future operate and maintain the Facilities.
- 7.06 <u>City Not the Employer</u>. It is understood that individuals performing work under this Agreement shall be considered employees of the Company and not employees of the City. The City shall have no obligation to the Company's employees for wages or any fringe benefits, including pension benefits, nor any obligation with regard to their income requirements. Nothing in this Article 7 shall be construed to place the City in the relationship of the employer of or to grant the City the rights to direct or control either employees of the Company or displaced employees.

7.07 Project Manager.

A. The Company shall assign an individual employee of the Company to act as the Project Manager. The City reserves the right to reject the Company's proposed Project Manager. The Project Manager shall be responsible, on a full-time basis, for the management and oversight of the safe and reliable provision of all Services. The Project Manager shall be dedicated on a full-time basis to the Facilities and shall maintain an office in one of the Buildings within the Facilities or at another office of the Company located in the City. The Project Manager shall be expected to directly supervise the daily activities of Company personnel employed to operate, support and monitor all activities associated with the Services. The Project Manager will be the primary Company liaison with the City.

- B. The Project Manager shall hold and continue to hold a valid Ohio Treatment Works Class III Operators Certification, in accordance with OEPA requirements.
- C. The Project Manager shall be required to have working knowledge of the requirements of all Applicable Laws. The Project Manager shall work cooperatively with the City with respect to service quality, providing operational data, planning future service, responding to complaints and comments from customers and the general public, and responding to specific requests for other assistance.
- D. As required, the Project Manager shall attend all meetings and hearings pertaining to the provision of Services. This includes, but is not limited to, City Council meetings, and meetings with the Department. In the event the Project Manager is unable to attend such meetings, the Project Manager shall appoint a staff member with the authority to act on the Company's behalf and to appear as an agent of the Company in his or her place.
- E. In the event the Project Manager is temporarily unable to perform his or her duties, the Project Manager promptly shall appoint a staff member to serve in his or her place. The Company shall provide the City's Authorized Representative prior written notice whenever such appointment shall occur. If the Project Manager is unable to perform his or her duties for more than two (2) consecutive weeks, the Company shall assign another individual employee of the Company to act as a substitute Project Manager, subject to City approval and Applicable Laws.
- Operator of Record. The Company shall be required to employ a minimum of two (2) individuals having Ohio Class III Treatment Works Operator Certificates with one (1) being assigned as the Operator of Record for each wastewater treatment plant in accordance with OEPA requirements. Additional Class III Operators shall be employed, if necessary, to meet the minimum required staffing hours for Operators of Record at each wastewater treatment plant in accordance with OEPA requirements. The Operator of Record for each treatment plant shall also serve as the Operator of Record for the contributory sewerage collection system. The duties of the Operator of Record for the contributory sewerage collection system will include the technical supervision of the respective collection system in accordance with Ohio Administrative Code Section 3745-7 and the preparation and submission of the Ohio Environmental Protection Agency's Sanitary Sewer Overflow annual reports. The City will supply the Company with the necessary records and information required to prepare the Sanitary Sewer Overflow annual reports; and will repair and maintain both sewerage collection systems. The Company is not responsible for supervising the City's staff for the sewerage collection systems.
- 7.09 Background Checks: Drug Testing. The Company shall conduct appropriate background checks on applicants for employment, including, but not limited to, criminal record checks subject to Applicable Law. Subject to the terms of any agreement in effect on the Commencement Date between the Company and an authorized labor organization representing employees of the Company, the Company may require individuals from the Company's work force to whom it makes offers, or any other future applicant, to undergo the background checks and drug testing required by the Company; provided, however, that the Company shall, pursuant to Applicable Law, require drug tests of employees who hold commercial driver licenses in connection with their positions with the Company and shall further require pre-employment drug tests of employees who are required to possess federal, State or local licenses or certifications in connection with their positions with the Company. The Company shall have the right to reject for employment any person not passing the background checks and drug testing required by the Company or any person declining to allow the Company to perform the background check or drug testing. The Company shall maintain at all times a drug screening policy for its employees that is at least as comprehensive as the City's policy for its employees.

ARTICLE 8 - COMPENSATION

- 8.01 <u>Service Fee.</u> From and after the Commencement Date, the City shall pay the Company an annual Service Fee as compensation for the Company's performing the Services under this Agreement. The Service Fee shall be calculated and paid to the Company in accordance with this Article 8.
 - A. Service Fee Formula. The annual Service Fee shall be calculated in accordance with the following formula:

$$SF = FF + RC - CR$$

where,

SF	= Service Fee
FF	= Fixed Fee
RC	= Reimbursable Costs Charge
CR	= Credits

Each component of the Service Fee shall be determined in accordance with this Article 8.

B. <u>Fixed Fee</u>. The Fixed Fee shall be compensation for each Contract Year for all Services to be provided by the Company under this Agreement except where otherwise noted herein. The Fixed Fee for the first Contract Year shall be Two Million, Five Hundred Seventy Three Thousand, One Hundred Thirty Three Dollars (\$2,573,133.00). For subsequent Contract Years the Fixed Fee shall be adjusted taking into account inflation, treatment plant flows and number of lift stations in accordance with the following formula:

$$FF_n = CPI_n/CPI_i(FF_i + (TF_n - TF_i)TF + (LS_n - LS_i)LS)$$

Where,

FF_n	= The annual Fixed Fee for Contract Year "n"
FF_i	= The annual Fixed Fee for the first Contract Year which is \$2,573,133.00
CPl_i	 The Consumer Price Index on the September 1st immediately preceding the Commencement Date
CPI _n	The Consumer Price Index on the September 1st immediately preceding "n" Contract Year
TF	= The annual Fixed Fee flow adjustment price per 0.1 million gallons of flow which is \$74.19
TF_{i}	 The combined average daily effluent flow from both wastewater treatment plants for the period from September 1, 2023 to September 1, 2024 rounded to the nearest 0.1 million gallons which is 1.8 million gallons.
TF _n	The combined average daily effluent flow from both wastewater treatment plants for the one year period preceding the September 1 st immediately preceding Contract Year "n" rounded to the nearest 0.1 million gallons.
LS	= The annual Fixed Fee lift station adjustment price per lift station which is \$3,595.00
LS _i	 The initial number of lift stations included in the Facilities on October 1, 2024 which is 19.
LF_n	= The actual number of lift stations included in the Facilities on October 1 of Contract Year "n"

In the case of the addition or deletion of Lift Stations, the Fixed Fee shall be adjusted reflecting the addition of a Lift Station on the date that the Lift Station becomes operational and reflecting the deletion of a Lift Station on the date that the Lift Station is taken out of service. The adjusted Fixed Fee shall be reflected in the next subsequent Billing Month invoice.

The annual Fixed Fee flow adjustment is conditioned upon the flow weighted annual average influent pollutant

concentrations for the one year period preceding the September 1st immediately preceding Contract Year "n" being within ten percent (10%) of the flow weighted annual average influent flow pollutant concentrations for the one year period preceding the September 1st immediately preceding the Commencement Date. The flow weighted annual average influent pollutants shall include CBOD₅, TSS, NH₃ and Phosphorus all expressed in milligrams per liter (mg/l). Flow weighting of each pollutant shall be in accordance with the following formula:

$$PC = ((PC_B \times F_B) + (PC_C \times F_C))/(F_B + F_C)$$

Where,

PC	=	Flow weighted annual average influent pollutant concentration for specific pollutant
PC_{B}	=	Average annual influent pollutant concentration at Treatment Plant B
PC _C	=	Average annual influent pollutant concentration at Treatment Plant C
F_{B}	****	Average annual daily effluent flow for Treatment Plant B
F_C	<u> </u>	Average annual daily effluent flow for Treatment Plant C

If the flow weighted annual average influent pollutant concentrations for any of the named pollutants for the one year period preceding September 1st immediately preceding Contract Year "n" vary by more than ten percent (10%) from the flow weighted annual average influent concentration for the one year period preceding the September 1st immediately preceding the Commencement Date, the City and Company will negotiate an equitable annual Fixed Fee flow adjustment. The initial flow weighted annual influent pollutant concentrations are CBOD₅ (141 mg/l), TSS (254 mg/l), NH₃ (20.8mg/l) and Phosphorus (4.6 mg/l).

- C. <u>Reimbursable Cost Charge</u>: The Reimbursable Costs Charge shall be an amount equal to the actual and direct expenses (without markup for profit, administrative costs or otherwise) paid by the Company to unrelated third parties for any charges incurred in the operation of the Facilities that are the responsibility of the City pursuant to this Agreement. Such charges for which the City is responsible (and for which the Company shall be reimbursed to the extent paid by the Company) shall be identified and approved by the City.
- D. <u>Credits.</u> Except as otherwise provided in this Agreement, the City shall receive a credit for expenses paid by the City (without markup for administrative costs or otherwise) for charges incurred in the operation of the Facilities that are the responsibility of the Company pursuant to this Agreement.

8.02 Additional Services: Change in Scope of Services.

- A. A change in the scope of services under this Agreement shall occur when the Company's costs for providing service under this Agreement changes as a result of the following:
 - City Requested Additional Services. If the City requests the Company to provide Additional Services, the City and the Company shall negotiate the costs associated with the Additional Services subject to Cost Substantiation as provided in Section 17.04 hereof. In the event that the Parties cannot agree to an appropriate fee for such Additional Services, the procedures set forth in Section 16.01 of this Agreement shall apply. The Company will provide Additional Services on a twenty- four (24) hour, seven (7) day a week basis at the request of the City.
 - 2. Uncontrollable Circumstances. In the event that there occurs an Uncontrollable Circumstance of a Party to this Agreement that results in a change in the scope of services under this Agreement or a material change in the Company's costs for providing service under this Agreement, the City and the Company agree to negotiate in good faith such amendments to this Agreement as such Parties shall determine to reflect the change in scope of services and provide for an equitable adjustment to the Company's Service Fee.

B. The Authorized Representative of each Party shall have full authority to approve changes in the scope of services of this Agreement and compensation therefor, execute written change orders reflecting such changes, render decisions promptly, and furnish information expeditiously to the other Party when necessary subject to Applicable Law and subject to approval of the City's Council when required.

8.03 Billing and Payment.

A. <u>Billing</u>. The City shall pay the Service Fee in monthly installments in an amount equal to the sum of: (1) one-twelfth (1/12) of the annual Fixed Fee; (2) any monthly Reimbursable Costs Charge described in Section 8.01 hereof; (3) less any credits described in Section 8.01 hereof determined on a monthly basis. The Company shall provide the City with an invoice no later than the first (1st) day of each Billing Month which sets forth the monthly Fixed Fee portion of the Service Fee for such Billing Month and of Reimbursable Costs Charge or any Additional Services charge for the prior Billing Month and such other documentation or information as the City may reasonably require to determine the accuracy and appropriateness of the invoice.

If the September 1st Consumer Price Index is not published by the United States Department of Labor by October 1st of Contract Year "n", the annual Fixed Fee adjustment shall occur as soon as practicable after publication. The subsequent invoice after the Consumer Price Index is published will include any incremental adjustment in Fixed Fee retroactive to October 1st of Contract Year "n".

B. <u>Payment</u>. The City shall pay the invoice, less any Credits described in Section 8.01 hereof within thirty (30) days after receipt of the invoice. The City shall provide the Company with a statement setting forth Credits taken.

8.04 Estimates and Adjustments.

- A. <u>Pro Rata Adjustments</u>. Any computation made on the basis of a stated period shall be adjusted on a pro rata basis to take into account any initial or final period which is a partial period. For purposes of this subsection, a Billing Month shall be taken as a month containing thirty (30) days.
- B. <u>Budgeting</u>. For City budgeting purposes, no later than September 1st of the year, the Company shall provide to the City a written statement setting forth for the next Contract Year its reasonable estimate of the aggregate Service Fee, each component thereof, and the estimated annual Additional Services charge, if any. The estimate shall not be binding on the Company but shall establish the basis for monthly billing for such Contract Year.
- C. Adjustment to Service Fee for Material Change in Operation and Maintenance Costs Due to Capital Improvement. In the event that the City implements a Capital Improvement with a construction, installation, or purchase value in excess of One Hundred Thousand Dollars (\$100,000) and, pursuant to Section 9.01 hereof, the Company or the City requests an adjustment to the Fixed Fee of the Service Fee after implementation of such Capital Improvement to account for a material increase or decrease to the Company's operation and maintenance costs under this Agreement, the City and the Company agree to negotiate in good faith such amendments to this Agreement as such Parties shall determine. Any adjustments to the Fixed Fee of the Service Fee resulting from an amendment to this Agreement shall be deemed to be a material revision of the compensation arrangements under this Agreement.
- 8.05 <u>Interest on Overdue Payments</u>. All payments outstanding after the applicable due date shall be compounded annually at the prime rate as reported in the Wall Street Journal plus one percent (1%).
- 8.06 <u>Payment Disputes</u>. If any Party shall dispute an amount owing to the other Party, such Party shall utilize the provisions of Article 16 hereof regarding dispute resolution, and shall:
 - A. Give notice at least fifteen (15) days prior to the due date for such payment from the other Party of such disputed amount together with sufficient information to allow the other Party to understand the nature of the dispute delivered on or before the due date of the amount disputed; and
 - B. Pay all undisputed amounts on the due date. Interest at the rate specified in Section 8.05 hereof shall accrue from the original due date on disputed amounts, or the portions thereof, to the Party which is ultimately determined to be entitled to such disputed amount, or any portions of such disputed amounts.

8.07 Offsets.

- A. In order to provide the City with unencumbered, unilateral authority to clean, repair, replace, operate, maintain, haul, dispose or conduct any of the Services required of the Company pursuant to this Agreement in the event the Company is not in compliance with the terms of this Agreement or in the event the City is entitled to indemnification pursuant to Article 13 of this Agreement or in the event fines, penalties or any other amounts are owed but unpaid by the Company pursuant to any provision of this Agreement, including without limitation, Sections 6.06 and 8.06, or any Event of Default, the City shall have the right, upon the giving of notice as provided below, to offset the amount the City may incur to remedy the Company nonperformance of the Services including but not limited to indemnity claim, fines, penalties or other amounts against amounts then owing or to become owing by the City to the Company pursuant to the terms and provisions of this Agreement, including any Exhibits hereto. Notwithstanding the foregoing, the City shall not offset against or delay payments due the Company as a result of, or based on, any failure of the City to receive payment properly billed to users of the Facilities and/or the inability of the City to collect monies owed the City. The rights of the City under this Section 8.07 shall be in addition to, and not in limitation of, any other rights, which it may have.
- B. In the event that the City determines that there is reason to suggest that the Company is not performing any of its obligations under this Agreement, the City may, in its sole and absolute discretion, provide written notification of withholding to the Company describing its intent to remedy such nonperformance, including a good faith cost estimate to perform such remedy.
- C. Within forty-eight (48) hours of receipt of a withholding notification from the City, the Company shall respond to the City with a written response explaining the cause of such noncompliance and outlining the corrective measures to be performed within seven (7) calendar days of the receipt of such withholding notification.
- D. In the event that the Company fails to commence corrective action as described in this response to the City, the City shall take all actions necessary to remedy the conditions in questions and shall offset the amount of the cost of such remedy against amounts then owing or to become owing by the City to the Company pursuant to the terms and provisions of this Agreement, including any Exhibits hereto.
- E. The conditions of nonperformance giving rise to the right of set-off under this Section 8.07 may or may not be an Event of Default as defined in Article 14 of this Agreement.
- 8.08 <u>Tax Exemption of Facilities</u>. It is the intent of the Parties that the Facilities shall continue to be City owned property and not subject to real property taxation.
- 8.09 Sales Tax. In its performance of the Services, the Company acknowledges that construction materials and supplies acquired by the Company or any Subcontractor in connection with the Capital Improvements and operating supplies relating to the performance of the Services are subject to State sales tax. The Company further acknowledges that these taxes have been priced into the Fixed Fee of the Service Fee and agrees to pay all such taxes imposed with its performance of the Services without reimbursement from the City.

ARTICLE 9 - CAPITAL IMPROVEMENTS

- 9.01 General. Any Capital Improvements implemented during the Term shall comprise Facilities for which the Company shall be responsible for operation and maintenance pursuant to this Agreement as part of the Services. The City will determine, in its sole discretion, when a Capital Improvement has reached Substantial Completion, but the Company may review and comment upon the City's determination. The Company or the City may request an adjustment to the Fixed Fee of the Service Fee after implementation of a Capital Improvement with a construction, installation, or purchase value in excess of One Hundred Thousand Dollars (\$100,000) to account for a material increase or decrease to the Company's operation and maintenance costs. Adjustment, if any, to the Fixed Fee of the Service Fee to reflect changes in operational conditions due to Capital Improvements shall be made as set forth in Section 8.04 (C) hereof.
- 9.02 <u>Procedures for Implementing of Capital Improvements</u>. The City shall determine in it sole discretion which Capital Improvements shall be completed and how the Capital Improvements shall be implemented in accordance with Applicable Law. All items which meet the terms of the definition of Capital Improvement as set forth in Section 3.02 hereof, but which have a construction, installation or purchase value (without

combining a series of smaller projects or cost items) that is equal to or less than the Capital Improvement Threshold Amount shall be completed and paid for by the Company as part of the Services included in the Fixed Fee of the Service Fee. The City may request the Company to complete any Capital Improvement or it may complete the Capital Improvement on its own behalf, through contractors designated by it through a bidding process completed in compliance with Applicable Law. If the City elects to complete the Capital Improvement through contractors other than the Company, the Company shall use all commercially reasonable efforts to cooperate with the City and its contractors in completing such improvements by (a) providing reasonable access to the areas of the Facilities necessary or useful to complete the work and (b) adjusting operations at the Facilities at the direction of the City or its contractors, in such a manner as will permit completion of the work without impairing or impeding the Company's ability to meet its obligations under this Agreement. The City agrees to use all commercially reasonable efforts to minimize disruption to the Company's operation of the Facilities during the completion of the work, but nothing in this Section 9.02 shall be construed as prohibiting the City from taking any action necessary to ensure that the work can be completed, including cessation of operations at portions of the Facilities. If the City elects to require the Company to complete the Capital Improvement, the City shall pay all costs incurred by the Company in connections therewith subject to the Cost Substantiation provisions of Section 17,04 hereof.

9.03 Capital Improvements at Company's Request. The Company may, upon the prior written approval by the City, and at the Company's sole cost and expense, make Capital Improvements to the Facilities to increase the efficiency of the Facilities and achieve cost savings. The City may approve such Capital Improvements, in its sole discretion, upon submission to the City for its review and comment of appropriate plans and specifications and price proposals for such Capital Improvement. At such time as the City approves the Capital Improvement requested by the Company hereunder, the Company and the City shall agree upon and determine the useful life and amortization schedule of such Capital Improvement. All Capital Improvements made by the Company shall be owned by the City and shall become part of the Facilities. In the event the City terminates this Agreement pursuant to Section 15.04 or 15.09 hereof, the City shall pay the Company certain of its undepreciated costs for any Capital Improvements approved by the City and paid for by the Company pursuant to this Section 9.03, which undepreciated costs to be paid by the City are expressly understood to be limited to solely the purchase price paid by the Company to the independent vendor and any labor costs related to such Capital Improvement.

9.04 Capital Improvements Due to Change in Law or Uncontrollable Circumstance.

- A. The Company and the City shall each notify the other in writing within thirty (30) days following them becoming aware of a Change in Law or as provided in Section 6.16 hereof with regard to an Uncontrollable Circumstance. Within a time that reasonably permits the City to evaluate and implement changes necessary to comply with such Change in Law or to address such Uncontrollable Circumstance following such notice, the Company shall prepare an evaluation of the expected impact of such Change in Law or Uncontrollable Circumstance on (1) the operation or maintenance of the Facilities including any cost increases or decreases associated therewith, (2) the modifications to the Facilities including any Capital Improvements, that may be required as a direct consequence of such Change in Law or Uncontrollable Circumstance, and (3) the ability of the Company to continue to operate the Facilities in accordance with this Agreement.
- B. As soon as is reasonably practicable following notice and evaluation of the expected impact of a Change in Law or Uncontrollable Circumstance, the Company shall prepare and deliver to the City a proposed response to such Change in Law or Uncontrollable Circumstance including (1) an evaluation of the Change in Law or Uncontrollable Circumstance prepared pursuant to paragraph (A) above, and (2) a list of proposed recommendations to address the Change in Law or Uncontrollable Circumstance. The proposed response shall be prepared by the Company with the objective of providing the most cost- effective long-term response options available to the Change in Law or Uncontrollable Circumstance, taking into account projected capital costs and operating and maintenance costs during the remaining portion of the Term (or, at the request of the City, for a greater period of time). The proposed response shall include, but not be limited to (1) the proposed Capital Improvements and/or operational changes, if any, (2) the estimated cost of the proposed Capital Improvements and/or operational changes, (3) the proposed effect on the operation, maintenance and management of the Facilities by item, and (5) the effect, if any, on the ability of the Company to comply with the terms and conditions of this Agreement including operating in compliance with Applicable Law.
- C. The City shall review the proposed response submitted to it by the Company and, within a reasonable

period of time, notify the Company that the City either (1) agrees with and accepts the Company's proposal, (2) rejects the Company's proposal, or (3) accepts the Company's proposal with modifications (with the proposed modifications specified). If the City rejects the Company's proposal, the City may request the Company to submit a revised proposal or the City may submit its own proposal. The Company shall accept any proposal submitted by the City or any modifications to its proposal, provided, however, the Company shall notify the City of the impact, if any, of any City proposal or modification on (1) the ability of the Company to continue to operate the Facilities in accordance with this Agreement and (2) the anticipated cost impact of the City's proposal or proposed modifications, compared to the Company's proposal.

- D. If a Change in Law or Uncontrollable Circumstance will require a Capital Improvement to the Facilities, the City may (1) request the Company to complete the Capital Improvements, (2) complete the Capital Improvement on its own behalf, either directly or through contractors designated by it, or (3) not elect in the City's reasonable discretion consistent with Prudent Industry Practices to complete the Capital Improvement.
- E. If the City elects to complete the Capital Improvement itself or through a contractor, the City shall bear all costs and expenses of the project including, but not limited to, costs of acquisition and installation of the Equipment and related construction work. If the City elects to require the Company to complete the Capital Improvement, the City shall pay all costs incurred by the Company in connection therewith subject to the Cost Substantiation as provided in Section 17.04 hereof.
- F. Upon the Company becoming aware of a Change in Law and prior to the implementation and completion of a response to the Change in Law, the Company shall use all reasonable efforts to mitigate the harm or effect that will be caused by the Change in Law and to avoid or minimize the cost impact of such Change in Law to the City.

ARTICLE 10 - PERFORMANCE GUARANTEES

10.01 General. The Company shall at all times comply with the Performance Guarantees set forth herein, except to the extent compliance is prevented or excused by Uncontrollable Circumstances, agreement of the Parties or by provisions of this Agreement. If the Company fails to comply with any Performance Guarantee, the Company shall, to the extent required and not by way of limitation: (1) promptly notify the City within twenty-four (24) hours of the Company's having knowledge of any such noncompliance, except if such noncompliance results in a violation of the terms of the NPDES Permits, or any other Applicable Law, consists of a major spill or jeopardizes public health, in which case the Company shall notify the City as soon as possible but no later than two (2) hours after such noncompliance; (2) promptly provide the City within twenty-four (24) hours with copies of any notices sent to or received from EPA, OEPA or any other Regulating Entity having regulatory jurisdiction with respect to any violations of Applicable Law; (3) pay any resulting direct damages, fines, judgments or awards, including penalties, levies, assessments, penalties or other charges resulting therefrom and indemnify the City in accordance with Section 13.01 hereof; (4) at its own cost and expense, take any action necessary within the scope of the Company's responsibilities under this Agreement (including, without limitation, recommending Capital Improvements and making repairs, replacements, and operating and management practices changes) in light of the nature, extent and repetitiveness of such noncompliance, in order to comply with such Performance Guarantee, to continue or resume performance hereunder, to eliminate the cause of such noncompliance, and to reasonably assure that such noncompliance will not recur; (5) promptly prepare all Public Notifications required by Applicable Law, and submit such Public Notifications for publication; and (6) assist the City with all public relations matters necessary to adequately address any public concern caused by such noncompliance, including, but not limited to, preparation of press releases, attendance at press conferences, and participation in public information sessions and meetings.

10.02 Effluent Guarantee.

- A. <u>Applicable Law and Additional Standards</u>. Except to the extent relieved as provided in Section 10.07 hereof, the Company shall operate the Wastewater Facilities on a continuous, uninterrupted twenty-four (24) hour per day, seven (7) day per week basis so as to receive and treat all Influent and discharge Effluent in compliance with the requirements of Applicable Law, including the NPDES Permit, and that is within the design capabilities of the Facilities.
- B. Indemnity for Loss from Non-Complying Effluent. In the event that any Effluent discharged by the

Company fails to comply with the Effluent Guarantee, except to the extent such failure of compliance is caused by an Uncontrollable Circumstance, the Company, in addition to its obligations under Section 6.06 hereof, shall indemnify, defend and hold harmless the City in accordance with Section 13.01 hereof from the Loss of any tort, environmental or other liability resulting in any Legal Proceeding originated by any third party arising from the discharge of such non-complying Effluent. This indemnity shall extend to any liability resulting from property loss or damage or death or personal injury suffered or alleged to be suffered by any person from exposure to such non-complying Effluent based on any theory of recovery, including theories of environmental impairment. The Loss to which the indemnity provided in this Section 10.02 extends shall not be construed to constitute consequential or other damages as set forth in Article 13 hereof, as to which both Parties have waived any rights of recovery.

- C. Change in Law Affecting Effluent. The Parties acknowledge that a Change in Law may hereafter affect Effluent standards or impose more stringent requirements relating to equipment or processes than those established hereunder as of the Agreement Date. In the event a Change in Law affecting Effluent occurs, the Company shall not be entitled to performance relief or additional compensation under this Agreement unless (1) such Change in Law imposes a regulatory standard or operating requirement with respect to any particular Effluent characteristic or parameter which is more stringent or burdensome to comply with than the Contract Standards applicable to such characteristic or parameter, or requires equipment or processes not then in place or practiced at the Wastewater Treatment Facilities, and (2) the Company is unable, after taking all mitigation measures required under Section 6.16 hereof with respect to such a Change in Law, to avoid the necessity for such performance relief or additional compensation.
- 10.03 Wet Weather Operations Guarantee. The Company shall operate and maintain the Facilities to maximize the volume of flows received and treated at the Wastewater Treatment Facilities in accordance with Applicable Law.
- 10.04 <u>Residuals Guarantee</u>. Except to the extent relieved by Uncontrollable circumstances, the Company shall operate all Residuals processing systems and deliver Residuals in accordance with the requirements of Applicable Law and the Services.
- 10.05 Odor Control Guarantees. Except to the extent relieved by the occurrence of an Uncontrollable Circumstance, the Company shall comply with all limits and requirements established by Applicable Law in operating the Facilities with respect to odor control.
- 10.06 Upsets and Non-Specified Influent Affecting Company Compliance with Performance Guarantees.
 - A. Required Demonstration. The occurrence of an Upset or the receipt of Non-Specified Influent shall not be considered to be an Uncontrollable Circumstance, and the Company shall not be entitled to relief from a Performance Guarantee due to the occurrence of an Upset or the receipt of Non-Specified Influent, except to the extent that the Company affirmatively demonstrates through properly signed, contemporaneous operating logs, or other relevant evidence that:
 - 1. an Upset actually occurred or Non-Specified Influent was actually received; and
 - the occurrence or receipt thereof could not have been prevented by compliance with the Contract Standards.
 - B. Response Measures to Upsets and Non-Specified Influent. If an Upset occurs or a Wastewater Treatment Facility received Non-Specified Influent, the Company shall, without limiting its obligations under the Contract Standards; (i) use all reasonable efforts consistent with Prudent Industry Practices to maintain Wastewater Treatment Facility performance as if the Upset had not occurred or Non-Specified Influent had not been received; (ii) advise the City of the situation as soon as possible and no later than four (4) hours of the Company's first knowing of the occurrence of an Upset or the receipt of Non-Specified Influent and advise the City of the Company's planned course of action as soon as possible and not later than twelve (12) hours of the Company's first knowing of the occurrence of an Upset or the receipt of Non-Specified Influent; (iii) submit any notice thereof required by Applicable Law, and (iv) use all reasonable efforts consistent with Prudent Industry Practices to return the Effluent to compliance with the requirements of Applicable Law and the Performance Guarantees as soon as reasonably possible, but, in any event, within fourteen(14) days after an Upset and three (3) days after the Wastewater Treatment Facility has ceased receiving Non-Specified Influent.

- C. Reimbursement of Costs. To the extent the occurrence of an Upset or the receipt of Non-Specified Influent constitutes an Uncontrollable Circumstance hereunder, the City will reimburse the Company an amount equal to the reasonable costs incurred by the Company with respect to such Uncontrollable Circumstance, including the reasonable, documented and substantiated costs incurred by the Company in responding to the effect of the Uncontrollable Circumstance on the Facilities and on the treatment and disposal of Effluent and Residuals, but excluding any such costs which would have been avoided had the Company complied with any remedial measures required under Applicable Law and appropriate mitigating measures required by Section 6.16 hereof.
- 10.07 Notwithstanding any of the City's specific remedies as set forth above, the City in connection with any of the Company's Guarantees will reserve any and all of its rights and remedies provided to it under law or equity.

ARTICLE 11 - PERFORMANCE BOND

11.01 <u>Performance Bond</u>. The Company shall provide an annual performance bond equal to the then current Fixed Fee of the Service Fee as security for the value of the services/work to be performed by the Company under this Agreement. This bond will be issued for a one-year period effective on the Commencement Date and shall be renewed annually thereafter. The form of such bond shall be substantially in the form submitted by the Company and attached hereto as <u>Exhibit 4</u>, and further subject to review and approval by the City's Law Director.

Such bond shall be the security for the performance of the Company's obligations under this Agreement and shall be provided by a surety company approved to transact business in the State of Ohio.

Neither non-renewal by the surety company, nor failure, nor inability of the Company (principal) to file a performance bond for subsequent terms under this Agreement shall constitute loss to the City (oblige) recoverable under the bond. The maximum liability under the bond for the surety is the penal sum of the bond and not cumulative in amount.

ARTICLE 12 - INSURANCE

- 12.01 <u>Insurance Procurement: Duty to Maintain: Obligation to Provide Continuous Coverage.</u>
 - A. <u>Procurement.</u> Throughout the Term of this Agreement the Company, on its own behalf and on behalf of anyone directly or indirectly employed by it for whose acts or omissions it may be liable, shall secure, or cause to be secured, and maintain at its cost and expense, including premium payments the insurance policies described in <u>Exhibit 5</u> and attached hereto. The cost and expense, including premium payments will be included in the Fixed Fee of the Service Fee.
 - B. <u>Duty to Maintain</u>. Each policy shall be secured prior to the Commencement Date and the policies shall be continuously maintained through the Term of this Agreement.
 - C. <u>Continuous Coverage</u>. The Company shall assure continuous coverage if any policy is canceled, not renewed or materially changed. The Company shall, at its own expense, pay such extra premium as required to assure no lapse of coverage for any time period.
 - D. Evidence of Insurance. With sufficient notice copies of all policies of insurance can be made available to the City for the City's review. If a policy is canceled, not renewed or materially changes, a certificate for the substitute policy shall be submitted to the City as soon as possible before the commencement of the policy period. The Company shall annually supply the City with proof of insurance in the form of a Certificate of Insurance which reflects all coverages required by this Agreement.
 - E. <u>Property Insurance</u>. The City shall procure and maintain property insurance as deemed appropriate for Facilities. The City and the Company hereby waive and release each other with respect to damages caused by events that are covered by the property insurance required herein. The City agrees to provide the Company a waiver of subrogation on behalf of itself and its insurance carriers, to the extent available. The City shall provide the Company with a Certificate of Insurance.
- 12.02 <u>Policy Requirements and Company Obligations.</u> The policy or policies procured, or caused to be procured hereunder, shall satisfy the following requirements:

- A. To the extent possible in the insurance marketplace, each policy shall specifically insure the Company's indemnification obligations.
- B. The Company shall promptly provide the City with notice prior to cancellation, non-renewal or material change of each policy subsequent to Company's receipt of such notice by its insurers.
- C. Each policy shall provide that the Company and the Company's insurers shall have no right of recovery or subrogation against the City or the Company. The intention of the Parties is that any insurance policy by the Company shall protect both such Parties and shall be the primary coverage for any losses covered by the insurance policies.
- D. The City and the employees of the City shall be named as additional insureds on a primary and non-contributory basis with respect to the Company's commercial general liability policy, and auto liability policies. The additional insured status is limited to liability arising out of Facilities operations under this contract conducted by the Company. The Company will be responsible for premium payment and will receive loss proceeds on the part of the City.
- E. The Company's insurance policy shall provide that the insurance company shall have no recourse against the City for payment of any premiums or for assessments under any form of policy.
- F. The Company shall be solely responsible to satisfy any and all deductibles and self-insured retentions contained in its insurance coverages as well as any excluded loss or losses if the same are within the Company's liability under this Agreement. The Company shall promptly notify the City of any material change to any and all deductibles and self-insured retentions.
- G. Each company providing coverage required by this Agreement shall be licensed or authorized to transact business in the State of Ohio.
- H. The Company's failure to secure and maintain the insurance required under this Agreement shall not relieve the Company of its liability for any losses intended to be insured thereby. These insurance provisions shall not be construed or interpreted so as to conflict with the indemnification obligations of Article 13 hereof.
- I. The Company's failure to secure and maintain the insurance required under this Agreement, notwithstanding any other provision of this Article 12, shall be deemed an Event of Default for purposes under Article 14 hereof.

ARTICLE 13 - INDEMNIFICATION AND LIMITATION OF LIABILITY

13.01 Indemnification by the Company.

A. Except as otherwise set forth in Section 13.01 (B) or (C) hereof and except to the extent that a matter is covered by insurance required to be obtained hereunder, the Company shall defend, indemnify and hold harmless the City, the Department and their respective officials, officers, employees, agents, representatives, consultants, contractors and subcontractors (individually, a "City Indemnitee," and collectively, the "City Indemnitees") from and against any and all liabilities, actions, damages, indirect damages, consequential damages, personal injuries, death claims, other claims, lawsuits, demands, judgments, suits, losses, deficiencies, obligations, fines, penalties, costs and expenses (including legal, accounting and consulting fees) (collectively, "Losses"), to the extent arising out of or relating to, directly or indirectly, (i) performance by the Company or any of its officers, members, employees, agents, representatives, consultants, contractors or subcontractors of the terms and provisions of this Agreement; or (ii) any inaccuracy or-misrepresentation in or breach of any representation, warranty, covenant or agreement made by the Company in this Agreement or in any document, certificate or affidavit delivered by the Company pursuant to the terms and provisions of this Agreement, notwithstanding the fact that any such inaccuracy, misrepresentation, breach or noncompliance has been cured by the Company and regardless of whether or not any such inaccuracy, misrepresentation, breach or noncompliance has been cured by the Company and regardless of whether or not any such inaccuracy, misrepresentation, breach or noncompliance constitutes an Event of Default under Article 14 hereof; provided the Company's indemnity shall not extend to any matter for which the City is required to indemnify the Company hereunder.

The Company shall use commercially reasonable efforts to incorporate this indemnification obligation in all subcontracts entered into with suppliers of materials or services, and all labor organizations who furnish skilled and unskilled labor, or who may perform any such labor or services in connection with a contract entered into hereunder.

The Indemnification obligation under this Section 13.01 shall not be limited in any way be any limitation on the amount or type of damages, equitable relief, compensation or benefits payable by or for the Company, any Subcontractor or any Subcontractor of a Subcontractor under worker's compensation acts, or other employee benefit acts. The City shall promptly notify the Company of all notices of claims and tender the defense of claims. The Parties agree to exercise all reasonable efforts to cooperate with one another to the extent their respective interest may appear.

The Company's indemnity obligation includes indemnification for all reasonable expenses, court costs and attorney fees, including those incident to appeals incurred by or imposed upon the City Indemnitees in connection with enforcement or defense of the City Indemnitees' rights to indemnity hereinabove provided. In addition, the Company agrees that the City Indemnitees may employ any attorney (or attorneys) of their choice and/or may use its in-house counsel in a matter to enforce or defend the City Indemnitees' right to the indemnity hereinabove provided. However, if the City Indemnitees engage their own legal counsel, and the Company has engaged or offered to engage legal counsel to defend the City Indemnitees in the matter, the City Indemnitees shall bear their own costs and expenses of their legal counsel, unless the Company's and the City Indemnitees' positions in the matter are in conflict, in which case all reasonable costs and expenses of the City Indemnitees' legal counsel shall be borne by the Company.

- B. Notwithstanding the foregoing subsection (A), provided the Company is in compliance with the provisions of Sections 10.01 and 10.06 hereof, to the extent a City Indemnitee incurs any Losses as a result of the Company's unexcused failure to satisfy the Performance Guarantees or other nonperformance by the Company of the terms of this Agreement, which failure or nonperformance is a demonstrated direct result of Non-Specified Influent or a failure by the City to provide a required Capital Improvement pursuant to Section 9.04 (D) of this Agreement, the Company shall indemnify the City Indemnitee to the full extent of the Company's insurance coverage required pursuant to the terms of Article 12 of this Agreement for such Losses, if any, but shall otherwise be relieved of its obligation to indemnify the City Indemnitee to the extent and in such amount as the Losses incurred by the City Indemnitee exceed the Company's insurance coverage required pursuant to Article 12 of this Agreement for such Losses, irrespective of any deductible the Company is required to pay, if any, in connection with such insurance coverage. Notwithstanding the foregoing, at no time shall Company be required to obtain and/or maintain any additional insurance that is not otherwise specifically required pursuant to the terms of this Agreement.
- C. The Company need not indemnify nor defend the City Indemnitees for Losses arising from construction project design performed by third parties or construction performed by third parties contracting with the City. In such event, the Parties shall bear their own costs of defense and such Losses as the law assigns to each.
- D. With respect to this Section 13.01, the Parties shall have the right to defend their respective interests. The costs for such defense shall be included as part of the costs of Losses of the responsible Party pursuant to this Section 13.01.

13.02 Indemnification by the City

A. To the extent permitted by Applicable Law, and subject to future lawful appropriation of funds, and except to the extent a matter is covered by insurance required to be obtained thereunder, the City shall protect, indemnify and hold harmless the Company and its officials, officers, employees, agents, representatives, consultants, contractors and Subcontractors (individually, a "Company Indemnitee" and collectively, the "Company Indemnitees") from and against any and all Losses arising out of or relating to, directly or indirectly, (i) the negligence or intentional misconduct of the City or any of its officials, employees, agents, representatives, engineers, consultants, contractors or subcontractors in connection with the performance by the City of the City's responsibilities, if any, under the terms and provisions of this Agreement; or (ii) any inaccuracy or misrepresentation in or breach of any representation, warranty, covenant or agreement made by the City in this Agreement or in any document, certificate or affidavit

delivered by the City pursuant to the terms and provisions of this Agreement; provided the City's indemnity shall not extend to any matter for which the Company is required to indemnify the City hereunder.

B. To the extent a City Indemnitee requires any indemnity in any contract with a City contractor relative to the Facilities the City Indemnitee hereby agrees to include the Company as an indemnified party in any such indemnification. The Company shall give prompt written notice of a claim and tender the defense when invoking any right of indemnification.

13.03 Procedure in Event of Indemnity.

- A. Notice to the indemnifying party shall be given promptly after receipt by any City Indemnitee or any Company Indemnitee of actual knowledge of the commencement of any action or the assertion of any claim that will likely result in a claim by it for indemnity pursuant to this Agreement. Such notice shall set forth in reasonable detail the nature of such action or claim to the extent know, and include copies of any written correspondence from the party asserting such claim or initiating such action. The indemnifying party shall be entitled, at its own expense, to participate in the defense of such action or claim or, if (i) the action or claim involved seeks (and continues to seek) solely monetary damages, (ii) the indemnifying party is obligated to indemnify and hold harmless the other party with respect to such damages in their entirety pursuant to Section 13.01 or 13.02 hereof, and (iii) the indemnifying party shall admit in writing it obligation to indemnify in connection therewith, then such party shall be entitled to assume and control such defense with counsel chosen by such party; provided, that a decision or judgment with respect to such action or claim will not have any direct or indirect material adverse effect upon the person seeking indemnification. The person seeking indemnification shall be entitled to participation therein after such assumption, the costs of such participation following the assumption to be at its own expense. Upon assuming such defense, the indemnifying party shall agree to be fully responsible for, and to pay, the entire amount of any monetary judgment or settlement, and shall have full rights to enter into any monetary compromise or settlement which is dispositive of the matters involved; provided that such settlement will not have any direct or indirect material adverse effect upon the person seeking indemnification. In the event that the indemnifying party assumes the defense of such action or claim, it shall be conducted by counsel by such party and approved by the party seeking indemnification, which approval shall not be unreasonably withheld.
- B. With respect to actions as to which (i) the indemnifying party does not have the right to assume the defense, or (ii) it shall not have exercised its right to assume the defense, the party seeking indemnification shall assume and control the defense of and contest such action with counsel chosen by it and approved by the indemnifying party, which approval shall not be unreasonable withheld. The indemnifying party shall be entitled to participate in the defense of such action, the cost of such participation to be at its own expense. The indemnifying party shall be obligated to pay the reasonable attorneys' fees and expenses of the party seeking indemnification to the extent that such fees and expenses related to claims as to which indemnification is payable under Section 13.01 or 13.02 hereof, as such expenses are incurred. The party seeking indemnification shall have full rights to dispose of such action and enter into any monetary compromise or settlement, provided, however, that in settling any action in respect of which indemnification is payable under Section 13.01 or 13.02 hereof, it shall act reasonably and in good faith.
- C. Both the indemnifying party and the indemnified party shall cooperate fully with one another in connection with the defense, compromise or settlement of any such claim or action, including without limitation, by making available to the other all pertinent information and witnesses within its control.
- D. Subject to Section 17.12 of this Agreement, in any action relating to a claim for Losses, the parties shall bear such Losses as the law assigns to each.
- 13.04 Environmental Indemnification. The Company shall not cause or permit any Hazardous Substance to be brought upon, kept or used in or about the Facilities, by the Company, its agents, employees, contractors, subcontractors or invitees, except in compliance with all Applicable Laws and with the prior written consent of the City (which the City shall not unreasonably withhold as long as the Company demonstrates to the City's reasonable satisfaction that such Hazardous Substance is necessary or useful to the Company's business and will be used, kept and stored in a manner that complies with all Applicable Laws regulating any such Hazardous Substance so brought upon or used or kept in or about the Facilities. If the Company breaches the obligations stated in the preceding sentence, or if the presence of Hazardous Substances within the Facilities

caused or permitted by the Company results in contamination of the Facilities or the environment, or if contamination of the Facilities or the environment by Hazardous Substances otherwise occurs for which the Company is responsible to the City for damage resulting therefrom, in addition to and not in limitation of Section 13.01 hereof, the Company shall indemnify, defend and hold the City Indemnitees harmless from and against any and all Losses, (including, without limitation, diminution in value of the Facilities and sums paid in settlement of claims, attorney's fees, court costs, consultant fees and expert fees) which arise during or after the Term as a result of such contamination. This indemnification of the City by the Company includes, without limitation, costs incurred in connection with any investigation of site conditions or any cleanup, remedial, removal or restoration work required by any Regulating Entity because of Hazardous Substances present in the soil or ground water on, under or within the facilities or the environment resulting from the Company's breach of its obligations pursuant to this Agreement, or the presence of Hazardous Substances within the Facilities caused or permitted by the Company or contamination of the Facilities or the environment by Hazardous Substances for which the Company is otherwise responsible to the City. Without limiting the foregoing, if the presence of any Hazardous Substances on, under or within the Facilities or the environment caused or permitted by the Company results in any contamination of the Facilities or the environment, the Company shall promptly take all actions at its sole cost and expense as are necessary to return the Facilities to the condition existing prior to the introduction of any such Hazardous Substances to the Facilities and shall promptly take all actions at its sole cost and expense as are necessary to remediate the environment in accordance with all Applicable Laws; provided that the City's approval of such action shall first be obtained, which approval shall not be unreasonably withheld so long as such actions would not potentially have any adverse short-term or long-term effect on the Facilities. The Company's cleanup and remedial obligations as set forth above shall not extend beyond this cleanup and remedial obligations required by any Regulating Entity pursuant to Applicable Law or as otherwise required pursuant to Applicable Law.

Notwithstanding any provision to the contrary contained in this Section 13.04, and provided that the Company is otherwise in compliance with the terms and provisions of this Agreement in all material respects, including without limitation, Sections 10.01 and 10.06 of this Agreement, the Company shall not be liable to the City (i) for any Biologically Toxic Influent contained in the Influent, or (ii) Hazardous Substances anywhere within the subsurface of the Facilities which are existing prior to the Commencement Date to the extent the Company has not aggrayated or exacerbated such pre-existing environmental conditions.

13.05 Remedies Cumulative. The remedies provided for herein shall be cumulative.

13.06 Limitation of Liability.

- A. <u>Disclaimer of Consequential Damages</u>. The Company and the City acknowledge and agree that because of the unique nature of the undertakings contemplated by this Agreement, it is difficult or impossible to determine with precision the amount of damages that would or might be incurred by the City or the Company as a result of a breach of this Agreement. In no event, however, shall the City or the Company be liable for or obligated in any manner, except to the extent of indemnification of claims of third parties pursuant to Section 13.01 hereof, to pay incidental, special, punitive or consequential damages of any nature to the other Party because of a breach of this Agreement, warranty, delay or otherwise, arising out of the performance or nonperformance by the Company or the City, as applicable, or their respective obligations under this Agreement, except where such damages are subject to a claim by the Company for insurance, including coverage pursuant to a policy secured as required by Article 12 of this Agreement.
- B. <u>Damage Cap.</u> Notwithstanding anything to the contrary contained in this Agreement, the aggregate liability of the Company to the City under this Agreement during the Initial Term shall not exceed Twenty One Million Dollars (\$21,000,000), regardless of whether such liability arises out of breach of contract, guarantee or warranty, tort, product liability, indemnity, contribution, strict liability or any other legal or equitable theory; provided, however, that such damage cap shall apply only if the Company has at all times (i) complied with all insurance requirements under Article 12 of this Agreement and (ii) made good faith efforts to demand insurance coverage and proceeds for claims, losses, and damages.
- 13.07 <u>Pre-Existing Conditions</u>. The Company shall not be responsible for any fines, penalties or other costs connected with any complaint or a violation of any law, regulation, guideline, permit, judgment or order in effect or in existence on any date prior to the Commencement Date of this Agreement or which arise from acts or omissions occurring prior to the Commencement Date of this Agreement.
- 13.08 Survival. This Article 13 shall survive the termination or expiration of this Agreement.

ARTICLE 14 - EVENTS OF DEFAULT

- 14.01 Events of Default by the Company. Each of the following shall constitute an Event of Default by the Company after the Commencement Date:
 - A. To the extent such failures or refusals are not otherwise covered in this Section 14.01 persistent and repeated failure or refusal of the Company to operate and maintain the Facilities in accordance with the performance measures set forth in Article 6 of this Agreement and in accordance with the Performance Guarantees set forth in Article 10 of this Agreement or to perform timely any other obligation under this Agreement, unless such failure or refusal is clearly recognized, justified and excused by the terms and conditions of this Agreement.
 - B. Failure of the Company to pay amounts owed to the City under this Agreement within sixty (60) days following the date they become due and owing:
 - C. Failure of the Company to meet any NPDES Permit conditions or requirements on a regular basis, unless such failure is clearly recognized, justified and excused by the terms and conditions of this Agreement;
 - D. Failure to secure and maintain the insurance required under this Agreement as provided in Article 12 hereof;
 - E. Failure to maintain solvency, as determined under the applicable definition of "insolvent" contained in 11 U.S.C.§101(32), as amended. The occurrence of any of the following are deemed a failure to maintain solvency;
 - Inability, failure or refusal to pay debts as they mature; entry into an arrangement by the Company with or for the benefit of their creditors; the Company's to or acquiescense in the appointment of a receiver, trustee or liquidator for a substantial part of the Company's property; or
 - 2. A bankruptcy, winding up, reorganization, insolvency, arrangement or similar proceeding instituted by or against the Company under the laws of any jurisdiction, which proceeding is not dismissed within sixty (60) days; or
 - Any action or answer in a bankruptcy, winding up, reorganization, insolvency, arrangement or similar proceeding in which the Company approves of, consents to, or acquiesces in any such proceeding; or
 - 4. The levy of any distress, execution or attachment upon the property of the Company which shall substantially interfere with its performance hereunder; provided, however, that with respect to the Company only, this form of insolvency shall not be deemed to have occurred if the insolvency is caused primarily by the City's failure to make a payment due pursuant to Article 8 hereof within forth-five (45) days of when it becomes due and payable.
- 14.02 Events of Default by the City. Each of the following shall constitute an Event of Default on the part of the City after the Commencement Date:
 - A. To the extent such failures or refusals are not otherwise covered in this Section 14.02, persistent and repeated failure or refusal of the City to perform timely and material obligations under this Agreement unless such failure or refusal is clearly recognized, justified and excused by the terms of and conditions of the Agreement; or
 - B. Failure of the City to pay undisputed amounts owed to the Company under this Agreement within ninety (90) days following the time they become due and payable.
- 14.03 <u>Default Notices: Opportunity to Cure.</u> With the exception of a termination for the reason described in Section 15.01 (B) hereof, this Agreement shall not be terminated for an Event of Default unless and until (i) the Party contemplating termination gives the offending Party written notice in reasonable detail of each Event of Default, the offending Party is alleged to have committed or permitted, and (ii) the offending Party shall have failed to cure such Event of Default within thirty (30) days (or such longer period as may reasonably be

required to diligently effect such cure) following delivery of the Default Notice to the offending Party. Notwithstanding the foregoing, if there are repeated Company Defaults under Sections 14.01 (A) through (D), then, regardless of attempts by the Company to cure the same, the City, in its sole discretion, may terminate this Agreement without giving a Default Notice or affording the Company a period to cure.

ARTICLE 15 - TERMINATION

15.01 Termination of Agreement for a Company Default.

- A. If the City gives the Company notice pursuant to Section 14.03 hereof of the occurrence of a Company Default under Section 14.01 (A) or (B) hereof, and the Company Default is not cured within the period set forth in Section 14.03 hereof, the City may terminate this Agreement.
- B. Upon the occurrence of a Company Default under one or more of Sections 14.01 (C), (D), or (E) hereof, the City may terminate this Agreement immediately by delivery of a Default Notice to the Company.
- 15.02 <u>Termination of Agreement for a City Default</u>. If the Company gives the City a Default Notice pursuant to Section 14.03 hereof of the occurrence of a City Default, and such City Default is not cured within the period set forth in Section 14.03 hereof, the Company may terminate this Agreement.
- 15.03 Termination for Labor Unrest. If, on or after the Commencement Date, personnel employed by the Company and performing services pursuant to the Company's obligations under this Agreement shall go on a labor strike or slowdown, or if a work stoppage, walkout or secondary boycott shall occur, for any reason or cause whatsoever, and such act or event effectively prevents the Company from performing its material obligations under this Agreement, the City, during the pendency of the period in which performance is prevented, may, in its sole discretion, by notice to the Company, terminate this Agreement immediately.
- 15.04 <u>Termination for Uncontrollable Circumstances</u>. If an Uncontrollable Circumstance shall occur after the Commencement Date relative to a material obligation of the Company or the City under this Agreement and such Uncontrollable Circumstance or the effect thereof prevents performance of such material obligation for a period of thirty (30) days, the City and the Company shall, during or after such thirty (30) day period, meet to review the situation. If, despite the good faith efforts of the Parties to reach an agreement, no agreement is reached within a reasonable time considering the nature and extent of the Uncontrollable Circumstance, then either Party may terminate this Agreement upon notice to the other Party.
- 15.05 <u>Termination for Insufficient Funding</u>. In the event sufficient funds to pay the Service Fee are unavailable, through the failure of any entity to appropriate funds or otherwise, the City shall have the right to terminate the Agreement upon thirty (30) days prior notice.
- 15.06 <u>Termination for Breach of Assignment Provision</u>. The Company will not (i) assign or transfer this Agreement or its right, title or interest or obligations therein, in whole or in part, or (ii) voluntarily or involuntarily undergo a Change in Control without, in each instance, the City's advance written approval, which the City has the sole discretion to withhold. If assignment of this Agreement is to an affiliate, subsidiary, or related entity of the Company, the City will not unreasonably withhold consent. Violation of this Section 15.06 will constitute a breach of this Agreement and the City may, in its sole discretion, terminate this Agreement. All rights, title and interest of the Company will thereupon cease and terminate.
- 15.07 Remedies of the City. If the City terminates this Agreement pursuant to Section 15.01, 15.03 or 15.06 hereof, the City Shall have the right to seek legal and equitable remedies provided by law. If the City shall terminate this Agreement pursuant to Section 15.03 hereof, the City shall pay the Company in addition to those payments and reconciliation amounts specified in Section 15.11 hereof, the Company's out-of-pocket costs as of the date of termination, which shall in no event be greater than the amount of the Fixed Fee of the Service Fee for one Billing Month determined using the Billing Month of the termination. If the City shall terminate this Agreement pursuant to Section 15.04 (unless there has been no meeting) or 15.05 hereof, or for failure of the Parties to agree on a renegotiation of this Agreement, such termination shall be treated as a termination for convenience pursuant to Section 15.09 of this Agreement.
- 15.08 Remedies of the Company. If the Company terminates this Agreement pursuant to Section 15.02 or 15.03 hereof, the City shall pay the Company the payments and reconciliation amounts specified in Section 15.11 hereof.

- 15.09 <u>Termination for Convenience</u>. The City may terminate this Agreement in its sole discretion for its convenience and without cause at any time upon one hundred twenty (120) days prior written notice to the Company. If the City exercises its right of convenience termination, the City shall not be liable to the Company for demobilization costs, termination fees or any other costs or expenses except for:
 - A. With respect to any convenience termination occurring during the Initial Term, the amount of Ninety Thousand Dollars (\$90,000) per year remaining on the Initial Term pro rata; and
 - B. With respect to any convenience termination occurring during the Extended Term, the amount of One Hundred Fifteen Thousand Dollars (\$115,000) per year remaining on the Extended Term pro rata.

15,10 Operations Cooperation and Transfer of Personnel.

- A. If the City or the Company terminates this Agreement, the Company shall, from the date of the notice of termination make fully available its managers and employees performing services at the Facilities for at least six (6) months after the Termination Date pursuant to this Section 15.10 to continue to perform all the operation, maintenance, repair and management services contemplated in this Agreement. The City may determine that it requires a lesser amount of services, managers, employees and intellectual property in order to provide a smooth and orderly transition of the operation and maintenance of the Facilities to City administrators, managers and personnel or, as applicable, the City's contracted private company; provided, however, in no event shall such provision of services by the Company exceed the tenth (10th) anniversary date of this Agreement as measured from the Commencement Date. The Company shall immediately transfer to the City all intellectual property owned by the City and used or created by the Company during the Term of this Agreement, including, but not limited to, the City's licenses, data, source codes and software, used in, updated or created for the operation of the Facilities. The Company shall fully cooperate with the City to effectuate such a transition, including the provision of training and "know-how" in the procedures and techniques employed by the Company in meeting its obligations under Articles 6 and 10 hereof.
- B. Notwithstanding the termination of this Agreement, the City shall compensate the Company for performing the services specified in Section 15.10 (A) hereof on a daily basis in an amount equal to the daily allocated cost of such services calculated on the basis of the Fixed Fee of the Service Fee for the last full Billing Month immediately prior to the Termination Date; provided, however, such daily portion of the Fixed Fee shall be reduced on a pro rata basis to reflect the number of Company employees performing services and the operation and maintenance services performed (the management fee and profit shall be reduced pro rata to reflect the reduction in personnel and services) on a daily basis. The Company shall invoice the City for such Fixed Fee as calculated pursuant to this Section 15.10 (B) within fifteen (15) days after the end of each month after the Termination Date, and the City shall pay to the Company the amount due and owing pursuant to this Section 15.10(B) within forty-five (45) days thereafter. The Company shall comply with the invoicing and date and information provisions of this Agreement in submitting any such invoice to the City.
- C. Upon receipt of notice of termination, the Company shall, at the option of the City, cancel outstanding commitments for procurement of services, materials and supplies. In addition, the Company must exercise all reasonable diligence to cancel or divert to other activities its outstanding commitments for procurement of personal services, if the City, in its sole discretion so requires. If, after serving notice of termination for nonperformance or default, the City determines that the reasons for nonperformance or default are excusable and are not the fault of and beyond the control of the Company, the City may, in its sole discretion, authorized the Company to resume work.
- D. The Company recognizes and understands that the transition outlined in this Section 15.10 may well result in the City employing or attempting to employ some or all of the managers or personnel employed by the Company and performing services at the Facilities. The Company shall have no covenant not to compete or other restrictions on the City hiring Company employees working on the Facilities.
- E. Upon the termination or expiration of this Agreement, the Company shall assign to the City its interest in all contracts entered into by the Company relative to the Facilities if requested by the City, if such contracts do not prohibit such assignment. The City's right to request assignment of certain contracts shall not be read as an obligation by the City to assume all or any of such contracts. The City shall, however, assume the payment and performance of all contracts assigned to it and shall pay any penalties and costs incurred by the Company with respect to the assignment of such contracts. The Company shall

- exercise all reasonable efforts in negotiating contracts relative to the Facilities to (i) obtain the written consent of the other parties to such contracts to the assignment by the Company of its rights therein to the City and (ii) secure contract terms and conditions that do not include damages or penalties to any assignee with respect to any assignment.
- F. In the event of a Company Default, the City may, in its discretion, determine to perform any Company obligation under this Agreement that the Company has failed to perform. The City may issue a Default Notice informing the Company of the City's intent to perform such obligation(s). The Company shall be obligated to reimburse the City for all costs the City incurs in the performance of such Company obligation(s). The City's performance under this paragraph (F) shall not effect a cure of the Company Default; such cure period shall be tolled during the period the City is performing the Company's obligations.
- G. If the City has notified the Company of a Company Default and the City determines, in its sole discretion, that the public safety is threatened by the Company Default, the City may assume operation of the Facilities pending termination of this Agreement and direct Company employees, or contract with others, to take such actions as the City deems necessary or appropriate to ensure the continuity of services or protect the public safety. Damages incurred by the City in such respect shall be paid by the Company.
- 15.11 Manner of Termination Payment. All performance and payment obligations under this Agreement, including payment of the Service Fee that is due and owing, shall continue pursuant to the terms of this Agreement until this Agreement terminates and any amount accrued but unpaid prior to termination shall, if due and owing, be payable in accordance with this Section 15.11. Except as otherwise specifically provided in this Agreement with respect to the time of payment following termination, within ninety (90) days following termination of this Agreement, the City and the Company shall reconcile all amounts then due and payable to each other under the terms of this Agreement. Upon reaching, as a result of such reconciliation, the total amount of the outstanding unpaid balance which the City and the Company owe the other, the City and the Company shall, within thirty (30) days thereafter, make the final payments in complete discharge of their obligations under this Agreement, except those obligations which survive the termination or expiration of this Agreement. Payment obligations under this Section 15.11 are subject to Sections 8.05 and 8.06 hereof.
- 15.12 Exclusive Remedies. The remedies specifically set forth in this Agreement are exclusive, and the Parties waive any other remedies they may have at law or in equity subject among other sections to Section 15.07; provided, however, that either Party may seek judicial enforcement of any remedy provided herein and any amounts payable hereunder. The Parties agree and acknowledge that the damages provided for in this Article 15 are to be liquidated damages and shall be the sole and exclusive measure of damages or liability for termination of this Agreement by a Party under this Article 15 and that the provisions for damages set forth herein are intended to measure as accurately as possible the direct damages of the Party entitled to such damages and are not intended to include punitive, special, consequential or incidental damages.
- 15.13 Survival. This Article 15 shall survive the termination or expiration of this Agreement.

ARTICLE 16 - DISPUTE RESOLUTION

- 16.01 <u>Dispute Resolution</u>. To the extent the Company and the City cannot, after good faith attempts, resolve any controversy or dispute that may have arisen under this Agreement, either the Company or the City, to the extent its respective interest is adversely affected, may refer the matter to mediation in the State, the cost of which shall be borne equally by the City and the Company. Any litigation commenced after mediation must be commenced in the Ohio court system and filed in the County, exclusive of any other judicial forum that may have jurisdiction over the matter.
- 16.02 <u>Covenant to Perform</u>. The Company and the City shall continue to perform their respective obligations under this Agreement, including payment obligations, except as provided in Section 8.06 hereof, without interruption or slowdown, pending resolution of any dispute(s), unless the matter at issue precludes such continued activity.
- 16.03 Survival. This Article 16 shall survive the termination or expiration of this Agreement.

ARTICLE 17 - MISCELLANEOUS

17.01 Assignment. The City reserves the right to assign its rights and obligations under this Agreement to any

validly constituted agency, department or authority of the State, or a duly created municipal corporation or authority. The City will provide the Company with prior notice of such an assignment. Such assignee will have full authority to enforce and manage this Agreement, unless otherwise specified by the City.

Except as permitted above, this Agreement shall not be assigned by either Party without the prior written consent of the other Party. The Company shall not assign or transfer this Agreement or its rights, title or interests or obligations under this Agreement, in whole or in part, without in each instance the City's prior written consent, which consent may be withheld by the City in its sole discretion; provided, however, that the Company may assign its interest without such consent to any Affiliate, successor or parent of the Company if (i) the Company shall remain liable for all obligations under this Agreement, and (ii) the Guarantors, pursuant to the Guarantee, fully guarantee the performance of such assignee's obligations under this Agreement. Violation of the terms of this paragraph will constitute a breach of this Agreement and the City may, in such event and within its discretion, cancel this Agreement upon written notice. All right, title and interests of the Company will thereupon cease and terminate. It is understood and agreed between the Parties that this Section 17.01 shall not be construed or interpreted to restrict the Company's ability to employ Subcontractors in connection with the performance of portions of its obligations hereunder.

- 17.02 <u>Further Assurances</u>. Each Party agrees to execute and deliver any instruments and to perform any acts that may be necessary or reasonably requested in order to give full effect to this Agreement. The City shall execute such further instruments and documents and take such action as may be reasonably requested by the Company and not inconsistent with the provisions of this Agreement and not involving the assumption of obligations other than those provided for in this Agreement to carry out the intent of this Agreement.
- 17.03 Relationship of Parties. Except as otherwise explicitly provided herein, neither the Company nor the City shall have any responsibility whatsoever with respect to services provided or contractual obligations assumed by the Company or the City, respectively, and nothing in this Agreement shall be deemed to constitute the Company or the City a partner, agent, employee or legal representative of any other party or to create any fiduciary relationship, partnership or joint venture between or among the Parties. The Parties agree that the Company has entered into this Agreement and shall be performing the services contemplated herein as an independent contractor. As an independent contractor the Company is solely responsible for the means, methods, techniques, procedures and schedules used to perform the work. The Company has the sole right to control and direct the means, manner and method by which the obligations of this Agreement are satisfied.

Nothing in this Agreement may be interpreted to mean the City may exercise control over how services are provided by the Company nor how the Company satisfied its obligations under this Agreement. Nothing in this Agreement may be interpreted to give the appearance that the Company possesses the apparent or actual authority to act or speak for the City and the Company shall not by words, acts or representations convey to the general public, any Person or any governmental unit the impression that the Company has the authority to speak or act for the City. If any Person believes that the Company has the necessary power to bind the City or believes that the City has the power to control how services are provided by the Company, the Company shall take actions as are necessary to correct the erroneous inferences and prevent reliance on such a mistake of fact.

17.04 Cost Substantiation.

- A. Cost Substantiation Generally. The Company shall provide Cost Substantiation for the cost for which the City is financially responsible hereunder, other than the Fixed Fee of the Service Fee. In the case of Cost Substantiation for Subcontractors, the Company shall use its best efforts to provide all information requested by the City. In incurring costs which are or may be subject to Cost Substantiation, the Company shall utilize competitive practices to the maximum reasonable extent (including, where practicable and except with respect to costs of the Company to which the Service Fee applies, obtaining three (3) bids or utilizing a contractor selection process in compliance with Applicable Law to complete a Capital Improvement with costs that are expected to be in excess of the Capital Improvement Threshold Amount), and shall enter into subcontracts on commercially reasonable terms and prices in light of the work to be performed and the City's potential obligation to pay for it.
- B. Costs Requiring Cost Substantiation. Cost Substantiation shall be provided as soon as reasonably practicable after the costs which require substantiation have been incurred by the Company. Examples of costs which require substantiation include (i) work done on an emergency basis to respond to an Uncontrollable Circumstance, where it is not reasonably practicable for the parties in advance to negotiate price for the work, (ii) Capital Improvements, and (iii) other Additional Services. Cost Substantiation

shall also be required if the parties agree that the Company shall perform work on a cost-plus basis.

C. Cost Substantiation Certificate. Any certificate delivered hereunder to substantiate cost shall be signed by the Project Manager for the Company, shall state the amount of such cost and the provisions of this Agreement under which such cost is chargeable to the City, shall describe the competitive or other process utilized by the Company to obtain the commercially reasonable price, shall state that such cost is a fair market price for the service or materials supplied and shall state that such services and materials are reasonably required pursuant to this Agreement. The Cost Substantiation certificate shall be accompanied by copies of such documentation as shall be necessary to reasonable demonstrate that the cost as to which Cost Substantiation is required has been paid or incurred. Such documentation shall be in a format reasonably acceptable to the City and shall include reasonably detailed information concerning all subcontracts and, with respect to self-performed work, (i) the amount and character of materials, equipment and services furnished or utilized, the persons from whom purchased, the amounts payable therefor and related deliver and transportation costs and any sales or personal property taxes; (ii) a statement of the equipment used and any rental payable therefor; (iii) Company employee hours, duties, wages, salaries, benefits and assessments; and (iv) Company and Subcontractor profit, administration cost, bonds, insurance, taxes, premiums overhead, and other expenses. The Company's entitlement to reimbursement of Cost Substantiated costs of the Company shall be subject to the limitations set forth in this Section 17.04.

Company billing rates shall include actual labor costs and all benefits, and shall not be marked up except as provided in 17.04 (E) hereof.

- D. <u>Technical Service</u>. Company personnel and personnel of Subcontractors providing technical services shall be billed at their then currently applicable rates for similar services on projects of similar size and scope to the Services. The Company shall use commercially reasonable efforts to use available Company personnel for additional work hereunder before using Subcontractors.
- E. Mark-Up. On all costs incurred by the Company for work performed directly by the Company or any of its Affiliates which are subject to Cost Substantiation, the Company shall be entitled to a mark-up not to exceed ten percent (10%). The price payable to all Subcontractors, including Subcontractor overhead and mark-ups for risk and profit, shall be commercially reasonable. No mark-up will be added to the Company's or Subcontractor's costs for lodging, meals, travel, labor costs of Company employees, or the purchase of Vehicles.
- F. Evidence of Costs Incurred. To the extent reasonably necessary to confirm direct costs required to be Cost Substantiated, copies of time sheets, invoices, canceled checks, expense reports, receipts and other documents, as appropriate, shall be delivered to the City with the request for reimbursement of such costs.
- 17.05 <u>Public Records Law.</u> The Company recognizes the requirements of Ohio Revised Code Section 149.43 *et seq.*, otherwise known as the "Ohio Public Records Law," and fully waives and releases any claim against the City and any of its officers and employees relating to the release of any of its documents or information. Furthermore, the Company agrees to defend, indemnify, and hold the City and its officers and employees harmless from any and all claims arising from the release of any document or information made available to the City unless subject to lawful exemption.
- 17.06 Municipal Income Tax. The Company, if located within the City or doing business within the City, shall withhold all City income taxes due or payable under the provisions of the Municipal Income Tax ordinance for wages, salaries, and commissions paid to its employees pursuant Chapter 880, Section 880.17 of the Codified Ordinances of the City of Strongsville. The Company shall likewise require its subcontractors to withhold any such City income taxes due for services performed under this Agreement.

17.07 Notices and Authorized Representatives.

A. All notices, requests, demands and other communications required or permitted under this Agreement shall be in writing and shall be deemed sufficient and to have been received (i) when delivered by personal delivery, or (ii) upon delivery by a nationally recognized overnight courier service that provides evidence of delivery, or, if refused, upon the first date of attempted delivery, or (iii) when five (5) days have lapsed after its transmittal by registered or certified mail, postage prepaid, return receipt requested, addressed to the Party to whom directed or that Party's address as it appears below or another address of which that Party has given notice, or (iv) when delivered by facsimile transmission and a copy thereof is

also delivered in person or by overnight courier, or (v) when delivered electronically and confirmation of receipt is provided by the Party to whom the notice is directed.

To the Company:
Christopher Earle
Regional Vice President
Veolia Water North America – Central, LLC
700 East Butterfield Road, Suite 201
Lombard, Illinois 60148
Christopher,earle@veolia.com

With a copy to the General Council: General Legal Counsel Veolia Water North America - Central, LLC 53 State Street, 14th Floor Boston, Massachusetts, 02109 george.kidd@veolia.com

To the City:
City of Strongsville
Department of Public Service
Attn: Director of Public Service
16099 Foltz Parkway
Strongsville, Ohio 44149
servicedept@strongsville.org

With a copy to the Law Director: City of Strongsville 16099 Foltz Parkway Strongsville, Ohio 44149 strongsville.law@strongsville.org

Changes in the respective addresses to which such notices may be directed may be made from time to time by any Party by written notice to the other party.

B. For purposes of this Agreement, the Authorized Representatives are as follows:

For the Company: Richard Meloy, Project Manager

For the City: Joseph Walker, Strongsville Director of Public Service

Either Party may change its Authorized Representative at any time by written notice to the other Party.

- 17.08 Waiver. Neither the failure nor any delay on the part of any Party to exercise any right, remedy, power or privilege under this Agreement shall operate as a waiver thereof, and the waiver by either Party of a default or a breach of any provision of this Agreement by the other Party shall not operate or be construed to operate as a waiver of any subsequent default or breach. The making or the acceptance of a payment by either Party with knowledge of the existence of a default or breach shall not operate or be construed to operate as a waiver of such default or breach or any subsequent default or breach.
- 17.09 Entire Agreement; Modification and Amendments. This Agreement (and the Exhibits attached hereto) contain the entire understanding between the Parties with respect to the subject matter hereof, and supersedes all prior and contemporaneous agreements and understanding, inducements and conditions, expressed or implied, oral or written, except as herein contained. The express terms hereof control and supersede any course of performance or usage of the trade inconsistent with any of the terms hereof. This Agreement (including the Exhibits attached hereto) may only be modified or amended by an Amendment or Memorandum of Understanding approved by the City Council in accordance with law. The Company and the City shall be bound by any such Amendment or Memorandum of Understanding.
- 17.10 Governing Law. This Agreement and all questions relating to its validity, interpretation, performance and enforcement shall be governed by and construed in accordance with the laws of the State of Ohio.
- 17.11 Consent to Jurisdiction. The Company hereby irrevocably consents to the Cuyahoga County Common Pleas Court being the venue for all legal actions, in connection with any action or proceeding arising out of or relating to this Agreement.
- 17.12 Non-Waiver of Immunity. Nothing in this Agreement shall be construed to constitute a waiver of any immunity or defense conferred by law upon or otherwise available to the City, its departments, officers, employees and duly authorized agents as against the Company or any Person.
- 17.13 <u>Counterparts.</u> This Agreement will be executed by two counterparts, each of which shall be deemed to be an original. All such counterparts shall together constitute but one and the same instrument.
- 17.14 Severability. The provisions of this Agreement and of each section or other subdivision herein are

independent of and separable from each other, and no provision shall be affected or rendered invalid or unenforceable by virtue of the fact that for any reason any other or others of them may be invalid or unenforceable in whole or in part unless this Agreement is rendered totally unenforceable thereby.

- 17.15 <u>Liability of Officers and Employees</u>. No director, officer, agent, consultant, representative or employee of either Party shall be charged personally by the other or held contractually liable thereto under any term or provision of this Agreement, merely because of either Party's execution or attempted execution or because of any breach or alleged breach thereof; provided, however, that all Persons remain responsible for any of their own criminal actions.
- 17.16 Third Party Beneficiary. This Agreement is intended to be solely for the benefit of the Company and the City and their successors and permitted assigns and is not intended to and shall not confer any rights or benefits on any third party not a signatory hereto, except as specifically set forth herein.

This Agreement and all of the covenants hereof shall inure to the benefit of and be binding upon the City and the

Company respectively and their respective partners, successors, permitted assigns and legal representatives.

IN WITNESS WHEREOF, City and Company have signed this Agreement in two counterparts. One counterpart each has been delivered to City and Company.

Attest:	(s)	CITY: City of Strongsville, Ohio
Name:	Joseph Walker	By:(s)
Title:	Director of Public Service	Name: Thomas P. Perciak
		Title: Mayor
		Date:
	(SEAL)	Address: 16099 Foltz Parkway
		Strongsville, Ohio 44149 (Attach evidence of authority to sign)
Attest:	(s)	COMPANY: Veolia Water North America-Central LLC
Name:		By:(s)
Title:		Name: Jason R. Costa, Principal
		Title: Vice President
		Date:
	(SEAL)	Address: 53 State Street, 14th Floor
		Boston, MA 02109
		(If Contractor is a corporation, a partnership or a joint venture, attach evidence of authority to sign)

CERTIFICATE OF ASSISTANT SECRETARY OF

VEOLIA WATER NORTH AMERICA-CENTRAL, LLC

The undersigned, Whitney Fawcett, Assistant Secretary of Veolia Water North America-Central, LLC, a Delaware limited liability company (the "Company"), does hereby certify that Jason Costa is the duly elected and acting Vice President of the Company and in such capacity is authorized to execute documents and make commitments with regard to the following:

WASTEWATER FACILITIES OPERATIONS, MAINTENANCE AND MANAGEMENT SERVICES CITY OF STRONGSVILLE, OHIO

IN WITNESS WHEREOF, I have hereunto set my hand and affixed the seal of the Company this 7th day of August, 2024.

DocuSigned by:

Whitney Fawcett, Assistant Secretary

1981A032DF2C40F...

SEAL

CITY'S DIRECTOR OF FINANCE CERTIFICATION

Title: Law Director

Date:

CITY OF STRONGSVILLE, OHIO

ORDINANCE NO. 2024 - 136

By: Mayor Perciak and All Members of Council

AN ORDINANCE AMENDING SECTIONS 1242.07(b)(10) and 1258.06(a)(6), OF TITLE SIX OF PART TWELVE-PLANNING AND ZONING CODE OF THE CODIFIED ORDINANCES OF THE CITY OF STRONGSVILLE IN ORDER TO UPDATE THE REGULATION OF CERTAIN ESTABLISHMENTS AS CONDITIONAL USE PERMITS IN CERTAIN DISTRICTS, AND DECLARING AN EMERGENCY.

BE IT ORDAINED BY THE COUNCIL OF THE CITY OF STRONGSVILLE, COUNTY OF CUYAHOGA AND STATE OF OHIO:

Section 1. That Section 1242.07(b)(10) of Chapter 1242 of Title Six of Part Twelve-Planning and Zoning Code of the Codified Ordinances of the City of Strongsville concerning Conditional Use Permits, be and is hereby amended to read as follows:

1242.07 CONDITIONAL USE PERMITS.

- (b) <u>Standards for Evaluating Conditional Use Permits</u>. An application for a conditional use permit shall not be approved unless the following conditions and standards are complied with as set forth for the following districts:
- (10) <u>Pawn Shops, Paraphernalia Stores, Vape Shops, Vaporizer Stores, Tattee Parlors, Hookah Lounges, Smoke Lounges, and Vapor Lounges</u>. In addition to complying with all other requirements of this Zoning Code, these uses shall comply with the following requirements and standards:

A. No such use shall be established or operated within 500 feet of a school.

B. No such use shall be established or operated within 500 feet of an existing use of the same specific type.

C. No such use shall be operated or open for business between the hours of 12:00 midnight and 8:00 a.m.

(11) <u>Safeguards and conditions</u>. In addition to complying with the above general standards set forth in this section, conditions appropriate to each particular application may also be set forth in the permit.

(12) Approval. The approval of a conditional use permit shall become null and void if the construction of the building or site improvements are not started within a six-month period after date of approval.

(Ord. 2016-158. Passed 11-21-16.)

Section 2. That Section 1258.06(a)(6) of Chapter 1258 of Title Six of Part Twelve-Planning and Zoning Code of the Codified Ordinances of the City of Strongsville concerning Use Regulations; Restaurant-Recreational Services District, be and is hereby amended to read as follows:

1258.06 USE REGULATIONS; RESTAURANT-RECREATIONAL SERVICES DISTRICT.

Buildings and land shall be used and buildings shall be designed, erected, altered, moved or maintained in whole or in part in Restaurant-Recreational Services Districts only for the uses set forth in the following schedules and regulations:

(a) Main Buildings and Uses Permitted.

(6) Tattoo parlors, hHookah lounges, smoke lounges, vapor lounges, or other similar services, provided a conditional use permit is granted in accordance with the standards set forth in Section 1242.07 of this Zoning Code.

(Ord. 2016-158. Passed 11-21-16.)

Section 3. That it is found and determined that all formal actions of this Council concerning and relating to the adoption of this Ordinance were adopted in an open meeting of this Council; and that all deliberations of this Council, and any of its committees, that resulted in such formal action were in meetings open to the public in compliance with all legal requirements.

Section 4. That this Ordinance is hereby declared to be an emergency measure necessary for the immediate preservation of the public peace, property, health, safety and welfare of the City, and for the further reason that it is necessary to provide for the appropriate regulation of certain establishments as conditional permitted uses in certain districts, and to ensure the safety of the general public. Therefore, provided this Ordinance receives the affirmative vote of two-thirds of all members elected to Council, it shall take effect and be in force immediately upon its passage and approval by the Mayor; otherwise from and after the earliest period allowed by law.

First reading:	Referred to Planning Commission
Second reading:	
Third reading:	Approved:
Public Hearing:	
President of Council	Approved:Mayor
Date Passed:	Date Approved:

CITY OF STRONGSVILLE, OHIO ORDINANCE NO. 2024 – <u>136</u> Page 3

	<u>Yea</u>	<u>Nay</u>	Attest:Clerk of Council
Carbone Clark Kaminski Kosek Roff Short Spring			Ord. No 2024 - /3
			Public HrgRef:_ Adopted: Defeated:

CITY OF STRONGSVILLE

CONFIDENTIAL MEMORANDUM

DATE:

September 9, 2024

TO:

President and Members of Council

CC:

Mayor Thomas P. Perciak

Ted Hurst, Building Commissioner Charles Goss, Director of Public Safety Thomas O'Deens, Chief of Police

Brent Painter, Director of Economic Development

George Smerigan, City Planner Daniel J. Kolick, Asst. Law Director

FROM:

Neal M. Jamison, Law Director No.

SUBJECT:

Tattoo Parlors – Amendment of SCO Sections 1242.07 and 1258.06

THIS IS A CONFIDENTIAL COMMUNICATION WITHIN THE ATTORNEY-CLIENT PRIVILEGE AND IS NOT INTENDED FOR COPYING OR REVIEW BY ANYONE OTHER THAN THOSE IDENTIFIED IN THE CAPTION ABOVE.

As I informed you at the recent Council Meeting, the Administration has been working on an issue that has arisen in regard to allowing businesses in our City that utilize permanent make-up services. The use and scope of permanent make-up is not only used for cosmetic reasons. These techniques have improved and evolved over the years so that they are now used to help those individuals who are recovering from certain illnesses. This would include persons who have had significant hair loss who have undergone chemotherapy treatment while battling various forms of cancer. Permanent make-up is also used for those individuals who have suffered severe burns on their body, as well as persons who were born with a birth defect, like a cleft lip, or with alopecia. For all of these reasons, the use of permanent make-up procedures has become more prevalent and accepted in our culture.

The challenge that has arisen is the fact that permanent make-up procedures are considered a "tattoo" under Section 820.01(d) of the Strongsville Codified Ordinances. That section states as follows:

"Tattoo," "tattooed" or "tattooing" refers to any method of placing designs, letters, scrolls, figures, symbols or any other marks upon or under the skin with ink or any other substance resulting in the coloration of the skin by the aid of needles or any other instrument designed to touch or puncture the skin.

Permanent make-up procedures clearly fall within this definition. At the present time, tattoo services are only permitted in the Restaurant-Recreational Services (R/RS) zoning district with a conditional use permitted. We have a limited number of areas in the City that are zoned R/RS and currently these areas are almost fully occupied.

We recently had a young lady apply to open a salon that would provide permanent makeup procedures but she was denied a permit to do so because the location she had obtained was not zoned R/RS. This person appealed to the Board of Zoning Appeals and they denied her appeal because of this zoning restriction. I would point out that the Board of Zoning Appeals was sympathetic to her cause but they felt compelled to follow our zoning code,

The Administration initially thought that we could address this issue by amending Section 820.01(d) of the Strongsville Codified Ordinances by exempting permanent make-up procedures from this definition. In thinking this through, we concluded, however, that certain businesses that actually do perform tattoos would try to use this exemption to argue that they would not have to get a license under Chapter 820 of our code. We still want to license these establishments because they provide certain guidelines for their operation, and we have the ability to revoke their license if they are not in compliance. In addition, we think that businesses that perform permanent make-up procedures should also be licensed under Chapter 820 to give us the same ability to make sure they are compliant with the various rules and guidelines set forth therein, including certain sanitary and safety procedures.

The solution that we propose is to amend two sections of our zoning code that restrict "tattoo parlors" to the R/RS zoning district. Please find enclosed a copy of the proposed Ordinance to amend these two sections. If enacted, this Ordinance would amend our zoning code to allow tattoo operations, as well as permanent make-up procedures, in certain business districts in the City.

If you should have any questions, please feel free to contact me.

NMJ

Enclosure

CITY OF STRONGSVILLE, OHIO

ORDINANCE NO. 2024 - 137

By: Mayor Perciak and All Members of Council

AN ORDINANCE AUTHORIZING AND CONSENTING TO COLUMBIA TOWNSHIP PREPARING AND SUBMITTING AN APPLICATION FOR STATE OF OHIO ISSUE 1 FUNDING FOR THE ASPHALT RESURFACING OF NORTH MARKS ROAD FROM SNELL (ALBION) ROAD TO SPRAGUE ROAD, IN THE CITY OF STRONGSVILLE AND COLUMBIA TOWNSHIP; AUTHORIZING AND DIRECTING THE MAYOR TO EXECUTE A COOPERATION AGREEMENT REGARDING THE PROJECT; AND DECLARING AN EMERGENCY.

WHEREAS, North Marks Road is a county road located partly within the boundaries of the City of Strongsville (Cuyahoga County) and partly within Columbia Township (Lorain County); and

WHEREAS, the City of Strongsville and Columbia Township are desirous of making application to the State of Ohio for Issue 1 funding in connection with the asphalt resurfacing of North Marks Road from Snell (Albion) Road to Sprague Road (hereinafter referred to as the "North Marks Road Resurfacing Project"); and

WHEREAS, Columbia Township has agreed with the City of Strongsville to make a joint application to the State of Ohio for Issue 1 funding, on behalf of both entities, in connection with the North Marks Road Resurfacing Project, and also to mutually cooperate in seeking additional funding from Cuyahoga County; and

WHEREAS, it has been agreed that Columbia Township will be the lead agency in connection with the Project, and the parties, therefore, shall execute a Cooperation Agreement.

NOW, THEREFORE, BE IT ORDAINED BY THE COUNCIL OF THE CITY OF STRONGSVILLE, COUNTY OF CUYAHOGA AND STATE OF OHIO:

- **Section 1.** That the City of Strongsville consents to and authorizes Columbia Township to prepare and submit an application to the Ohio Public Works Commission for Issue 1 funding for the North Marks Road Resurfacing Project; and for the Township as lead agency to execute all documents and do all things necessary in furtherance thereof.
- **Section 2.** That the parties will execute a Cooperation Agreement substantially in the form attached as Exhibit 1, which is incorporated herein and approved, but subject to final adjustments, if any, by the Law Director.
- **Section 3.** That, if such application is approved, it is the intent of this Council to pay for the City of Strongsville's portion of the costs in connection with such Project from the General Capital Improvement Fund.
- **Section 4.** That the Clerk of Council be and is hereby authorized and directed to forward a certified copy of this Ordinance to the Service Director and/or other designated official representative of Columbia Township.

CITY OF STRONGSVILLE, OHIO ORDINANCE NO. 2024 – <u>137</u> Page 2

Section 5. That it is found and determined that all formal actions of this Council concerning and relating to the adoption of this Ordinance were adopted in an open meeting of this Council; and that all deliberations of this Council, and any of its committees, that resulted in such formal action were in meetings open to the public in compliance with all legal requirements.

Section 6. That this Ordinance is hereby declared to be an emergency measure immediately necessary for the preservation of the public peace, health, safety and welfare of the City, and for the further reason that the preparation and submission of the application for funding is immediately necessary in order to meet the State of Ohio application filing deadline, provide for improved highways and more efficient traffic flow, and conserve public funds. Therefore, provided this Ordinance receives the affirmative vote of two-thirds of all members elected to Council, it shall take effect and be in force immediately upon its passage and approval by the Mayor; otherwise from and after the earliest period allowed by law.

	President of	Council	Approved:	Mayor	_
Date Passed	:		Date Approved:		
	<u>Yea</u>	<u>Nay</u>	Attest:Clerk	of Council	_
Carbone Clark Kaminski Kosek Roff Short Spring			Ord. No. <u>2024-13</u> 1 st Rdg 2 nd Rdg 3 rd Rdg	Ref:	
			Public Hrg Adopted:	Ref: Defeated:	

COOPERATION AGREEMENT BY AND BETWEEN THE CITY OF STRONGSVILLE AND COLUMBIA TOWNSHIP

This Agreement, by and between the City of	of Strongsville, S	State of Ohio (hereinafter referred
to as "City") and Columbia Township, County	of Lorain, State	e of Ohio (hereinafter referred to as
"Township"), is hereby entered into on this $_$	day of	, 2024.

WITNESSETH:

Whereas, North Marks Road from Snell/Albion Road to Sprague Road in Columbia Township and the City of Strongsville is in need of repair and resurfacing; and

Whereas, the City and the Township are desirous of entering into a "Cooperation Agreement" for the repair and resurfacing of North Marks Road from Snell/Albion Road to Sprague Road (hereinafter referred to as the "Project").

Now, therefore, in consideration of the foregoing the parties hereto do mutually agree as follows:

1. Lead Agency

The City does hereby authorize the Township to be the Project sponsor and as such prepare, execute and submit a funding application to the Ohio Public Works Commission (OPWC) District 9 to assist with the project funding, bid and award the project. Additionally, the City authorizes the Township under the direction of the Lorain County Engineer to prepare the design specifications, bid documents and administer the construction contract for the project. The construction cost is estimated to be \$523,659.

2. Project Financing

- a. The Township, on behalf of itself and the City, has requested \$308,959 or 59% of the estimated project cost from the Ohio Public Works Commission, District 9.
- b. The City will provide \$68,076 or 13% of the total project cost, whichever is the lesser amount. Furthermore, the City agrees to pay its share of the cost upon Project completion.
- c. The Township will provide \$68,076 or 13% percent of the total project cost, whichever is the lesser amount.

- d. Cuyahoga County has made available, \$78,549 from their Annual Community Maintenance Program. A formal application will be submitted by the City in the fall of 2024.
- e. Any costs above and beyond the construction cost estimate or above the available grant funding will be split equally between the City and the Township.

_	
3.	Project Authorization a. This Project is authorized by Strongsville City Ordinance Number
	a. This Project is authorized by Strongsville City Ordinance Numberb. This Project is authorized by Columbia Township Resolution Number
4.	Term of Agreement
	This Agreement shall commence at the latest date set forth below.
5.	Successors and Assigns
	This Agreement shall be binding upon and inure to the benefit of the City and the
	Township. This agreement and its rights and responsibilities shall not be assigned or
	transferred without the prior written consent of the City and the Township.
6.	Severability
	If any section of this Agreement is found to be illegal, unconstitutional, improper or
	unenforceable, said section shall not affect the enforceability of the remainder of this
	Agreement.
7.	Jurisdiction
	This Agreement shall be construed in accordance with the laws of the State of Ohio.
IN WI	TNESS WHEREOF, Lorain County and Columbia Township, by their duly
autho	rized agents, have executed this Agreement on the dates set forth next to their signatures.
City o	f Strongsville
RV.	DATE:
	as Perciak, Mayor
Appro	eved as to form:

DATE:

IN

Neil M. Jamison, Law Director

Assistant County Prosecutor

CITY OF STRONGSVILLE, OHIO

ORDINANCE NO. 2024 - 138

By: Mayor Perciak and All Members of Council

AN ORDINANCE AUTHORIZING AND DIRECTING THE MAYOR TO ISSUE AND APPROVE CHANGE ORDER NOS. 12 THROUGH 14 FOR AN INCREASE IN THE CONTRACT PRICE CONSISTENT WITH THE PROVISIONS OF THE CONTRACT BETWEEN THE CITY OF STRONGSVILLE AND DIGIOIA-SUBURBAN EXCAVATING, LLC, IN CONNECTION WITH THE PROSPECT ROAD STORM SEWER IMPROVEMENTS PROJECT, AND DECLARING AN EMERGENCY.

WHEREAS, pursuant to Ordinance No. 2023-130, Council authorized the Mayor to enter into a contract with Digioia-Suburban Excavating, LLC in connection with the Prospect Road Storm Sewer Improvements Project from Albion Road to Fair Road, in the City of Strongsville, (the "Project"), in an amount not to exceed \$3,607,488.50; and

WHEREAS, by and though Ordinance Nos. 2024-044 and 2024-097, Council authorized the Mayor to approve Change Order Nos. 1 through 11, as recommended by the City Engineer, and that it would be in the City's best interest to include as part of the Project additional work required by the Cleveland Water Department and further work required due to unforeseen conditions encountered in connection with a watermain break, together with additional work required to open cut Prospect Road for sanitary sewer lateral connections, all in the amount of \$379,527.86 for an increase in the total Project cost to \$3,987,016.36; and

WHEREAS, the City's Engineer has now recommended that it would be in the best interests of the City to include changes in the work performed or to be performed on the Project by Digioia-Suburban Excavating, LLC, generally being additional work required for sanitary sewer long connections, removal of an unknown/abandoned gas line, and unforeseen field conditions encountered, all as more fully set forth collectively on Exhibit A attached hereto and incorporated herein as if fully rewritten, and to provide additional payment for such changes in the work in the amount of \$96,892.59, for a new total Project cost of \$4,083,908.95.

NOW, THEREFORE, BE IT ORDAINED BY THE COUNCIL OF THE CITY OF STRONGSVILLE, COUNTY OF CUYAHOGA AND STATE OF OHIO:

Section 1. That the Mayor be and is hereby authorized and directed to issue and approve Change Order Nos. 12 through 14 to the contract in the amount of \$96,892.59, as recommended by the City Engineer and reflected collectively on Exhibit A; and after the issuance and approval of said Change Order Nos. 12 through 14 and completion of such work, to direct the Director of Finance to make payment to **DIGIOIA-SUBURBAN EXCAVATING, LLC** in the additional amount of \$96,892.59, thereby increasing the total Project cost to \$4,083,908.95.

Section 2. That the funds necessary for this Ordinance have been appropriated and shall be paid from the General Capital Improvement Fund, and that the Director of Finance be and is hereby authorized and directed to issue the City's warrants for payment accordingly, and in accordance with the terms and conditions of the contract and change orders.

CITY OF STRONGSVILLE, OHIO ORDINANCE NO. 2024 – <u>138</u> Page 2

Section 3. That it is found and determined that all formal actions of this Council concerning and relating to the adoption of this Ordinance were adopted in an open meeting of this Council; and that all deliberations of this Council, and any of its committees, that resulted in such formal action were in meetings open to the public in compliance with all legal requirements.

Section 4. That this Ordinance is hereby declared to be an emergency measure necessary for the immediate preservation of the public peace, property, health, safety and welfare of the City, and for the further reason that it is immediately necessary to provide for changes in the work in order to properly and timely complete the Project, to facilitate payment to the contractor for unanticipated changes in the work, to avoid potential legal problems, and conserve public funds. Therefore, provided this Ordinance receives the affirmative vote of two-thirds of all members elected to Council, it shall take effect and be in force immediately upon its passage and approval by the Mayor; otherwise from and after the earliest period allowed by law.

	President o	f Council	Mayor	
Date Passed	l:		Date Approved:	
	<u>Yea</u>	<u>Nay</u>	Attest: Clerk of Council	
Carbone Clark Kaminski Kosek Roff Short Spring			Ord. No. <u>2024 - /38</u> Amended:	
			Public HrgRef:Ref:	

Annround.

CHANGE ORDER

Order No.

12-14

			Date: Agreement Date:	September 16, 2024 September 15, 2023
Name of PROJECT	: Prospect I	Road Storm Sev	wer Improvements -	– Albion to Fair Road
CONTRACTOR:	DiGioia-Su	ıburban Excav	ating, LLC	
The following chan	ges are herel	y made to the O	CONTRACTOR DO	CUMENTS:
Justification: Addit	tional work a	nd unforeseen o	conditions	
Change to CONTR	ACT PRICE			
Original CONTRA	CT PRICE:	\$3,607,488.50		
Current CONTRAC	CT PRICE ad	justed by Previo	ous CHANGE ORDI	ER: \$3,987,016.36
The CONTRACT F	PRICE due to	this CHANGE	ORDER will be inc	reased by: \$96,892.59
The new CONTRA	CT PRICE i	ncluding this CI	HANGE ORDER wi	ll be: \$4,083,908.95
Change to CONTR	ACT TIME:	N/A		
The CONTRACT T	TIME will be	(increased) (de calendar	, ·	
The date for comple	etion of all W	ORK will be (I	Date):	
Requested by:	D'C' '	N E E YO		
D 1.11	D1G101a-3	Suburban Exca	ivating, LLC	
Recommended by:_	Ken Mik	ula, P. E., City	Engineer	
Accepted by:	74.K			
	Mayor T	homas P. Perci	ак	

Change Order Request No. 00012



DIGIOIA-SUBURBAN EXCAVATING, LLC 11293 Royalton Rd. North Royalton, OH 44133

Phone: 440-237-1978 Fax: 440-237-1976

www.digioiasuburban.com

TI	T	ı	Ε	•

Open Cut Sanitary Connections

DATE:

5/29/2024

PROJECT:

Prospect Road Storm Sewer Improvements

JOB NO:

23-113

TO:

City of Strongsville

CONTRACT NO.:

0

16099 Foltz Pkwy

Strongsville, OH 44149 Phone: 440-580-3100 Fax:

ATTN: Jamie Kilbane

Description of Change Order Request

Please refer to the included "Extra Work Forms" for additional information and work descriptions

Total Amount of Change Order Request:

\$58,054.32

COLLECTED CHANGES

Item	Date	EWO#	Title	Amount
001	2/29/2024	N/A	Attempted 6" Long Side Laterals-February	\$48,456.11
			Subtotal Collected Changes:	\$48,456.11
			Collected Changes (Labor):	\$29,319.34
			Collected Changes (Equipment):	\$25,121.77
			Collected Changes (Material):	\$5,715.00
			Collected Changes (Subcontractor):	\$0.00
			Collected Changes (Misc/Fees):	\$0.00
			Collected Changes (Unit Prices):	(\$11,700.00)
			OH Labor & Equipment (15.0%):	\$8,166.17
			OH Material (15.0%):	\$857.25
			OH Subcontractor (15.0%):	\$0.00
			OH Misc/Fees (15.0%):	\$0.00
			OH Unit Prices	\$0.00
			Subtotal:	\$57,479.52
			Bond (1.0%):	\$574.80
			Total Change Order Request Cost:	\$58,054.32
By:			Ву:	
Printed Name:			Printed Name:	
Company:	•		Company:	
Date:			Date:	

Change Order Request No. 00013



DIGIOIA-SUBURBAN EXCAVATING, LLC 11293 Royalton Rd. North Royalton, OH 44133

Phone: 440-237-1978 Fax: 440-237-1976

www.digioiasuburban.com

Т	IT	1	E.

Existing Gas Line Removal

DATE:

8/2/2024

PROJECT:

Prospect Road Storm Sewer Improvements

JOB NO:

23-113

TO:

City of Strongsville

CONTRACT NO.:

0

16099 Foltz Pkwy

Strongsville, OH 44149

Phone: 440-580-3100 Fax:

ATTN: Jamie Kilbane

Description of Change Order Request

This change request is for the removal of 4,750 LF of existing unknown gas line in the storm trench.

Total Amount of Change Order Request:

\$30,875.00

COLLECTED CHANGES

COLLEGIED CHANGES				
Item	Date	EWO#	Title	Amount
001	8/2/2024	N/A	Existing Gas Line Removal - \$6.50 / LF	\$30,875.00
			Subtotal Collected Changes:	\$30,875.00
			Total Change Order Request Cost:	\$30,875.00
Ву:	Lancon and the second		Ву:	
Printed Name:	-	,	Printed Name:	
Company:	**		Company:	
Date:			Date:	

Change Order Request No. 00014



DIGIOIA-SUBURBAN EXCAVATING, LLC 11293 Royalton Rd. North Royalton, OH 44133

Phone: 440-237-1978 Fax: 440-237-1976

www.digioiasuburban.com

TI	ΤI	E:	

Water Main Break Delay

DATE:

7/8/2024

PROJECT:

Prospect Road Storm Sewer Improvements

JOB NO:

23-113

TO:

City of Strongsville

CONTRACT NO.:

0

16099 Foltz Pkwy

Strongsville, OH 44149

Phone: 440-580-3100 Fax:

ATTN: Jamie Kilbane

Description of Change Order Request

Please refer to the included "Extra Work Forms" for additional information and work descriptions.

Total Amount of Change Order Request:

\$7,963.27

COLLECTED CHANGES

Item	Date	EWO#	Title	Amount
001	5/22/2024	JR93	Existing Watermain Break @ 34+75	\$6,856.03
			Subtotal Collected Changes	\$6,856.03
			Collected Changes (Labor)	\$3,342.14
			Collected Changes (Equipment)	\$3,138.89
			Collected Changes (Material)	\$0.00
			Collected Changes Sales Tax (0.0%)	\$0.00
		¥	Collected Changes (Subcontractor)	\$375.00
			Collected Changes (Misc/Fees)	\$0.00
			OH Labor & Equipment (15.0%)	\$972.15
			OH Material (15.0%)	\$0.00
			OH Subcontractor (15.0%)	\$56.25
			OH Misc/Fees (15.0%)	\$0.00
			Subtota	l: \$7,884.43
			Bond (1.0%)	\$78.84
			Total Change Order Request Cost	\$7,963.27
Ву:			Ву:	
Printed Name:			Printed Name:	
Company:			Company:	
Date:			Date:	

CITY OF STRONGSVILLE, OHIO

ORDINANCE NO. 2024 - 139

By: Mayor Perciak and All Members of Council

AN ORDINANCE AUTHORIZING THE MAYOR TO ENTER INTO A STATE OF OHIO ADMINISTRATIVE SERVICES CONTRACT FOR THE PURCHASE OF ONE (1) NEW SUTPHEN STOCK #568 CUSTOM PUMPER ON SUTPHEN MONARCH CHASSIS WITH RELATED VEHICLE MODIFICATIONS AND APPURTENANCES, FOR USE BY THE FIRE DEPARTMENT, AND DECLARING AN EMERGENCY.

WHEREAS, Ohio Revised Code Section 5513.01(B) provides the opportunity for counties, townships and municipal corporations to participate in contracts of Ohio Department of Administrative Services for the purchase of vehicles, machinery, materials, supplies or other articles; and

WHEREAS, this Council wishes to take advantage of that opportunity to purchase one (1) new Sutphen Stock #568 Custom Pumper on Sutphen Monarch Chassis with related vehicle modifications and appurtenances (State Contract No. STS023111, Index No. STS618), for use by the Fire Department of the City.

NOW, THEREFORE, BE IT ORDAINED BY THE COUNCIL OF THE CITY OF STRONGSVILLE, COUNTY OF CUYAHOGA AND STATE OF OHIO:

- Section 1. That the Mayor accordingly be and is hereby authorized and directed to request authority in the name of the City of Strongsville to participate in an Ohio Department of Administrative Services contract with SUTPHEN CORPORATION (through authorized dealer Herb Fire Equipment) for the purchase of one (1) new Sutphen Stock #568 Custom Pumper on Sutphen Monarch Chassis with related vehicle modifications and appurtenances, in a total amount not to exceed \$995,300.00, as reflected in the Purchase Agreement attached hereto as Exhibit A and incorporated herein.
- **Section 2.** That the City of Strongsville hereby agrees to be bound by the terms and conditions prescribed by the Director of Administrative Services for such purchases, and to directly pay the vendor, under each such contract of the Ohio Department of Administrative Services in which the City participates for items it receives pursuant to the contract.
- **Section 3.** That the Mayor and Director of Finance be and are hereby authorized to enter into and execute such agreements and documents as may be necessary to participate in the Ohio Department of Administrative Services Office of Procurement Services Program.
- **Section 4.** That the funds for the purposes of said purchase and Agreement have been appropriated and shall be paid from the Emergency Vehicle Fund.
- **Section 5.** That it is found and determined that all formal actions of this Council concerning and relating to the adoption of this Ordinance were adopted in an open meeting of this Council; and that all deliberations of this Council, and any of its committees, that resulted in such formal action were in meetings open to the public in compliance with all legal requirements.

CITY OF STRONGSVILLE, OHIO ORDINANCE NO. 2024 – <u>139</u> Page 2

Section 6. That this Ordinance is hereby declared to be an emergency measure necessary for the immediate preservation of the public peace, property, health, safety and welfare of the City, and for the further reason that it is immediately necessary to enter into said contract to provide for the continuity of services and operation of the City of Strongsville Fire Department, to provide updated equipment for fire rescue services, to protect the lives of persons in the City of Strongsville, and conserve public funds. Therefore, provided this Ordinance receives the affirmative vote of two-thirds of all members elected to Council, it shall take effect and be in force immediately upon its passage and approval by the Mayor; otherwise from and after the earliest period allowed by law.

£	President of	Council	Approved:	Mayor	
Date Passed	:		Date Approved:		_
	<u>Yea</u>	<u>Nay</u>	Attest:CI	erk of Council	_
Carbone Clark Kaminski Kosek Roff Short Spring			Ord. No. <u>2024~</u> 1 st Rdg 2 nd Rdg 3 rd Rdg	Ref: Ref:	
			Public Hrg	Ref: Defeated:	



PURCHASE AGREEMENT

FOR SUTPHEN FIRE APPARATUS

20 <u>3</u> Oh Pui	IS AGREEMENT made and entered into this day of
1.	PURCHASE: Purchaser hereby agrees to buy and Sutphen hereby agrees to sell and furnish to Purchaser the apparatus and equipment according to the Sutphen Proposal dated 9/11/24 and made a part hereof, and the Proposal Price Page with the Sutphen Terms and Conditions also made part of this Agreement as attached hereto ("Proposal").
2.	PURCHASE PRICE: Purchaser agrees to pay for said apparatus and equipment the total Purchase Price of Nine Hundred Ninety Five Thousand Three Hundred & 0/100 (\$ 995,300.00).
	Changes to National Fire Protection Association ("NFPA") 1900, Environmental Protection Agency ("EPA") or changes legislated by Federal, State or Local Governments or changes in part availability or vendor relationships that impact the cost to manufacture the truck may incur additional charges which shall be borne by the purchaser. These may include but are not limited to changes that affect the major vendors of the fire apparatus industry such as pump manufacturers, seat manufacturers, electrical power supplies (generators) and powertrain (engine 8 transmission). Sutphen shall provide written notice to the Purchaser as soon as it reasonably believes any provision may be invoked. Sutphen shall provide, upon written request, documentation of such changes and increases.
	Any such changes shall be documented on a change order executed by both Sutphen and Purchaser.
3.	<u>PAYMENT</u> : Final payment shall be made at the time of final inspection at the factory as per Sutphen Proposal Terms and Conditions. Should payment be delayed, Sutphen reserves the right to charge interest at the rate of one and one-half percent (1.5%) per month, beginning on the day after payment is due.

- 4. <u>FINAL COMPLETION</u>: The apparatus and equipment being purchased hereunder shall be completed within approximately <u>February 2025</u> months after the receipt and acceptance of this agreement at Sutphen's office as per Sutphen Proposal, provided that such delivery date shall be automatically extended for delays beyond Sutphen's control, including, without limitation, strikes, labor disputes, riots, civil unrest, pandemics, war or other military actions, sabotage, government regulations or controls, fire or other casualty, or inability to obtain materials or services. If such delay occurs, Sutphen shall give notice of delay to Purchaser. Purchaser shall not be entitled to any discount or reduction in price for such delay and Sutphen shall not be liable for any damages (compensatory, incidental, consequential or otherwise) related to such delay.
- 5. MANUFACTURER'S CERTIFICATE OF ORIGIN: Delivery, payment, and transfer of the Manufacturer's Certificate of Origin (MCO) shall take place at Sutphen during final inspection, and upon payment in full in accordance with the terms of this Agreement. Sutphen reserves the right to withhold delivery of the MCO until payment in full is received. If Purchaser requires any third-party equipment mounting, the apparatus shall be moved to the third-party facility by the dealer or Purchaser for such mounting. Such third-party work shall not delay or offset payment to Sutphen.
- 6. <u>SUTPHEN WARRANTIES</u>: Sutphen warrants to Purchaser that all goods and services furnished hereunder will conform in all respects to the terms of this order, including any applicable change orders, drawings, specifications, or standards incorporated herein, and/or shall be free of defects in materials, workmanship, and free from such defects in design. In addition, Sutphen warrants that the goods and services are suitable for and will perform in accordance with the purposes for which they were intended, for a period of one year from the Warranty Registration Date, unless an extended warranty is purchased.
- 7. SHORTAGES AND OMISSIONS: The apparatus shall be tested per NFPA #1900 at Sutphen's manufacturing facility. Purchaser agrees that the apparatus and equipment being purchased hereunder shall not be driven or used in any manner until it is paid for in full. In the event there are any shortages or omissions with the apparatus at time of completion, Purchaser may withhold a sum equivalent to the price of any such shortages as determined by Sutphen.

- 8. <u>DEFAULT</u>: In the case of any default in payment hereunder or in the payment on any notes, negotiable paper, obligations or other instruments issued by Purchaser, Sutphen may take full possession of the apparatus and equipment or of the piece or pieces upon which default has been made, and any payments that have been made theretofore shall be applied as rent in full for the use of the apparatus and equipment up to the date of taking possession by Sutphen.
- PURCHASER GUARANTEES: With the signing of this agreement, Purchaser attests that it has the full power and legal authority to enter into this agreement and guarantees that funds for its purchase are available or in the process of collection.
- 10. **ACCEPTANCE**: This agreement shall not be binding until it is signed and approved by an officer of the Sutphen Corporation.
- 11. TAXES, ETC.: The purchase price provided for herein does not include any federal, state or local sales tax, duties, imposts, revenues, excise or other taxes which may hereafter be imposed by governmental authority or otherwise and which are made applicable to the apparatus or equipment covered by the agreement. In the event that any such taxes are subsequently imposed and become applicable, the purchase price herein shall be increased by the amount of such taxes and such sum shall be immediately paid by Purchaser to Sutphen. To the extent applicable, the prices and deliveries set forth herein are subject to the Defense Production Act.
- 12. **INSURANCE**: Sutphen shall provide insurance insuring the apparatus and equipment against loss by fire, theft, or collision and insuring against property damage and personal injury through the completion of the apparatus and transfer of the Manufacturer's Certificate of Origin.
- 13. <u>TERMINATION</u>: After the execution of this Agreement, Purchaser shall have no right to terminate the Agreement. Sutphen may, in its absolute and sole discretion, accept Purchaser's request to terminate the Agreement. In the event Sutphen accepts Purchaser's request to terminate the Agreement, Sutphen may charge a cancellation fee. The following charge schedule based on costs incurred may be applied, at Sutphen's sole discretion:(a) 10% after order is accepted by Sutphen; (b) 30% of the Purchase Price after production has commenced. The cancellation fee may increase accordingly as costs are incurred as the order progresses through engineering and into manufacturing.
- 14. <u>GENERAL</u>: This agreement and the Sutphen Proposal provided herein take precedence over all previous negotiations, oral or written, and no representations or warranties are applicable except as specifically contained in this agreement or in the Sutphen Proposal. No alteration, modification, amendment or change of this Agreement shall be binding unless executed in writing by the parties. No waiver of any of the provisions of this Agreement shall be deemed a waiver of any other provision, whether or not similar, nor shall any waiver constitute a continuing waiver. If a Purchase Order is also issued, this Agreement and the Sutphen Proposal Terms and Conditions shall supersede the terms in the Purchase Order where terms may be inconsistent. This Agreement may be signed in counterparts by the parties, each

of which shall be deemed an original, but all of which together shall be deemed to be one and the same Agreement. A signed copy of this Agreement delivered by facsimile, e-mail, or other means of electronic transmission shall be deemed to have the same legal effect as delivery of an original signed copy of this Agreement. This Agreement may only be amended, modified or supplemented by an agreement in writing signed by each party hereto.

15. GOVERNS: This Agreement shall be governed and controlled as to interpretation, enforcement, validity, construction, effect and in all other respects by the laws, statutes and decisions of the State of Ohio. Exclusive jurisdiction and venue for any litigation at all related to this in the Franklin County Court of Common Pleas, Columbus, Ohio, and the parties hereto consent and submit to the general jurisdiction of this court. All of the terms and provisions of this Agreement shall be binding upon and inure to the benefit of and be enforceable by Sutphen, Purchaser, their successors and assigns.

IN WITNESS WHEREOF, the parties hereto have caused this agreement to be duly executed and attested by its duly authorized representatives, effective as of the date below when accepted at Sutphen Corporations offices.

SALES REPRESENTATIVE OF THE SUTPHEN CORPORATION	PURCHASER
	City of Strongsville
Ву	
By Sales Representative	
	Ву
Accepted at offices of:	TitleThomas P. Perciak, Mayor
The Sutphen Corporation 6450 Eiterman Rd.	1 itle
Dublin, Ohio 43016	Date
Ву	Ву
Title	Title
Date	Date

CITY OF STRONGSVILLE, OHIO ORDINANCE NO. 2024 - 140 BY: MAYOR THOMAS P. PERCIAK

AN ORDINANCE MAKING APPROPRIATIONS FOR THE ANNUAL EXPENSES AND OTHER EXPENDITURES OF THE CITY OF STRONGSVILLE, OHIO, FOR THE YEAR 2024 AND REPEALING ORDINANCE NUMBER 2024 - 093

NOW THEREFORE, BE IT ORDAINED BY THE COUNCIL OF THE CITY OF STRONGSVILLE, COUNTY OF CUYAHOGA, AND STATE OF OHIO:

Section 1: THAT THERE BE APPROPRIATED FROM THE FOLLOWING FUNDS AND AS FURTHER DETAILED IN THE SCHEDULE ATTACHED HERETO AS EXHIBIT "A" AND INCORPORATED HEREIN:

	<u>General Fu</u>	ınc	<u>i - 101</u>		
Fund # Fund Activity	Personal Service		Other	Transfers & Advances	Total
101 Total General Fund	\$ 22,453,301.00	\$	10,919,370.00	\$ 22,259,500.00	\$ 55,632,171.00
	Special Revenu	e F	-unds - 200		
	Personal			Transfers &	
Fund # Fund Activity	Service		Other	Advances	Total
203 Police Pension	\$ 1,735,000.00	\$	•••	\$ _	\$ 1,735,000.00
204 Street Construction & Maintenance	5,481,850.00		7,048,750.00		12,530,600.00
205 State Highway Maintenance	_		450,000.00	-	450,000.00
206 Motor Vehicle License Tax	-		450,000.00	-	450,000.00
207 Emergency Vehicle Fund	-		2,976,788.00	-	2,976,788.00
208 Fire Levy	9,401,801.00		1,448,000.00	_	10,849,801.00
209 Fire Pension	1,800,000.00		-	-	1,800,000.00
211 Clerk of Court	-		23,000.00	_	23,000.00
212 Drainage Levy	-		1,323,000.00	-	1,323,000.00
213 Local Fiscal Recovery	938,908.00		-	300,000.00	1,238,908.00
214 Multi-Purpose Complex	3,472,577.00		2,406,090.00	-	5,878,667.00
215 Southwest General Hospital	=		370,000.00	_	370,000.00
216 Law Enforcement Federal Seizures	-		60,000.00	_	60,000.00
217 Law Enforcement State Seizures	-		5,000.00	•	5,000.00
218 Law Enforcement Drug Fine	•		1,000.00	-	1,000.00
219 Law Enforcement DWI/DUI	-		15,000.00	-	15,000.00
220 Tree Fund	-		183,000.00	_	183,000.00
222 Community Diversion	-		3,000.00		3,000.00
223 Bond Escrow			718,500.00	-	718,500.00
224 Earned Benefits	405,000.00		-	-	405,000.00
225 One Ohio Settlement Fund	_		30,000.00	 	30,000.00
200 Total Special Revenue Funds	\$ 23,235,136.00	\$	17,511,128.00	\$ 300,000.00	\$ 41,046,264.00

<u>Debt Service Funds - 300</u>								
Fund # Fund Activity		Personal Service		Other		Transfers & Advances		Total
331 General Bond Retirement	\$	60,000.00	\$	4,115,275.00	\$		\$	4,175,275.00
333 Pearl Road TIF # 1		-		580,288.00		-		580,288.00
334 Royalton Road TIF				176,700.00		_		176,700.00
335 Pearl Road TIF # 2		_		86,317.00		-		86,317.00
336 Pearl Road TIF # 3		-		45,583.00				45,583.00
337 Westwood Commons TIF				38,000.00		-		38,000.00
338 Giant Eagle TIF		-		132,620.00		-		132,620.00
339 GETGO TIF		-		20,000.00				20,000.00
340 Clover Senior TIF				157,100.00		-		157,100.00
341 Pearl Road TIF # 4		-		265,120.00		-		265,120.00
342 Cane's/Chase TIF		-		55,810.00		-		55,810.00
343 Brighton Best TIF				3,975.00				3,975.00
344 Pearl North TIF		-		108,075.00		t <u>-</u>		108,075.00
346 Camden Woods T/F		-		483.00		9,000.00		9,483.00
300 Total Debt Service Funds	\$	60,000.00	\$	5,785,346.00	\$	9,000.00	\$	5,854,346.00

		Personal		Transfers &	
Fund #	Fund Activity	Service	Other	Advances	Total
	441 Recreation Capital Improvement	\$ -	\$ 157,000.00	\$ -	\$ 157,000.00
	442 General Capital Improvement	-	19,901,884.00	-	19,901,884.00
	447 TIF Capital Improvements	5 =	334,256.00	· ·	334,256.00
	448 Town Center Improvement Fund	-	909,920.00	 5,753,500.00	6,663,420.00
	400 Total Capital Project Funds	\$ -	\$ 21,303,060.00	\$ 5,753,500.00	\$ 27,056,560.00

Fund #	Fund Activity	Service		Other	Advances	Total
	551 Sanitary Sewer	\$ 7,157,208.00	\$	5,199,959.00	\$ -	\$ 12,357,167.00
		Internal Service	e F	und - 600		
		Personal			Transfers &	
Fund #	Fund Activity	Service		Other	Advances	Total
	661 Health Insurance Reserve	\$ -	\$	6,862,500.00	\$ -	\$ 6,862,500.00
	664 Worker's Compensation Reserve	-		495,000.00	-	495,000.00
	600 Total Internal Service Funds	\$ -	\$	7,357,500.00	\$	\$ 7,357,500.00
	Grand Total All Funds	\$ 52,905,645.00	\$	68,076,363.00	\$ 28,322,000.00	\$ 149,304,008.00

Description	Amount
General Fund to General Capital Improvement Fund	7,000,000.00
General Fund to Fire Levy Fund	5,850,000.00
General Fund to Street Construction Fund	3,000,000.00
General Fund to Multi-Complex Fund	2,700,000.00
General Fund to Police Pension Fund	1,500,000.00
General Fund to Earned Benefits Fund	400,000.00
General Fund to Fire Pension Fund	1,200,000.00
General Fund to Recreation Capital Improvement Fund	500,000.00
General Fund to Tree Fund	100,000.0
Total Transfers	\$ 22,250,000.0
General Fund to Camden Woods TIF Fund	\$ 9,500.0
Town Center Capital Improvement fund to General Fund	5,753,500.0
Local Fiscal Recovery fund to General Fund	300,000.0
Camden Woods TIF fund to General Fund	9,000.0
Total Advance and Advance Repayments	\$ 6,072,000.0
Total Transfers, Advances and Advance Repayments	\$ 28,322,000.0

That all expenditures within the fiscal year ending December 31, 2024 shall be made in accordance Section 2: with the code accounts set forth above, and shall be made within the appropriations herein provided.

That it is found and determined that all formal actions of this Council concerning and relating to the Section 3: adoption of this Ordinance were adopted in an open meeting of this Council, and that all deliberations of this Council and any of its committees that resulted in such formal action were in meetings open to the public in compliance with all legal requirements.

AS AN ORDINANCE providing for the appropriation of monies and consistent with the City's Section 4: Charter Article III, Section 13, this Ordinance shall take effect immediately upon its passage and approval by the Mayor, or otherwise at the earliest time allowed by law.

Presi	dent of Council	Approved	l:	Mayor	
D	ate Passed			ORD. No. 2024 - 140	Amended:
	Attest:	Clerk of Council		1st Rdg.	Ref:
	Yea	<u>Nay</u>		2nd Rdg	Ref:
Carbone Clark				3rd Rdg.	Ref:
DeMio				Edward and the American	
Kaminski Kosek				Dub Ura	Ref:
Roff Short			2 of 4	Pub Hrg	Defeated:
SHOLL				Adopted.	

EXHIBIT "A" SCHEDULE OF BUDGETS BY DEPARTMENT - page 1 of 2

	Department	Services	Other	Advances	Total
011410	Council	\$ 397,200.00	\$ 41,800.00	\$ -	\$ 439,000
011411	Mayors Office	329,580.00	16,800.00	-	346,380
015412	Police Department	11,856,762.00	2,703,300.00	¥*	14,560,062
011413	Human Resources	288,600.00	74,040.00	-	362,640
011414	Finance Department	477,100.00	18,200.00	-	495,300
011415	Legal Department	536,700.00	70,350.00	-	607,050
011416	Communication & Technology	855,246.00	926,300.00		1,781,546
011417	Building Department	1,094,260.00	236,600.00	=	1,330,860
011418	Mayors Court	171,700.00	252,200.00	-	423,900
011420	Rubbish Department	-	3,103,500.00	-	3,103,500
011421	Cemetery Department	147,140.00	361,960.00		509,100
011422	Architectural Board of Review	-	4,000.00	-	4,000
011423	Planning Commission	125,943.00	67,500.00	-	193,443
011424	Civil Service		27,700.00	=	27,700
011425	Board of Appeals	-	22,850.00	-	22,850
011428	Parks Department	134,800.00	342,500.00	_	477,300
011429	Public Safety	227,263.00	-	_	227,263
011430	General Miscellaneous		2,114,200.00	_	2,114,200
011435	Economic Development	213,900.00	93,670.00	•	307,570
015415	OPIOID Grant	31,950.00	15,300.00	_	47,250
015414	Corrections Officers	1,269,050.00	126,700.00	_	1,395,750
015413	Regional Dispatch Center	4,296,107.00	299,900.00	_	4,596,007
011468	Non Government Transfers	-1,2,00,101,00	200,000.00	22,259,500.00	22,259,500
011700	Total General Fund	\$ 22,453,301.00	\$ 10,919,370.00	\$ 22,259,500.00	\$ 55,632,171
031000	Police Pension	1,735,000.00	\$ 10,010,010.00	Ψ <u>LL,L00,000.00</u>	1,735,000
		4,620,700.00	5,086,750.00		9,707,450
046419	Street Repairs	127,900.00	262,000.00	-	389,900
046426	Traffic Signal Maintenance	127,800.00	850,000.00	-	850,000
046427	Snow Removal	722 250 00		-	
046433	Municipal Garage	733,250.00	850,000.00	-	1,583,250
056000	State Highway Maintenance	•	450,000.00	-	450,000
066000	Motor Vehicle License Tax	-	450,000.00	•	450,000
075000	Emergency Vehicle Fund	9,401,801.00	2,976,788.00	-	2,976,788
085000	Fire Levy	9,401,001.00	1,177,800.00	-	10,579,601
085001	Fire Station Ward 1	-	57,200.00	-	57,200 44,500
085002	Fire Station Ward 2	-	44,500.00	•	
085003	Fire Station Ward 3	-	40,000.00	-	40,000
085004	Fire Station Ward 4	4 000 000 00	128,500.00	-	128,500
095000	Fire Pension	1,800,000.00	-	-	1,800,000
111000	Clerk of Court	-	23,000.00	-	23,000
121000	Drainage Levy	-	1,323,000.00		1,323,000
131000	Local Fiscal Recovery	938,908.00		300,000.00	1,238,908
143304	Sports Programs	312,000.00	275,700.00	-	587,700
143305	Recreation Administration	550,200.00	701,500.00	-	1,251,700
143306	Fitness	444,500.00	150,700.00	••	595,200
143309	Towncenter Park	164,977.00	126,700.00	-	291,677
143310	Aquatics	747,200.00	153,600.00	-	900,800
143311	Recreation Programs	138,200.00	98,500.00	=	236,700
143430	Special Events	-	31,190.00		31,190
143431	Old Town Hali	8,900.00	23,600.00	-	32,500
143439	Senior Services	581,800.00	564,100.00	-	1,145,900
143451	Recreation Maintenance	524,800.00	251,500.00	-	776,300
143500	Program Refunds	-	29,000.00	-	29,000
152000	Southwest General Hospital	-	370,000.00	-	370,000
165000	Law Enforcement Federal Seizures	-	60,000.00	-	60,000
175000	Law Enforcement State Seizures	-	5,000.00	100	5,000
185000	Law Enforcement Drug Fine	-	1,000.00	-	1,000
195000	Law Enforcement DWI/DUI		15,000.00	-	15,000
204000	Tree Maintenance	-	183,000.00	_	183,000
225000	Community Diversion	_	3,000.00		3,000
223100	Bond Escrow	_	718,500.00	_	718,500
224000	Earned Benefits	405,000.00		_	405,000
250000	One Ohio Settlement Fund	.00,000.00	30,000.00	_	30,000
~~~~~	CAS OTHO COMOMOTIC FUND		30,000.00		55,000

### EXHIBIT "A" SCHEDULE OF BUDGETS BY DEPARTMENT - page 2 of 2

ept#	Department	Personal Service		Other		Transfers & Advances		Total
-p					141-5	Juvanoco		1000
311000	General Bond Retirement	60,000.00	)	4,115,275.00		_		4,175,275.0
333000	Pearl Road TIF # 1			580,288.00				580,288.0
334000	Royalton Road TIF	_		176,700.00		_		176,700.0
335000	Pearl Road TIF # 2	_		86,317.00		_		86,317.0
336000	Pearl Road TIF # 3	_		45,583.00				45,583.0
337000	Westwood Commons TIF			38,000.00		-		38,000.0
338000	Giant Eagle TIF	_		132,620.00		_		132,620.0
339000	GETGO TIF	_		20,000.00		_		20,000.0
340000	Clover Senior TIF	-		157,100.00				157,100.0
341000	Pearl Road TIF # 4			265,120.00		_		265,120.0
342000	Cane's/Chase TIF	_		55,810.00		_		55,810.0
343000	Brighton Best TIF	_		3,975.00				3,975.0
344000	Pearl North TIF	_		108,075.00				108,075.0
346000	Camden Woods TIF	_		483.00		9,000.00		9,483.0
	Total Debt Service	\$ 60,000.00	\$		\$	9,000.00	\$	5,854,346.0
413000	Recreation Capital Improvement			157,000.00				157,000.0
421000	General Capital Improvement	=		19,901,884.00		<b></b>		19,901,884.0
447100	Pearl & Whitney TIF			52,625.00		-		52,625.0
447102	Prospect & Albion TIF	<b></b>		21,478.00		_		21,478.
447103	Goodyear & 5/3 TIF	_		51,547.00		_		51,547.
447104	42/82 TIF	_		112,760.00		_		112,760.0
447105	Dunkin Donuts TIF	_		10,739.00		-		10,739.0
447106	Pearl & Lunn TIF			21,478.00		_		21,478.0
447107	Progressive Quality Care	_		31,144.00		_		31,144.
447108	Infinium TIF	_		2,148.00		_		2,148.0
447109	Sprague Road TIF	_		18,256.00				18,256.0
447110	Freddy's TIF			6,443.00		_		6,443.0
447111	Arby's TIF	_		5,638.00		_		5,638.6
448108	Town Center Improvement Fund	_		909,920.00		5,753,500.00		6,663,420.0
770100	Total Capital Projects	\$ -	ŝ	21,303,060.00	\$	5,753,500.00	\$	27,056,560.0
	Total Capital Tojocc	<b>_</b>		#1,000,000.00	Ť	0,100,000.00	· · ·	21,000,000.
512501	Engineering and Administration	795,700.00	)	941,525.00		-		1,737,225.0
512502	Plant Expenditures	5,000,000.00	)	2,687,000.00		-		7,687,000.0
512503	Line Expenditures	1,361,508.00	)	903,000.00		-		2,264,508.0
512504	Sewer Capital Improvements	-		500,000.00		_		500,000.0
512505	Sewer Debt Payments	-		168,434.00		_		168,434.0
	Total Sanitary Sewer	\$ 7,157,208.00	\$		\$		\$	12,357,167.0
004500	Desire beautiful and the second of the secon			6 060 500 60				0.000.500
661000	Health Insurance Reserve	-		6,862,500.00		<del></del>		6,862,500.0
664000	Workers Compensation Reserve	<u>-</u>		495,000.00	•	<del>-</del>	<u>.</u>	495,000.0
	Total Internal Service	\$ -	\$	7,357,500.00	\$	-	\$	7,357,500.0
	Total internal octave	7	·····	.,,	Ť			

### CITY OF STRONGSVILLE, OHIO

### **RESOLUTION NO. 2024 - 141**

By: Mayor Perciak and All Members of Council

A RESOLUTION ACCEPTING THE AMOUNTS AND RATES AS DETERMINED BY THE BUDGET COMMISSION AND AUTHORIZING THE NECESSARY TAX LEVIES AND CERTIFYING THEM TO THE COUNTY FISCAL OFFICER, AND DECLARING AN EMERGENCY.

WHEREAS, pursuant to Sections 5705.34 and 5705.35 of the Ohio Revised Code, the Cuyahoga County Budget Commission has informed the City of Strongsville of the amounts and rates of the necessary tax levies for tax year 2024, to be collected in year 2025; and

WHEREAS, this Council, in accordance with the provisions of law, has previously adopted a Tax Budget for the next succeeding fiscal year commencing January 1, 2025; and

WHEREAS, the Budget Commission of Cuyahoga County, Ohio has certified its action thereon to this Council together with an estimate by the County Fiscal Officer of the rate of each tax necessary to be levied by this Council, and what part thereof is without, and what part within the ten mill tax limitation.

NOW, THEREFORE, BE IT RESOLVED BY THE COUNCIL OF THE CITY OF STRONGSVILLE, COUNTY OF CUYAHOGA AND STATE OF OHIO:

**Section 1.** That the amounts and rates, as determined by the Budget Commission in its certification, be and the same are hereby accepted.

**Section 2.** That there be and is hereby levied on the tax duplicate of the City of Strongsville, the rate of each tax necessary to be levied within and without the ten mill limitation as follows:

## SCHEDULE A SUMMARY OF AMOUNTS REQUIRED FROM GENERAL PROPERTY TAX APPROVED BY BUDGET COMMISSION AND COUNTY FISCAL OFFICER'S ESTIMATED TAX RATES

Fund	Amount to be Derived from	Amount Approved by Budget Commission	County Fiscal Officer Estimate of Tax Rate to be Levied		
rulia	Levies Outside 10 M. Limitation	Inside 10 M. Limitation	Inside 10 M. Limit	Outside 10 M. Limit	
	Column II	Column IV	V	VI	
General Fund				1.50	
General Bond Retirement Fund			2.30		
Police Pension Fund			0.30		
Fire Pension Fund			0.30		
Fire Fund				3.50	
Sewer Fund				0.40	
Southwest General H. C.				1.00	
Total			2.90	6.40	

### SCHEDULE B LEVIES OUTSIDE 10 MILL LIMITATION, EXCLUSIVE OF DEBT LEVIES

Fund		Maximum Rate Authorized to Be Levied	Co. Fiscal Officer's Est. of Yield of Levy (Carry to Schedule A. Column II)
General Fund:			
Current Expense Levy authorized by voters on for not to exceed years.	, 20		
Total General Fund outside 10m. Limitation.		Albania - I	
Park Fund: Levy authorized by voters on for not to exceed years.	, 20		
Recreation Fund: Levy authorized by voters on for not to exceed years.	, 20	AND THE PROPERTY OF THE PROPER	

**Section 3.** That the Clerk of Council be and is hereby authorized and directed to promptly certify and transmit a copy of this Resolution to the Cuyahoga County Fiscal Officer.

**Section 4.** That it is found and determined that all formal actions of this Council concerning and relating to the adoption of this Resolution were adopted in an open meeting of this Council; and that all deliberations of this Council, and any of its committees, that resulted in such formal action were in meetings open to the public in compliance with all legal requirements.

**Section 5.** That this Resolution is hereby declared to be an emergency measure necessary for the immediate preservation of the public peace, property, health, safety and welfare of the City, and for the further reason that it is immediately necessary in order to comply with all state, county and local requirements concerning tax amounts and rates. Therefore, provided this Resolution receives the affirmative vote of two-thirds of all members elected to Council, it shall take effect and be in force immediately upon its passage and approval by the Mayor; otherwise from and after the earliest period allowed by law.

# CITY OF STRONGSVILLE, OHIO RESOLUTION NO. 2024 – 141 Page 4

	President of	Council	Approved:	Mayor	-
Date Passed:		Date Approved:			
	<u>Yea</u>	<u>Nay</u>	Attest:Clerk	of Council	_
Carbone Clark Kaminski Kosek Roff Short Spring			Ord: No. 2004 - 19  1st Rdg		
			Public Hrg	Ref: Defeated:	

# CITY OF STRONGSVILLE, OHIO RESOLUTION NO. 2024 - 141 Page 5

) ss.

State of Ohio

### CERTIFICATE OF COPY ORIGINAL ON FILE

Cuyahoga County )		
whose custody of the files an kept, do hereby certify that the	ork of the Council of the City of Strongsville, within and food records of said Council are required by the Laws of the foregoing is taken and copied from the original Resignal bean compared by me with said original docume reof.	ne State of Ohio to be solution No. 2024-141
WITNESS my signat	ure, this 16 th day of September, 2024.	
	Aimee Pientka, Cle	rk of Council
	Resolution No. 2024 – 141	
	COUNCIL OF THE CITY OF STRONGSVILLE Cuyahoga County, Ohio	
	RESOLUTION ACCEPTING THE AMOUNTS AND RATES AS DETERMINED BY THE BUDGET COMMISSION AND AUTHORIZING THE NECESSARY TAX LEVIES AND CERTIFYING THEM TO THE COUNTY FISCAL OFFICER	
	(City Council)	
	Adopted September 16, 2024	
	Aimee Pientka, Clerk of Council	
	Filed, 2024	
	County Fiscal Officer	
	By	
	Deputy	
	<b>!</b>	l l

### CITY OF STRONGSVILLE, OHIO

### **ORDINANCE NO. 2024 – <u>142</u>**

By: Mayor Perciak and All Members of Council

AN ORDINANCE AUTHORIZING THE MAYOR TO ENTER INTO A PURCHASE AGREEMENT FOR CERTAIN PROPERTY LOCATED AT 21500 LUNN ROAD, AND DECLARING AN EMERGENCY.

WHEREAS, the City deems it to be in the best interest to purchase a parcel of property located at 21500 Lunn Road for municipal public purposes.

NOW, THEREFORE, BE IT ORDAINED BY THE COUNCIL OF THE CITY OF STRONGSVILLE, COUNTY OF CUYAHOGA AND STATE OF OHIO, BY UNANIMOUS AFFIRMATIVE VOTE:

- **Section 1.** That this Council hereby authorizes the Mayor to enter into a Purchase Agreement for the acquisition of property located on Lunn Road, a copy of such Purchase Agreement is attached hereto as Exhibit "1," for property identified as PPNs 393-12-003 and 393-12-004.
- **Section 2.** That upon receipt of a duly executed Deed from the Estate of Arlene A. Bislich, conveying said property to the City and evidence of title satisfactory to the Law Director, the Clerk of Council is hereby directed to cause the said Deed to be recorded with the Cuyahoga County Fiscal Officer.
- **Section 3.** That the Mayor and Director of Finance are hereby further authorized and directed to take any and all other necessary steps, to execute on behalf of the City any and all other documents necessary to effectuate and finalize the purchase, and to carry out all terms and conditions of the Purchase Agreement.
- **Section 4.** That the Director of Finance be and is hereby authorized and directed to pay to the Escrow Agent the amount required in Exhibit "1."
- **Section 5.** That the funds for the purposes of this Ordinance have been appropriated and shall be paid from the General Capital Improvement Fund.
- **Section 6.** That it is found and determined that all formal actions of this Council concerning and relating to the adoption of this Ordinance were adopted in an open meeting of this Council; and that all deliberations of this Council, and of any of its committees, that resulted in such formal action were in meetings open to the public in compliance with all legal requirements.
- **Section 7.** That this Ordinance is hereby declared to be an emergency measure necessary for the immediate preservation of the public peace, property, health, safety and welfare of the City, and for the further reason that it is immediately necessary to enter into an agreement for the purchase of such property in order to further economic development and improve lots and lands in the City. Therefore, provided this Ordinance receives the unanimous

# CITY OF STRONGSVILLE, OHIO ORDINANCE NO. 2024 - 142 Page 2

affirmative vote of all members elected to Council, it shall take effect and be in force immediately upon its passage and approval by the Mayor.

	President of Co	ouncil	Approved:	Mayor
Date Passed:_			Date Approved:	
	<u>Yea</u>	<u>Nay</u>	Attest:Clerk of	Council
Carbone Clark Kaminski Kosek Roff Short Spring			Ord. No 2024-142 1st Rdg	_Amended: Ref: _Ref: _Ref: _Ref:
			Public HrgAdopted:	_Ref:_ _Defeated:

### **PURCHASE AGREEMENT**

THIS AGREEMENT made this _____ day of _____, 2024, by and between the ESTATE OF ARLENE A. BISLICH, hereinafter referred to as "Seller" and the CITY OF STRONGSVILLE, hereinafter referred to as "Buyer."

WHEREAS, Seller is the fee owner of real property located at 21500 Lunn Road, in the City of Strongsville, County of Cuyahoga, State of Ohio, known as Permanent Parcel Nos. 393-12-003 & 393-12-004, which is further described in Exhibit "A" attached hereto and incorporated herein, together with all improvements thereon collectively, and together with all appurtenances, the "Property."

#### WITNESSETH THAT:

- PREMISES: Seller agrees to sell and convey the Property to Buyer and Buyer agrees to purchase the Property from Seller, for the purchase price and upon the terms and conditions hereafter set forth.
- 2. <u>CONDITION OF PROPERTY</u>: This Property is being purchased in its present physical condition "as-is." Buyer has not relied upon any representations, warranties or statements about the Property, including but not limited to, its condition or use, except for those which are expressly set forth in this Agreement. Buyer acknowledges it has not relied on any representations or statements concerning the condition or value of the Property, the use that can be made of the Property or the zoning ordinances applicable to it, or anything concerning the same, other than those statements and representations which are expressly set forth in this Agreement.
- 3. <u>PURCHASE PRICE AND PAYMENT</u>: The total purchase price shall be Two Hundred Seventy Five Thousand Dollars (\$275,000.00), payable at Closing, in cash, or immediately available and collected funds.

- 4. <u>REAL ESTATE TAXES</u>: Real estate taxes and assessments shall be prorated in escrow, as of the date of Closing. All utilities and sewage charges shall be prorated as of the date of Closing by the parties outside of escrow. Seller to pay all the above said costs and charges attributable to its ownership period and Buyer to pay said costs and charges thereafter.
- 5. <u>PLANS, SURVEYS, and ASSESSMENTS</u>: Seller agrees to provide to the Buyer, at no cost to the Buyer, immediately, but not later than ten (10) days after execution of this Agreement, any physical or topographic surveys, development information, soil boring or groundwater data, environmental assessments and other agreements affecting the Property, as well as any other records relating to the Property in the possession of the Seller.
- 6. <u>WARRANTIES; REPRESENTATIONS</u>: Seller warrants and represents the following:
- (a) That Patricia L. Wozniak is the Executor of the Estate of Arlene A. Bislich, Cuyahoga County Probate Court, Case No. 2021 EST 264725;
- (b) That as the Executor she has all necessary authority to execute this Purchase Agreement and sign the deed and any and all other documents required to carry out this Agreement and transfer the Property to the Buyer;
  - (c) There are no other contracts for sale or options involving the Property;
- (d) That the Property, to the best of Seller's knowledge, without any inquiry or investigation by the Seller, is not located in a flood plain; and there are no underground tanks or wells on the Property; and
- (e) That between the date the Seller executes this Agreement and the Closing date, the Seller shall not subject the Property to or consent to any leases, liens, encumbrances, covenants, conditions, restrictions, easements, rights of way, or agreements, or take any other action affecting or modifying the status of title or otherwise affecting the Property, without the written consent of the Buyer.

- ADVERSE FACTS: The Seller, without inquiry or investigation, knows of no materially adverse fact, affecting or threatening to affect the Property which has not been disclosed to the Buyer in writing. Between the date the Seller executes this Agreement and the Closing date, the Seller will notify the City in writing of any events which occur or any facts of which it becomes aware which would make any of its representations or warranties false or misleading. Except as otherwise permitted by the Buyer in writing, in its sole discretion, each of the warranties or representations made in this Agreement by the Seller shall be true and correct as of the Closing date.
- 8. <u>CLOSING</u>: Closing and delivery of possession shall be on or before <u>October 11, 2024</u>, or as soon thereafter as practicable, allowing a reasonable time for approval by City Council, preparation and approval of documents and correction of defects reported as a result of a title examination, survey or inspections of the Premises. Closing shall be held at the offices of the Escrow Agent, or at such other place as the parties may agree.

The Seller agrees to deliver to Buyer or Escrow Agent the following at or prior to Closing:

- (a) A Fiduciary Deed, fully executed by the Seller, conveying the Property in fee simple to the Buyer, in a form satisfactory to the Buyer;
- (b) A signed closing or settlement statement prepared or approved by Escrow Agent;
- (c) Any affidavit or document required by the Escrow Company ensuring that the individual(s) signing the deed and Closing documents have the authority on behalf of the Seller to execute the documents; and
- (d) Any other documents reasonably required by Escrow Agent.

### 9. TITLE EVIDENCE AND DEFECTS:

(a) On the Closing date, Buyer shall receive an ALTA Owner's Policy of Title Insurance, in currently utilized Form (amended) (the "Title Policy") in the amount of \$275,000.00,

showing fee simple title to the Premises to be in Buyer's name, subject only to those title exceptions permitted in accordance with this Agreement, including but not limited to the recorded easements, assessments, covenants and deed restrictions, with the so-called "printed" or "standard" exceptions deleted to the extent possible. The Title Policy shall be issued by the Escrow Agent or such other nationally recognized title company as shall be acceptable to Buyer (the "Title Company"). The Title Policy shall also affirmatively insure: (i) Buyer's right to use any appurtenant easements; (ii) that the Property will have the benefit of direct ingress and egress, both pedestrian and vehicular, to and from a public highway; and (iii) such other matters as Buyer may reasonably require to address material matters disclosed in the Title Commitment.

- (b) Upon execution of this Agreement, Buyer shall order from the Escrow Agent a commitment for the Title Policy (the "Title Commitment") and cause the same (with legible copies of all exceptions attached thereto) to be delivered to each party hereto no later than fifteen (15) days thereafter. The Title Commitment shall be updated and endorsed to include all liens, claims, encumbrances, easements, rights-of-way, encroachments, reservations, restrictions and any other matters affecting the Property, which shall commit to delete the standard printed exceptions and creditors rights' exclusion, and any matters disclosed by any Survey. Within fifteen (15) days after receipt of both the Commitment and the Survey, Buyer shall serve upon Seller a notice specifying those exceptions to title, if any, that interfere with the use of the Property for Buyer's intended use (the "Title Defects"). Buyer acknowledges that, for purposes of this section, real estate taxes and assessments, both general and special, that are not yet due and payable, the recorded easements, covenants and deed restrictions acceptable to the Buyer, and zoning ordinances do not constitute Title Defects. Any exceptions appearing after issuance of the Title Commitment and/or Survey shall also be deemed Title Defects unless approved in writing by Buyer.
  - (c) Upon receipt by Seller of Buyer's notice of Title Defects, if any, Seller shall

immediately and diligently pursue the removal of the Title Defects. Seller shall have fifteen (15) days after receipt of notice in which to cure such Title Defects (or, if the Title Defects are not readily curable within said fifteen (15) day period, then Seller may have such additional time as Buyer may permit in writing, in which case, the Closing Date shall, at Buyer's option, be extended accordingly) (said fifteen (15) day period, as the same may be extended, being hereinafter referred to as the ("Cure Period"). If some or all of the Title Defects can only reasonably be cured at Closing, then Seller may agree in writing to cure such Title Defects at Closing, subject to Buyer's reasonable consent. Should Seller fail to cure the Title Defects within the Cure Period, Seller shall notify Buyer of such fact prior to the expiration of the Cure Period, and Buyer shall have the option to: (i) accept the Property subject to the Title Defects, or (ii) declare this Agreement to be null and void and of no further force or effect, except to the extent a party is in breach of this Agreement and Buyer and Seller shall each be relieved of all further liability hereunder, and except that the parties shall equally share all title and escrow costs incurred to date. If Buyer elects not to terminate this Agreement as provided above, then Seller shall cause the Title Company to update the Title Commitment prior to the Closing Date.

### 10. ESCROW AGENT:

- (a) The escrow agent for this transaction shall be Maximum Title & Escrow Services, Inc. (the "Escrow Agent").
- (b) The Closing of this transaction described in this Agreement shall be conducted in escrow, and this Agreement shall serve as escrow instructions for that purpose. The Escrow Agent may incorporate its standard conditions of acceptance, to the extent that they are not inconsistent with the terms of this Agreement. Seller shall deposit the executed Fiduciary Deed and such other documents as the Escrow Agent may reasonably require, to complete the closing with the Escrow Agent. Buyer shall concurrently deposit the Purchase Price with the Escrow Agent in collected and immediately available funds, and with such other documents as the Escrow

Agent may reasonably require to complete the closing. At Closing, the Escrow Agent shall record the Fiduciary Deed, shall issue the Title Policy and shall disburse the Purchase Price to Seller (net of all closing expenses and prorations as calculated pursuant to this Agreement). The Escrow Agent shall charge the parties with costs and expenses as follows: (1) Seller shall be responsible for the costs of recording the Fiduciary Deed, releasing any leases, purchase options, and encumbrances, and filing any required affidavit, and for one-half of all escrow costs, including title examination and title insurance, and (2) Buyer shall be responsible for one-half of all escrow costs, including title examination and title insurance. Since the transfer is to a municipal corporation, there is no conveyance fee required to be paid by the Seller.

### 11. <u>INSPECTIONS</u>:

- (a) Seller shall give to Buyer and its designated agents and representatives full access to the Property during normal business hours throughout the Buyer Study Period as defined in Paragraph 11 (b), including the right, at Buyer's own risk, to cause its agents or representatives to enter upon the Property for the purpose of (i) making physical and topógraphic surveys; and (ii) conducting such tests, investigations and studies as Buyer may desire, including, but not limited to, those related to engineering, water, groundwater, sanitary and storm sewer, utilities and environmental matters, as well as soil borings. In the event the Property is not transferred to the Buyer, Buyer shall, at its expense, restore the Property to its prior condition to the extent of any changes made by its agents or representatives. Any private agents or representatives retained by the Buyer to do this work shall carry insurance to protect the Seller. The Seller shall furnish to Buyer during the Buyer Study Period all information concerning the Property which the Buyer may reasonably request and which is in the possession of the Seller.
- (b) Buyer shall have fifteen (15) days from the full execution of this Agreement, ("Buyer Study Period"), to complete the studies described in Paragraph 11(a) and to determine in its sole discretion that the condition of the Property is satisfactory for the intended use of the Buyer.

In the event that the Buyer is not so satisfied for any reason whatsoever at any time prior to the expiration of the Buyer Study Period, Buyer shall advise the Seller in writing of its intention not to proceed to Closing under the terms of this Agreement, and in such event, this Agreement shall automatically be terminated, and no party shall have any liability hereunder other than the Buyer covering any Title Company costs incurred to date.

- 12. ENVIRONMENTAL MATTERS: Seller states that it has not conducted any environmental studies on the Property and does not know the state of the Property. Seller warrants and represents that to the best of Seller's knowledge, information and belief, without inquiry or investigation, there have never been, nor are there now, any underground storage tanks on the Property; nor has there been any activity on the Property which has been conducted or is being conducted, except in compliance with all statutes, ordinances, regulations, orders, permits and common law requirements concerning (a) handling of any toxic or hazardous substances, (b) discharges of toxic or hazardous substances to the air, soil, surface water or groundwater, and (c) storage, treatment or disposal of any toxic or hazardous substances at or connected with any activity on the Property; nor is there any contamination present on or in the Property; nor is there any of the following present on or in the Property: (i) polychlorinated biphenyls or substances containing polychlorinated biphenyls; (ii) asbestos or materials containing asbestos; (iii) urea formaldehyde or materials containing urea formaldehyde; (iv) lead or lead-containing paint; or (v) radon. The term "contamination" shall mean the unconfined presence of toxic or hazardous substances on or in the Property or arising from the Property, which may require remediation under any applicable law. For purposes of this Agreement, "hazardous substance(s)" shall have the meaning of "hazardous substance" set forth in 42 U.S.C. §9601(14), as amended, and of "regulated substance" at 42 U.S.C. §6991(2), as amended, and of any other substances which may be the subject of liability pursuant to any environmental law of the United States and the State of Ohio.
  - 13. <u>CONTINGENCIES</u>: This Agreement is contingent upon the following:

Approval of this Agreement by City Council and all required City of Strongsville officials pursuant to law.

- 14. <u>BROKERAGE</u>: Both parties warrant and represent that they have not dealt with any realtor, broker, consultant, or like agent who might be entitled to any compensation in connection with the transaction contemplated hereby.
- 15. <u>DELIVERY OF DOCUMENTS</u>: After closing, the Escrow Agent shall deliver to Seller its escrow statement, and shall deliver to Buyer its escrow statement, along with the Title policy and the recorded Fiduciary Deed.
- 16. ENTIRE AGREEMENT AND MODIFICATIONS: This Agreement, including all exhibits attached hereto and hereby incorporated herein by reference, contains all of the terms and conditions agreed upon by the parties hereto, there being no oral conditions representations, warranties or agreements. Any subsequent conditions, representations, warranties or agreements shall not be valid and binding upon the parties unless in writing and signed by both parties.
- 17. <u>ASSIGNMENT</u>: Seller shall not assign or hypothecate this Agreement, in whole or in part, nor mortgage, transfer or convey the property to anyone except Buyer. Buyer shall not assign this Agreement, in whole or in part, without Seller's prior written consent.
- 18. <u>HEADINGS</u>: The headings of the paragraphs of this Agreement are for convenience only and shall not affect the meaning or construction of the contents of this Agreement.
- 19. <u>SUCCESSORS AND ASSIGNS</u>: This Agreement shall be binding upon and inure to the benefit of the parties, their respective heirs, administrators, executors, successors and assigns.
- 20. <u>GOVERNING LAW</u>: This Agreement shall be governed by and interpreted under and construed in accordance with the laws of the State of Ohio.

21. <u>MULTIPLE COPIES</u>: This Agreement shall be executed in duplicate, each of which shall be an original of this Agreement, but all of which taken together shall constitute one and the same.

IN WITNESS WHEREOF, the parties have hereunto set their hands on the day and year first above written.

WITNESSES:	SELLER:
J. A. Ganbino	ESTATE OF ARLENE A. BISLICH  Patricia L. Wozniak, Executor  8/2/24
	BUYER:
	CITY OF STRONGSVILLE
	By: Thomas P. Perciak, Mayor

### EXHIBIT A

#### LEGAL DESCRIPTION

### Parcel No. 1

Situated in the City of Strongsville, County of Cuyahoga, and State of Ohio and known as being part of Original Strongsville Township Lot No. 84, bounded and described as follows:

Beginning in the centerline of Lunn Road at its intersection with the Westerly line of said Original Lot No. 84;

Thence North 88° 35' 10" East along the centerline of Lunn Road, 744.55 feet to an angle point in said centerline;

Thence North 42° 52' 20" East along the centerline of Lunn Road, 621.16 feet to the principal place of beginning of premises herein described, said point being distant South 42° 52' 20" West measured along said centerline, 2,100.99 feet from its intersection with the centerline of Prospect Road;

Thence North 47° 07' 40" West, 300 feet to a point;

Thence North 42° 52' 20" East parallel with the centerline of Lunn Road, 150 feet to a point;

Thence south 47° 07' 40" East, 300 feet to a point in the centerline of Lunn Road;

Thence South 42° 52' 20" West along the centerline of Lunn Road, 150 feet to the principal place of beginning, according to the survey of Warren J. Root, Civil Engineers and Surveyors.

The courses used in this description are given to an assumed meridian and are used to indicate angles only.

Permanent Parcel No. 393-12-004

Property Address: 21500 Lunn Road, Strongsville, Ohio 44149

### Parcel No. 2

Situated in the City of Strongsville, County of Cuyahoga and State of Ohio: And known as being part of Original Strongsville Township Lot No. 84, bounded and described as follows:

Beginning in the center line of Lunn Road at its intersection with the Westerly line of said Original Lot No. 84;

Thence North 88'35' 10" East along the center line of Lunn road, 744.55 feet to an angle point in said centerline;

Thence North 42° 52' 20" East along the centerline of Lunn Road, 371.16 feet to the principal place of beginning of premises herein described, said point being distant South 42° 52' 20" West measured along said centerline, 2350.99 feet from its intersection with the centerline of Prospect Road;

Thence North 47° 07' 40" West, 300 feet to a point;

Thence North 42° 52' 20" East, parallel with the centerline of Lunn Road, 250 feet to a point;

Thence South 47° 07' 40" East, 300 feet to a point in the centerline of Lunn Road;

Thence South 42° 52' 20" West along the centerline of Lunn Road, 250 feet to the principal place of beginning according to the survey of Warren J. Root, Civil Engineers and Surveyors.

The courses used in this description are given to an assumed meridian and are used to indicate angles only.

Permanent Parcel No. 393-12-003

Property Address: Vacant lot on Lunn Road, Strongsville, Ohio 44149